NEUMANN PROBLEMS FOR SECOND ORDER ELLIPTIC OPERATORS WITH SINGULAR COEFFICIENTS

A THESIS SUBMITTED TO THE UNIVERSITY OF MANCHESTER FOR THE DEGREE OF DOCTOR OF PHILOSOPHY IN THE FACULTY OF ENGINEERING AND PHYSICAL SCIENCES

2012

Xue Yang School of Mathematics

Contents

Abstract							
D	Declaration						
С	Copyright Statement						
P۱	ublic	ations	7				
A	ckno	wledgements	8				
D	edica	tion	9				
1	Introduction						
	1.1	Motivation and Contribution	10				
	1.2	Structure of the Thesis	15				
2	Bac	Background Theory					
	2.1	Regular Symmetric Dirichlet Forms	16				
	2.2	Backward Stochastic Differential Equations	25				
	2.3	Useful Inequalities and Lemmas	28				
3	Two-sided Estimates on the Heat Kernels						
	3.1	Introduction	30				
	3.2	A Reduction Method	33				
	3.3	Upper Bound Estimates	39				

		3.3.1	Upper Bounds for Heat Kernels Associated with $G_1 \ldots \ldots$	40			
		3.3.2	Upper Bounds for Heat Kernels Associated with G_2	48			
	3.4 Lower Bound Estimates			52			
		3.4.1	Lower Bounds for Heat Kernels Associated with G_1	53			
		3.4.2	Lower Bounds for Heat Kernels Associated with G_2	58			
	3.5	Two-s	ided Estimates for the Heat Kernel $p(t,x,y)$ of Operator $G_{}$.	59			
		3.5.1	Upper Bounds for the Heat Kernel $p(t, x, y)$	59			
		3.5.2	Lower Bounds for the Heat Kernel $p(t, x, y)$	61			
4	Ne	Neumann Problems for Semilinear Elliptic PDEs					
	4.1	Introd	luction	63			
	4.2	BSDE	s with Singular Coefficients and Infinite Horizon	66			
	4.3	Linear	PDEs	80			
	4.4	Semili	near PDEs	80 87			
	4.5	Semili	near Elliptic PDEs with Singular Coefficients	96			
	4.6	L^1 Sol	lutions to the BSDEs and Semilinear PDEs	101			
5	Future Studies 1						
	5.1	An Ins	spiring Example	112			
	5.2	Future	e Studies	118			
Bi	Bibliography 12						

The University of Manchester

Xue Yang Doctor of Philosophy Neumann Problems for Second Order Elliptic Operators with Singular Coefficients May 18, 2012

In this thesis, we prove the existence and uniqueness of the solution to a Neumann boundary problem for an elliptic differential operator with singular coefficients, and reveal the relationship between the solution to the partial differential equation (PDE in abbreviation) and the solution to a kind of backward stochastic differential equations (BSDE in abbreviation).

This study is motivated by the research on the Dirichlet problem for an elliptic operator ([42]). But it turns out that different methods are needed to deal with the reflecting diffusion on a bounded domain. For example, the integral with respect to the boundary local time, which is a nondecreasing process associated with the reflecting diffusion, needs to be estimated. This leads us to a detailed study of the reflecting diffusion. As a result, two-sided estimates on the heat kernels are established.

We introduce a new type of backward differential equations with infinity horizon and prove the existence and uniqueness of both L^2 and L^1 solutions of the BSDEs. In this thesis, we use the BSDE to solve the semilinear Neumann boundary problem. However, this research on the BSDEs has its independent interest.

Under certain conditions on both the "singular" coefficient of the elliptic operator and the "semilinear coefficient" in the deterministic differential equation, we find an explicit probabilistic solution to the Neumann problem, which supplies a L^2 solution of a BSDE with infinite horizon. We also show that, less restrictive conditions on the coefficients are needed if the solution to the Neumann boundary problem only provides a L^1 solution to the BSDE.

Declaration

No portion of the work referred to in this thesis has been submitted in support of an application for another degree or qualification of this or any other university or other institute of learning.

Copyright Statement

- i. The author of this thesis (including any appendices and/or schedules to this thesis) owns certain copyright or related rights in it (the "Copyright") and s/he has given The University of Manchester certain rights to use such Copyright, including for administrative purposes.
- ii. Copies of this thesis, either in full or in extracts and whether in hard or electronic copy, may be made only in accordance with the Copyright, Designs and Patents Act 1988 (as amended) and regulations issued under it or, where appropriate, in accordance with licensing agreements which the University has from time to time. This page must form part of any such copies made.
- iii. The ownership of certain Copyright, patents, designs, trade marks and other intellectual property (the "Intellectual Property") and any reproductions of copyright works in the thesis, for example graphs and tables ("Reproductions"), which may be described in this thesis, may not be owned by the author and may be owned by third parties. Such Intellectual Property and Reproductions cannot and must not be made available for use without the prior written permission of the owner(s) of the relevant Intellectual Property and/or Reproductions.
- iv. Further information on the conditions under which disclosure, publication and commercialisation of this thesis, the Copyright and any Intellectual Property and/or Reproductions described in it may take place is available in the University IP Policy (see http://www.campus.manchester.ac.uk/medialibrary/ policies/intellectual-property.pdf), in any relevant Thesis restriction declarations deposited in the University Library, The University Librarys regulations (see http://www.manchester.ac.uk/library/aboutus/regulations) and in The Universitys policy on presentation of Theses.

Publications

[1] Weiguo Yang and Xue Yang, "A note on strong limit theorems for arbitrary stochastic sequences", Statist. Probab. Lett. 78 (2008).

[2] Xinfang Han, Li Ma and Xue Yang, "Perturbation of generalized Dirichlet forms by signed smooth measures and the associated Markov processes", (Chinese) Acta Math. Sci. Ser. A Chin. Ed. 30 (2010).

[3] Xinfang Han, Li Ma and Xue Yang, "Remarks on non-symmetric perturbed Dirichlet forms and switching identities". Chinese J. Appl. Probab. Statist. 27 (2011).

[4] Xue Yang and T.S. Zhang, "The estimates of heat kernels with Neumann boundary conditions", (Accepted for publication by Potential Analysis)

Pre-publication articles:

[5] Xue Yang and T.S. Zhang, "A probabilistic approach to Neumann problems of semilinear elliptic PDEs with singular coefficients". (Submitted)

[6] Xue Yang and T.S. Zhang, "Absolute continuity of the law of solutions of SPDEs driven by white noise with coefficients depending on the past of the solutions".(Preprint)

[7] Xue Yang and T.S. Zhang, "A probabilistic approach to Dirichlet problems with non-linear divergence term ". (Preprint)

Acknowledgements

First of all, I would like to express my most sincere thanks to my supervisor Prof. Tusheng Zhang for his help, guidance and patience throughout my three-year Ph.D study. Under his supervision, I have learned so much, not only the professional knowledge, but also the boundless dedication and braveness. His influence on all aspects of my work has been immeasurable.

Furthermore, I am very grateful to Prof. Zhi-ming Ma at the Chinese Academy of Sciences, for showing me the beauty of Mathematics and welcoming me into this research field. I also would like to thank Prof. Zhao Dong and Prof. Fuzhou Gong for their continuous help and encouragement over the years. I am grateful to Prof. Niels Jacob and Dr.Kees Van Schaik for reading this thesis so carefully and giving me so many helpful and inspiring suggestions.

I am indebted to my parents, Chenglin Yang and Furong Shi. It is their love, support and constant encouragement that give me the solid foundation upon which everything else can be built. I also would like to dedicate this thesis to my fiancé, Xinxin Shu, because of his enormous love, understanding and care. I am grateful to him for sharing my happiness and sadness even though we are living at the opposite sides of this world during the long time I have been studying in the United Kingdom.

I am fortunate to have joined the Marie Curie initial training network, not only for the financial support, but also for the precious chance to communicate with so many excellent scholars in my research fields. Last but not least, a big thank you goes to all of the friendly faculty and students at the school of mathematics. It is them who make the time at Manchester the most memorable in my life.

Dedication

To My Parents and Fiancé

Chapter 1

Introduction

1.1 Motivation and Contribution

This thesis is devoted to the study of the semilinear Neumann boundary problem for an elliptic differential operator with singular coefficients, and culminates in an explicit probabilistic solution of this problem. This study is motivated by the previous research on the reflecting diffusions and the Dirichlet problems for such kind of operators.

The theory of reflecting diffusion on a bounded domain plays an important role in this thesis. Reflecting Brownian motion (RBM in abbreviation) on a bounded domain has been studied in different ways. For example, in the view of Skorohod equations (see [3], [21]), RBM (X_t) on bounded domain D can be decomposed as a semimartingale

$$X_t = X_0 + B_t + \frac{1}{2} \int_0^t \vec{n}(X_s) \, dL_s,$$

where (B_t) is a standard Brownian motion, $\vec{n}(x)$ is the unit inward normal vector at $x \in \partial D$ and L_t is a continuous increasing process but increases only when $X_t \in \partial D$. Furthermore, considering the RBM in the framework of Dirichlet form, we know that RBM is a diffusion process associated with the regular Dirichlet form:

$$\mathcal{E}(u,v) = \frac{1}{2} \sum_{i,j} \int_D \frac{\partial u}{\partial x_i} \frac{\partial v}{\partial x_j} dx$$

The generator of RBM is $G = \frac{1}{2} \triangle$ equipped with the Neumann boundary condition $\frac{\partial}{\partial \vec{n}} = 0$ on ∂D .

Reflecting diffusion is a generalization of RBM, by adding the diffusion matrix $A(x) = (a_{ij}(x))$ and the drift term $\sum_i b_i \frac{\partial}{\partial x_i}$ (see [5],[26]). In general, the reflecting diffusion (X_t) has a decomposition in the following form:

$$X_t = X_0 + \int_0^t \sigma(X_s) \, dB_s + \int_0^t b(X_s) \, ds + \frac{1}{2} \int_0^t A\vec{n}(X_s) \, dL_s$$

where the matrix $\sigma(x)$ is the positive definite symmetric square root of the matrix A(x). Inspired by the method in [3],[21], we know that estimates for the local time, from which the integrability of the semigroup is derived, are necessary for the probabilistic solution to a Neumann boundary problem. To this end, a detailed study on the reflecting diffusion is needed.

The operator we consider in the thesis is the following

$$L = \frac{1}{2} \nabla \cdot (A\nabla) + B \cdot \nabla - div(\hat{B}\cdot) + q$$

= $\frac{1}{2} \sum_{i,j=1}^{d} \frac{\partial}{\partial x_i} \left(a_{ij}(x) \frac{\partial}{\partial x_j} \right) + \sum_{i=1}^{d} B_i(x) \frac{\partial}{\partial x_i} - \sum_{i=1}^{d} \frac{\partial \hat{B}_i}{\partial x_i} + q(x).$

L acts on the functions defined on a smooth bounded domain D and the mixed boundary condition

$$\frac{1}{2} < A \nabla u, \vec{n} > - < \hat{B} u, \vec{n} > = \Phi$$

on ∂D is required. The precise description of L is given in Section 3.1. Please note that in the following discussion, when we say a operator is defined on a domain D, it actually means that the operator acts on the functions defined on domain D. The Dirichlet problem to an elliptic differential operator with singular coefficients

$$\begin{cases} Lu(x) = -F(x, u(x), \nabla u(x)), & \text{on } D\\ u(x) = \Phi(x) & \text{on } \partial D \end{cases}$$
(1.1)

has been studied ([7],[8] and [42]). The method dealing with the "bad" term $div(\hat{B}\cdot)$ called "time-reversal", which is the intrinsic motivation of our research, will be used in this thesis.

This thesis mainly studies the following three problems.

(1) Two-sided estimates for the heat kernels associated with the operator L equipped with mixed boundary condition.

Although there has been a great amount of literature on the estimates for heat kernels with Dirichlet boundary conditions (see [7] [33] [40], [41] and references therein), there is not so much work on estimates of heat kernels with Neumann boundary conditions. Here we mention three papers. Two-sided estimates of the heat kernel of reflecting Brownian motion (A = I, $B = \hat{B} = 0$) on Lipschitz domains are obtained in [3]. When the coefficients A and B are smooth and $\hat{B} = 0$, the heat kernels under mixed boundary conditions are constructed in [22] and [34], but the Gaussian bounds are not established for the heat kernel there. Using the estimates on heat kernels established by us, we get the integrability of the semigroup associated with the operator L.

(2) Existence and uniqueness of the solutions to BSDEs with infinite horizon.

Thanks to the development of the BSDEs in recent years, it is possible to represent the solutions of the nonlinear PDEs by the solutions of certain BSDEs associated with a diffusion process generated by some linear operator \mathcal{A} . The first result on a probabilistic interpretation for solutions of semilinear parabolic PDEs is obtained by Peng in [32] and subsequently in [31], in both of which the terminal time of the BSDE is finite. But in our situation, considering the reflecting diffusion, we have to solve the BSDEs with infinite horizon. The integrability of the solution to the BSDE becomes crucial and makes the problem much harder.

Since the term $\int_0^t A\vec{n}(X(s))dL_s$ is involved in the decomposition of reflecting diffusion process X, the BSDE, which we use to solve the nonlinear Neumann problem, also involves an integral with respect to the local time L_t . This is a new type of BSDE. The research on such a kind of BSDE has an independent interest.

(3) Probabilistic solution of the Neumann boundary problem associated with the operator L.

Based on the first two topics, we use probabilistic methods to solve the mixed boundary value problem for semilinear second order elliptic partial differential equations in the following form:

$$\begin{cases} Lu(x) = -F(x, u(x), \nabla u(x)), & \text{on } D\\ \frac{1}{2} \frac{\partial u}{\partial \gamma}(x) - \widehat{B} \cdot n(x)u(x) = \Phi(x) & \text{on } \partial D \end{cases}$$
(1.2)

Probabilistic approaches to boundary value problems of second order differential operators have been adopted by many authors and the earliest work went back as early as 1944 in [24]. So far there has been a lot of studies on the Dirichlet boundary problem (see [8],[16], [36] and [42]). However, there are not many articles on the probabilistic approach to the Neumann boundary problem. Here we only mention one reference. When A = I, B = 0 and $\hat{B} = 0$, the following Neumann boundary problem

$$\begin{cases} \frac{1}{2} \triangle u(x) + qu(x) = 0, & \text{on } D\\ \frac{1}{2} \frac{\partial u}{\partial n}(x) = \phi(x) & \text{on } \partial D \end{cases}$$

is solved in [3], which gives the solution the following representation:

$$u(x) = E_x\left[\int_0^\infty e^{\int_0^t q(\bar{B}_u)du}\phi(\bar{B}_t)dL_t^0\right].$$

Here $(\bar{B}_t)_{t>0}$ defined on the probability space $(\Omega, E_x, x \in D)$ is the reflecting Brownian motion associated with the infinitesimal generator $G = \frac{1}{2}\Delta$, and L_t^0 , t > 0 is the boundary local time satisfying $L_t^0 = \int_0^t I_{\partial D}(\bar{B}_s) dL_s^0$.

There are two essential difficulties in the third topic. One lies in the divergence term $div(\hat{B}\cdot)$ of the operator L. The term $div(\hat{B}\cdot)$ is hard to deal with because the divergence does not exist as \hat{B} is only a measurable vector field. It should be interpreted in the distributional sense. Since \hat{B} is not differentiable, the term $\nabla \cdot (\hat{B}\cdot)$ can not be handled by Girsanov transform or Feynman-Kac transform. Therefore, the "time reversal" method is used here. The other difficulty lies in the boundary local time in the decomposition of the reflecting diffusion. The method dealing with the boundary local time is inspired by the paper [19]. However the equation considered in [19] is linear

$$\begin{cases} (\frac{1}{2} \bigtriangleup - \nu) u(x) = 0 & \text{on } D\\ \frac{\partial u}{\partial n} = \phi & \text{on } \partial D, \end{cases}$$

and only a probabilistic interpretation of the solution to the Neumann problem is given.

In conclusion, the analysis of the reflecting diffusion in the first topic helps to build the BSDE of new type in the second topic. The two topics handled first are the basis of the third one. Note that, every topic has its own interest.

1.2 Structure of the Thesis

This thesis is organized in five chapters. The first chapter is a brief survey of the literature. We summarize the motivation and contribution of this thesis, and indicate the difficulties we meet in the course of our study as well.

Chapter 2 introduces the basic theories of Dirichlet form, backward stochastic differential equations and some inequalities used in the following chapters.

Chapter 3 provides both upper and lower bound estimates on the heat kernel of Gaussian type associated with operator L equipped with the Neumann boundary conditions.

Chapter 4 considers the existence and uniqueness of the solutions (Y, Z) to the following BSDE with infinite horizon:

$$Y_{x}(t) = Y_{x}(T) + \int_{t}^{T} F(X(s), Y_{x}(s), Z_{x}(s)) ds - \int_{t}^{T} e^{\int_{0}^{s} q(X(u)) dt} \Phi(X(s)) dL_{s} - \int_{t}^{T} \langle Z_{x}(s), dM_{x}(s) \rangle, \quad for \quad t < T,$$

where $M_x(t)$ is the martingale of the reflecting diffusion X(t). Actually, both L^1 and L^2 solutions of the BSDE are obtained in this chapter. By using them, we get the solutions of the Neumann boundary problem, which require different conditions on the operator L.

Chapter 5 discusses a future work we are interested in. We want to consider that the divergence term could also be nonlinear, for example, in the form of $div(\hat{B}(x, u(x)))$.

Chapter 2

Background Theory

In this chapter, we recall some background material which will be used in the following chapters.

2.1 Regular Symmetric Dirichlet Forms

Let E be a locally compact separable Hausdorff space. m is a Radon measure with support on E. $L^2(E, m)$ denotes the space of functions defined on E square integrable with respect to the measure m. Let (\cdot, \cdot) denote the inner product on $L^2(E, m)$.

Definition 2.1.1. A family of linear bounded operators $\{T_t, t > 0\}$ with domain $\mathcal{D}(T_t) = L^2(E, m)$ is called a symmetric contraction semigroup, if the following conditions are satisfied:

(1) (Symmetry) $\int_E T_t g(x) f(x) m(dx) = \int_E g(x) T_t f(x) m(dx)$, for any t > 0 and $f, g \in L^2(E, m)$;

(2) (Semigroup property) $T_tT_s = T_{t+s}, t, s > 0;$

(3) (Contraction property) The norm of the operator satisfies $||T_t|| \leq 1$.

Moreover, $\{T_t, t > 0\}$ is said to be strongly continuous if

(4) $(T_t u - u, T_t u - u) \to 0$, as $t \to 0$, for $u \in L^2(E, m)$.

Definition 2.1.2. A family of linear bounded operators $\{G_{\alpha}, \alpha > 0\}$ with domain

 $\mathcal{D}(G_{\alpha}) = L^2(E,m)$ is called a symmetric contraction resolvent, if the following conditions are satisfied:

(1) (Symmetry) $\int_E G_{\alpha}g(x)f(x)m(dx) = \int_E g(x)G_{\alpha}f(x)m(dx)$, for any $\alpha > 0$ and $f, g \in L^2(E, m)$;

(2) (Resolvent equation) $G_{\alpha} - G_{\beta} + (\alpha - \beta)G_{\alpha}G_{\beta} = 0;$

(3) (Contraction property) The norm of the operator satisfies $||G_{\alpha}|| \leq \alpha^{-1}$, for any $\alpha > 0$.

Moreover, $\{G_{\alpha}, \alpha > 0\}$ is said to be strongly continuous if (4) $(\alpha G_{\alpha}u - u, \alpha G_{\alpha}u - u) \rightarrow 0$, as $\alpha \rightarrow \infty$, for $u \in L^{2}(E, m)$.

Definition 2.1.3. For a strongly continuous semigroup $\{T_t, t > 0\}$ on $L^2(E, m)$, the operator $(A, \mathcal{D}(A))$ defined as follows,

$$\begin{aligned} \mathcal{D}(A) &:= \{ f \in L^2(E,m) | \lim_{t \to 0} \frac{T_t f - f}{t} \quad exits \}, \\ Af &:= \lim_{t \to 0} \frac{T_t f - f}{t}, \quad for \quad f \in \mathcal{D}(A), \end{aligned}$$

is called the generator of the semigroup.

The following theorem reveals the relationship between the generator, semigroup and resolvent.

Theorem 2.1.4. ([38]Hille-Yosida's Theorem)

(1) For a strongly continuous semigroup $\{T_t, t > 0\}$ on $L^2(E, m)$, define

$$G_{\alpha}u = \int_0^{\infty} e^{-\alpha t} T_t u \, dt,$$

then $\{G_{\alpha}, \alpha > 0\}$ is a strongly continuous resolvent and every T_t , t > 0 has the following representation:

$$T_t := \lim_{\alpha \to \infty} e^{\alpha(\alpha G_\alpha - 1)t}.$$

(2) For the linear densely defined self-adjoint operator A such that, for any $\alpha > 0$, the inverse operator $(\alpha - A)^{-1}$ exists and is a linear bounded operator on $L^2(E,m)$ satisfying $\|\alpha(\alpha - A)^{-1}\| \leq 1$. We define

$$G_{\alpha} := (\alpha - A)^{-1}.$$

Then $\{G_{\alpha}, \alpha > 0\}$ is a resolvent and the operator can be expressed as $A = \alpha - G_{\alpha}^{-1}$.

Definition 2.1.5. Suppose $\mathcal{D}(\mathcal{E})$ is a dense linear subset in $L^2(E, m)$ and $\mathcal{E}(\cdot, \cdot)$ is a symmetric bilinear form defined on $\mathcal{D}(\mathcal{E}) \times \mathcal{D}(\mathcal{E})$ with values in R. Then $(\mathcal{E}, \mathcal{D}(\mathcal{E}))$ is called a symmetric Dirichlet form if the following conditions are satisfied: (i) $\mathcal{D}(\mathcal{E})$ is a Hilbert space equipped with the inner product $\mathcal{E}_1(,) := \mathcal{E}(,) + (,)$. (ii) (Markovian) For any $\varepsilon > 0$, there exists a real function $\phi_{\varepsilon}(t), t \in R^1$, satisfying

$$\phi_{\varepsilon}(t) = t, \quad \forall t \in [0, 1], \quad -\varepsilon \le \phi_{\varepsilon}(t) \le \varepsilon + 1, \quad \forall t \in \mathbb{R}^{1}, \quad and$$
$$0 \le \phi_{\varepsilon}(t') - \phi_{\varepsilon}(t) \le t' - t \quad whenever \quad t < t',$$

such that

$$u \in \mathcal{D}(\mathcal{E}) \Rightarrow \phi_{\varepsilon}(u) \in \mathcal{D}(\mathcal{E}), \quad \mathcal{E}(\phi_{\varepsilon}(u), \phi_{\varepsilon}(u)) \le \mathcal{E}(u, u).$$

Moreover, the measure m is called the reference measure of the Dirichlet form $(\mathcal{E}, \mathcal{D}(\mathcal{E})).$

The following theorem reveals the relationship between the Dirichlet form, generator and resolvent.

Theorem 2.1.6. ([15], [30])

(1) Given a symmetric Dirichlet form $(\mathcal{E}, \mathcal{D}(\mathcal{E}))$ defined on $L^2(E, m)$, we can define a corresponding non-positive self-adjoint operator in the following way: define

$$\mathcal{D}(A) := \{ f \in \mathcal{D}(\mathcal{E}) : g \mapsto \mathcal{E}(f,g) \text{ is a continuous linear functional on } L^2(E,m) \},$$

then for every $f \in \mathcal{D}(A)$, let Af denote the unique element in $L^2(E,m)$ such that $(-Af,g) = \mathcal{E}(f,g)$ for all $g \in \mathcal{D}(\mathcal{E})$. (2) Let $\{G_{\alpha}, \alpha > 0\}$ be the resolvent which is associated with the operator A defined as in (1), then

$$\mathcal{E}_{\alpha}(G_{\alpha}u, v) := \mathcal{E}(G_{\alpha}u, v) + \alpha(G_{\alpha}u, v) = (u, v).$$

Moreover, for $u \in L^2$, $u \in \mathcal{D}(\mathcal{E})$ if and only if $\lim_{\alpha \to \infty} \alpha(u - \alpha G_{\alpha}u, u)$ exists. In this case,

$$\mathcal{E}(u,u) = \lim_{\alpha \to \infty} \alpha(u - \alpha G_{\alpha} u, u).$$

The Dirichlet forms are also strongly related to a class of Markov processes, so that it is possible to apply the analytic theories to deal with the stochastic processes.

A Markov process $(\Omega, \mathcal{F}, X_t, \mathcal{F}_t, P_x)$ with the state space E is called a Hunt process if $\{X_t\}_{t\geq 0}$ is strong Markovian and quasi-continuous with respect to the σ -filtration $\{\mathcal{F}_t\}_{t\geq 0}$. θ_t and γ_t are the shift and reverse operators on Ω respectively, defined by

$$X_s(\theta_t(\omega)) = X_{t+s}(\omega), s, t \ge 0,$$

$$X_s(\gamma_t(\omega)) = X_{t-s}(\omega), s \le t.$$

Set $\mathcal{B}_b := \{ \text{Borel measurable and bounded function on } E \}.$

Denote by p_t the Markov transition function, i.e. $p_t f(x) = E_x[f(X_t)]$, for any $f \in \mathcal{B}_b$. Then $\{X_t\}_{t\geq 0}$ is called m-symmetric if

$$(u, p_t v) = (p_t u, v),$$

for any $u, v \in \mathcal{B}_b$.

Theorem 2.1.7. ([15])

Every symmetric regular Dirichlet form $(\mathcal{E}, \mathcal{D}(\mathcal{E}))$ with reference measure m is associated with a m-symmetric Hunt process (Ω, \mathcal{F}) , in the sense that $p_t f = T_t f m - a.e.$, for $f \in \mathcal{B}_b \bigcap L^2(E, m)$.

In the rest of the section, we introduce several kinds of functionals associated with

the Hunt process and Dirichlet form.

The reference measure m now is not "fine" enough as long as the Hunt processes are considered, so that a kind of Choquet capacity is used to describe the "small" set. For an open set $G \in E$, the capacity of G is defined as:

$$Cap(G) = \inf \{ \mathcal{E}_1(u, u) | u \in \mathcal{D}(\mathcal{E}), u \ge 1 \text{ on } G \}$$

and $Cap(G) = \infty$ if G is an empty set.

For any subset $B \subset E$, the set function:

$$Cap(B) = \inf\{Cap(G) | \quad G \supset B \quad \text{is a open set}\}$$

can be proved to be a Choquet capacity (in [15]).

A set N, which is a Borel set (i.e. $N \in \mathcal{B}(E)$), is called an exceptional set if Cap(N) = 0. It is proved in [15] that, N is an exceptional set if and only if $P_x(h(N) < \infty) = 0, m - a.e.$, where h(N) is the hitting time for the set N of the process $\{X_t\}_{t>0}$, that is, $h(N) := \inf\{t \ge 0 | X_t \in N\}$.

A function f is said to be quasi-continuous if for any $\varepsilon > 0$ there is an open subset $G \subset E$ with $Cap(G) < \varepsilon$ such that the restriction of f on E - G, $f|_{E-G}$ is continuous. It is proved in [15] that every function $h \in \mathcal{D}(\mathcal{E})$ has a quasi-continuous version, denoted by \tilde{h} .

Definition 2.1.8. An extended real valued process $\{A_t\}_{t\geq 0}$ defined on Ω is an additive functional (AF in abbreviation) if the following conditions are satisfied:

(1) A_t is \mathcal{F}_t -measurable for any t > 0;

(2) there exists a set $\Lambda \in \mathcal{F}_{\infty} := \bigcup_{t} \mathcal{F}_{t}$ and an exceptional set $N \subset D$ with Cap(N) = 0 such that $P_{x}(\Lambda) = 1$ for $x \in E \cap N^{c}$ and $\theta_{t}\Lambda \in \Lambda$ for all $t \geq 0$. Moreover, for any $\omega \in \Lambda$, $t \mapsto A_{t}(\omega)$ is right continuous and has left limit in $t \in [0, \infty]$ with $A_{0}(\omega) = 0$ and $|A_{t}(\omega)| < \infty$ and $A_{t+s}(\omega) = A_{t}(\omega) + A_{s}(\theta_{t}\omega)$.

If $t \mapsto A_t(\omega)$ is positive and continuous, then $\{A_t\}_{t\geq 0}$ is a positive continuous

additive functional (PCAF in abbreviation).

In the following discussion, we use one capital letter A to denote the process $\{A_t\}_{t\geq 0}$ for convenience.

Definition 2.1.9. A positive Borel measure ν is smooth if the following conditions are satisfied:

- (1) $\nu(N) = 0$ if Cap(N) = 0;
- (2) there exists an increasing sequence $\{F_n\}$ of closed sets satisfying $\lim_{n\to\infty} \nu(K E_n)$

 F_n = 0 for any compact set K, such that $\nu(F_n) < \infty$ and $\nu(E - \bigcup_n F_n) = 0$.

It is proved (in [15]) that there is a one-to-one correspondence between the set of smooth measures and the set of PCAFs:

$$\lim_{t \downarrow 0} \frac{1}{t} E_{h \cdot m}(A_t) := \lim_{t \downarrow 0} \frac{1}{t} \int_E h(x) E_x(A_t) m(dx) = \int_E \tilde{h}(x) \nu(dx), \tag{2.1}$$

for any positive function $h \in \mathcal{D}(\mathcal{E})$.

Define

$$\mathcal{M} := \{ M | M \text{ is an AF with exceptional set N}, \forall t > 0, \quad E_x M_t^2 < \infty, \\ E_x M_t = 0, \quad for \quad x \in E - N \}.$$

 \mathcal{M} is called the set of martingale additive functionals (MAF in abbreviation). For $M \in \mathcal{M}$, we define

$$e(M) := \sup_{t>0} \frac{1}{2t} E_m M_t^2 (\leq \infty).$$

e(M) is called the energy of M. A MAF M is said to have finite energy if $M \in \mathring{\mathcal{M}} := \{M \in \mathcal{M} | e(M) < \infty\}.$

By the definition of MAF, it is known that $\{M_t, \mathcal{F}_t, P_x\}_{t\geq 0}$ is a square integrable martingale for $x \in E - N$. Let $\langle M \rangle$ be the sharp bracket process of M ([18]), then $\langle M \rangle$ is a PCAF. We denote by $\mu_{\langle M \rangle}$ the smooth measure associated with the PCAF $\langle M \rangle$ and we call it the energy measure of MAF M. By simple calculation, it can be shown that

$$e(M) = \frac{1}{2}\mu_{\langle M \rangle}(E), \quad M \in \mathcal{M}.$$

In fact, by (2.1), it follows that

$$e(M) = \sup_{t>0} \frac{1}{2t} E_m M_t^2 = \frac{1}{2} \sup_{t>0} \frac{1}{t} E_m < M >_t = \frac{1}{2} \mu_{}(E).$$

A continuous AF (CAF in abbreviation) $\{N_t\}_{t\geq 0}$ is called a CAF of zero energy, if N_t belongs to the following set,

$$\mathcal{N}_c$$
: = {N| N is CAF with exception set Z, $e(N) = 0$,
 $\forall t > 0, \quad E_x N_t < \infty, \quad x \in E - Z$ }.

Theorem 2.1.10. ([15] Fukushima's decomposition) For $u \in \mathcal{D}(\mathcal{E})$, an AF $\{u(X)\}_{t\geq 0}$ can be decomposed as

$$u(X_t) = u(X_0) + M_t^u + N_t^u, \quad t \ge 0,$$

where $M^u \in \mathring{\mathcal{M}}$ and $N^u \in \mathcal{N}_c$.

Moreover, if N^u is a CAF of bounded variation on [0,t] for any t > 0, and μ is the signed measure associated with N^u satisfying $|\mu|(E) < \infty$, then for any bounded $v \in \mathcal{D}(\mathcal{E})$,

$$\mathcal{E}(u,v) = -\int_E v(x)\,\mu(dx)$$

The following example reveals a relationship between the Markov process, Dirichlet form and the various kinds of functionals introduced above.

Example 1. D is a d-dimensional smooth bounded Euclidean domain, dx is the ddimensional Lebesgue measure and $\lambda(dx)$ is the (d-1)-dimensional Lebesgue measure on ∂D .

Let $A(x) = (a_{ij})_{1 \le ij \le d}$: $R^d \to R^d \otimes R^d$ be a smooth, symmetric matrix-valued function and assume that A is uniformly elliptic. That is, there is a constant $\lambda > 1$ such that

$$\frac{1}{\lambda} I_{d \times d} \le A(\cdot) \le \lambda I_{d \times d}.$$

Consider the bilinear form:

$$\begin{cases} \mathcal{E}_0(u,v) = \frac{1}{2} \sum_{i,j} \int_D a_{ij}(x) \frac{\partial u}{\partial x_i} \frac{\partial v}{\partial x_j} \, dx, \\ D(\mathcal{E}_0) = W^{1,2}(D) := \{ u : u \in L^2(D), \frac{\partial u}{\partial x_i} \in L^2(D), i = 1, ..., d \}. \end{cases}$$

It is easy to verify that $(\mathcal{E}_0, D(\mathcal{E}_0))$ is a regular symmetric Dirichlet form and it is associated with a Hunt process $\{X\}_{t\geq 0}$.

For any bounded functions $u, f \in D(\mathcal{E}_0)$, by Fukushima's decomposition, it follows that

$$\begin{split} \int_{D} f(x) \, \mu_{\langle M^{u} \rangle}(dx) &= \lim_{t \downarrow 0} \frac{1}{t} E_{f \cdot m}(u(X_{t}) - u(X_{0}))^{2} \\ &= \lim_{t \downarrow 0} \frac{1}{t} \int_{D} (p_{t}u^{2}(x) - 2u(x)p_{t}u(x) + u^{2}(x))f(x) \, dx \\ &= \lim_{t \downarrow 0} \frac{2}{t} \int_{D} u(x)f(x)(u(x) - p_{t}u(x)) \, dx \\ &\quad -\lim_{t \downarrow 0} \frac{1}{t} \int_{D} u^{2}(x)(f(x) - p_{t}f(x)) \, dx \\ &= 2\mathcal{E}_{0}(uf, u) - \mathcal{E}_{0}(u^{2}, f) \\ &= \sum_{i,j} \int_{D} a_{ij}(x) \frac{\partial u}{\partial x_{i}} \frac{\partial u}{\partial x_{j}} f(x) \, dx. \end{split}$$

Therefore by the fact that $\mu_{<M^u>}(dx) = \sum_{i,j} a_{ij}(x) \frac{\partial u}{\partial x_i} \frac{\partial u}{\partial x_j} dx$, we obtain

$$\langle M^u \rangle_t = \int_0^t \sum_{i,j} a_{ij}(X_s) \frac{\partial u}{\partial x_i}(X_s) \frac{\partial u}{\partial x_j}(X_s) \, ds,$$

and

$$M_t^u = \sum_{i,j} \int_0^t \sigma_{ij}(X_s) \frac{\partial u}{\partial x_i}(X_s) \, dB_s^j = \int_0^t \langle \sigma \nabla u(X_s), dB_s \rangle \, .$$

Here the matrix $\sigma(x)$ is the positive definite symmetric square root of the matrix A(x). $\{B_t\}_{t>0}$ is a d-dimensional standard Brownian motion.

For a positive integer number $i, 1 \leq i \leq d$, we set $u_0(x) = x_i$ on D, where x_i is the *i*th coordinate of x. By Fukushima's decomposition, it follows that

$$X_t^i = X_0^i + \sum_j \int_0^t \sigma_{ij}(X_s) \, dB_s^j + N_t^{u_0}, \quad t > 0.$$

By further calculation, we get that, for any bounded function $f \in D(\mathcal{E}_0)$,

$$\mathcal{E}_{0}(u_{0},f) = \frac{1}{2} \int_{D} \sum_{j} \int_{D} a_{ij}(x) \frac{\partial f}{\partial x_{j}} dx$$

$$= -\frac{1}{2} \int_{D} (\sum_{j} \frac{\partial a_{ij}}{\partial x_{j}}) f(x) dx - \frac{1}{2} \int_{\partial D} (\sum_{j} a_{ij} n^{j})(x) f(x) \lambda(dx). \quad (2.2)$$

Here \vec{n} denotes the inward normal vector to the boundary ∂D . For any $\xi \in \partial D$, $U(\xi)$ denotes a neighborhood of ξ . If there exists a smooth function ψ such that $\partial D \cap U(\xi) = \{x : \psi(x_1, ..., x_d) = 0\}$, and $D \cap U(\xi) = \{x : \psi(x_1, ..., x_d) > 0\}$, then \vec{n} is given by locally, i.e., for $x \in \partial D \cap U(\xi)$,

$$\vec{n}(x) = (n^1(x), ..., n^d(x)) = \left(\frac{\partial \psi}{\partial x_1}(x), ..., \frac{\partial \psi}{\partial x_d}(x)\right) / \left(\sum_{i=1}^d \left(\frac{\partial \psi}{\partial x_i}(x)\right)^2\right)^{\frac{1}{2}}.$$

Denote by $\{L_t\}_{t\geq 0}$ the PCAF associated with the measure $\frac{1}{2}\lambda$. By (2.2), we know that N^{u_0} is associated with the smooth measure

$$\nu(dx) = \frac{1}{2} \left(\sum_{j} \frac{\partial a_{ij}}{\partial x_j}\right) dx + \frac{1}{2} \left(\sum_{j} a_{ij} n^j\right)(x) \lambda(dx).$$

Therefore,

$$N_t^{u_0} = \frac{1}{2} \int_0^t \left(\sum_j \frac{\partial a_{ij}}{\partial x_j}\right) (X_s) \, ds + \left(\sum_j a_{ij} n^j\right) (X_s) \, dL_t.$$

Now we get the decomposition of the Hunt process associated with the Dirichlet

form $(\mathcal{E}_0, D(\mathcal{E}_0))$:

$$X_{t}^{i} = X_{0}^{i} + \sum_{j} \int_{0}^{t} \sigma_{ij}(X_{s}) \, dB_{s}^{j} + \frac{1}{2} \int_{0}^{t} (\sum_{j} \frac{\partial a_{ij}}{\partial x_{j}})(X_{s}) \, ds + \int_{0}^{t} (\sum_{j} a_{ij}n^{j})(X_{s}) \, dL_{s},$$

$$i = 1, ..., d,$$

which can be written simply as follows:

$$X_t = X_0 + \int_0^t \sigma(X_s) \, dB_s + \frac{1}{2} \int_0^t \nabla A(X_s) \, ds + \int_0^t A\vec{n}(X_s) \, dL_s.$$

2.2 Backward Stochastic Differential Equations

Given a probability space (Ω, \mathcal{F}, P) , we denote by E the expectation under the measure P. $\{W_t\}_{t\geq 0}$ is a d-dimensional standard Brownian motion. The σ -filtration (\mathcal{F}_t) is generated by $\{W_t\}_{t\geq 0}$,

$$\mathcal{F}_t := \mathcal{N} \lor \sigma\{W_s; 0 \le s \le t\} := \sigma\{\mathcal{N}, \quad \sigma\{W_s; 0 \le s \le t\}\},\$$

where \mathcal{N} is the set of P-null sets in $\mathcal{F}_{\infty} = \sigma\{W_s; 0 \le s < \infty\}.$

Let $\langle \cdot, \cdot \rangle$ denote the scalar product in the Euclid space \mathbb{R}^n and $|\cdot|$ the length of a vector in \mathbb{R}^n . For any T > 0, define the set of all of the \mathcal{F}_t -adapted, square integrable processes on [0, T] as follows

$$\mathcal{M}(0,T;\mathbb{R}^n) := \{\{v_t\} \mid \{v_t\} \text{ is } \mathbb{R}^n \text{-valued}, \mathcal{F}_t \text{-adapted and } E \int_0^T |v_t|^2 dt < \infty\}.$$

Suppose a function

$$g = g(\omega, t, y, z) : \Omega \times [0, T] \times \mathbb{R}^n \times \mathbb{R}^{n \times d} \to \mathbb{R}^n$$

satisfies the following conditions:

(1) for any $(y, z) \in \mathbb{R}^n \times \mathbb{R}^{n \times d}$, $g(\cdot, y, z)$ is \mathbb{R}^n -valued, \mathcal{F}_t -adapted;

(2) (Lipschitz condition) there is a constant C > 0, such that for any $y, y' \in \mathbb{R}^n$ and $z, z' \in \mathbb{R}^{n \times d}$,

$$|g(t, y, z) - g(t, y', z')| \le C(|y - y'| + |z - z'|);$$

(3) $\int_0^T |g(\cdot,0,0)| ds \in L^2(\Omega, \mathcal{F}_T, P; R).$

Consider the following backward stochastic differential equation:

$$Y_t = \xi + \int_t^T g(s, Y_s, Z_s) ds - \int_t^T Z_s dW_s$$

$$\xi \in \mathcal{F}_T.$$
 (2.3)

Here the processes Y and Z are unknown, and a pair of processes (Y, Z) satisfying (2.3) is called a solution of the BSDE.

The following theorem is a classical result of the existence and uniqueness of the solution (Y, Z).

Theorem 2.2.1. ([31])

Suppose that the function g satisfies the above conditions (1)-(3). Then for any terminal condition $\xi \in L^2(\Omega, \mathcal{F}_T, P; \mathbb{R}^n)$, the BSDE (2.3) has a unique solution $(Y, Z) \in \mathcal{M}(0, T; \mathbb{R}^n \times \mathbb{R}^{n \times d})$. Moreover, Y_0 and Z_0 are constants.

The following example comes from the Chapter 2 in [12] and gives a method to solve a kind of BSDE, and it reveals a relationship between the BSDEs and the PDEs.

Example 2. Set the initial condition $\zeta \in L^2(\Omega, \mathcal{F}_t, P; \mathbb{R}^n)$. Suppose the coefficients b and σ satisfy the Lipschitz condition: there exists a constant C > 0 such that for any $t \in [0, T]$ and $x, x' \in \mathbb{R}$,

$$|b(t, x) - b(t, x')| + |\sigma(t, x) - \sigma(t, x')| \le C|x - x'|.$$

Denote by $(X^{t,\zeta})$ the solution of the following stochastic differential equation:

$$dX_s^{t,\zeta} = b(s, X_s^{t,\zeta})ds + \sigma(s, X_s^{t,\zeta}) dW_s, \quad s \in [t, T],$$

$$X_t^{t,\zeta} = \zeta.$$

By Ito's formula, it follows that $(X^{t,\zeta})$ is associated with the generator:

$$L = \frac{1}{2} \sum_{i,j} a_{ij} \frac{\partial^2 \cdot}{\partial x_i \partial x_j} + \sum_i b_i \frac{\partial}{\partial x_i}.$$

Suppose that $u: [0,T] \times \mathbb{R}^n \to \mathbb{R}^n$ is the solution of the following partial differential equation:

$$\partial_t u(t, x) + Lu(t, x) + f(t, x, u, \sigma \nabla u) = 0,$$

 $u(T, x) = \Phi(x).$

Suppose that the coefficients f and Φ satisfy the following conditions,

(1) (Lipschitz condition) f(t, x, y, z) is Lipschitz continuous with respect to the variables y and z;

(2) (α Hölder condition) $\alpha \in (0,1)$, $\forall (x,y,z), (x',y',z') \in \mathbb{R}^n \times \mathbb{R} \times \mathbb{R}^{1 \times d}$, it follows that

$$|\Phi(x) - \Phi(x')| + |f(t, x, y, z) - f(t, x', y'z')| \le C(|y - y'| + |z - z'| + |x - x'|^{\alpha});$$

(3) (Linear growth) $|f(t, x, 0, 0)| + |\Phi(x)| \le C(1 + |x|),$

then $(u, \sigma \nabla u)(s, X_s^{t,\zeta})$ is a solution of the following BSDE:

$$Y_s^{t,\zeta} = \Phi(X_T^{t,\zeta}) + \int_s^T f(u, X_u^{t,\zeta}, Y_u^{t,\zeta}, Z_u^{t,\zeta}) \, du - \int_s^T Z_u^{t,\zeta} \, dW_u, \quad s \in [t,T].$$

2.3 Useful Inequalities and Lemmas

Lemma 2.3.1 ([10] Gronwall's Inequality). Let I denote an interval of the real line of the form $[a, \infty)$ or [a, b] or [a, b) with a < b. Let α , β and u be real-valued functions defined on I. Assume that β and u are continuous and that the negative part of α is integrable on every closed and bounded subinterval of I.

(1) If β is non-negative and if u satisfies the integral inequality

$$u(t) \le \alpha(t) + \int_{a}^{t} \beta(s)u(s) \, ds, \quad t \in I,$$

then

$$u(t) \le \alpha(t) + \int_a^t \alpha(s)\beta(s)e^{\int_s^t \beta(r)dr} \, ds, \quad t \in I.$$

(2) If, in addition, the function α is non-decreasing, then

$$u(t) \le \alpha(t) e^{\int_a^t \beta(r) dr}, \quad t \in I.$$

Lemma 2.3.2 ([4], [18] **Doob's Inequality**). $M = (M_t, t > 0)$ is a continuous martingale on a probability space (Ω, P) . Set $M_t^* = \sup_{0 \le s \le t} |M_s|$. For any $p \in (1, \infty)$, the following inequality holds,

$$E(M_t^*)^p \le (\frac{p}{p-1})^p E(|M_t|^p).$$

For any $p \in (0, 1)$, the following inequality holds,

$$E(M_t^*)^p \le \frac{1}{1-p} (E(|M_t|))^p.$$

Lemma 2.3.3 ([18] Burkholder-Davis-Gundy Inequality). For any continuous martingale $\{X_t\}_{t\geq 0}$ with $X_0 = 0$, any stopping time τ and any 0 , the

following inequality holds,

$$c_p E(\langle X \rangle_{\tau}^{\frac{p}{2}}) \le E(\sup_{0 \le s \le \tau} |X_s|^p) \le C_p E(\langle X \rangle_{\tau}^{\frac{p}{2}}),$$

where the constants c_p and C_p only depend on the choice of p.

Assume D is a domain in \mathbb{R}^d with smooth boundary. Define the Sobolev space

$$W^{1,p}(D) := \{ f \in L^p(D) | \text{ weak derivative } \frac{\partial f}{\partial x_i} \in L^p(D), \quad i = 1, ..., d \},$$

equipped with the norm $||f||_{W^{1,p}} = \left(\int_D \left(|f|^p + |\nabla f|^p\right) dx\right)^{\frac{1}{p}}$.

Define the Hölder space: for $\gamma \in (0, 1)$,

$$C^{\gamma}(D) := \{ f | \sup_{x, y \in D; x \neq y} \frac{|f(x) - f(y)|}{|x - y|^{\gamma}} < \infty \},$$

with the Hölder coefficient $||f||_{C^{\gamma}} = \sup_{x,y \in D; x \neq y} \frac{|f(x) - f(y)|}{|x - y|^{\gamma}}.$

Lemma 2.3.4 ([20] Sobolev's Embedding Theorem). Let p > d and $\gamma = 1 - \frac{d}{p}$. Suppose $f \in W^{1,p}(D)$, then it holds that $f \in C^{\gamma}(D)$.

Moreover, there is a constant C > 0, such that for any $f \in W^{1,p}(D)$,

$$||f||_{C^{\gamma}} \le C ||f||_{W^{1,p}}.$$

Lemma 2.3.5 ([20] **Poincaré Inequality**). Assume that $1 \le p \le \infty$ and that D is a bounded connected open subset of the d-dimensional Euclidean space \mathbb{R}^d with a Lipschitz boundary. Then there exists a constant C, depending only on D and p, such that for every function u in the Sobolev space $W^{1,p}(D)$:

$$\|u - \bar{u}\|_{L^p} \le C \|\nabla u\|_{L^p},$$

where $\bar{u} = \frac{1}{m(D)} \int_D u(x) dx$ and m(D) is the Lebesgue measure of the domain D.

Chapter 3

Two-sided Estimates on the Heat Kernels

3.1 Introduction

Consider an elliptic operator as follows,

$$L = \frac{1}{2} \nabla \cdot (A\nabla) + B \cdot \nabla - \nabla \cdot (\hat{B} \cdot) + Q$$

$$= \frac{1}{2} \sum_{i,j=1}^{d} \frac{\partial}{\partial x_i} \left(a_{ij}(x) \frac{\partial}{\partial x_j} \right) + \sum_{i=1}^{d} B_i(x) \frac{\partial}{\partial x_i} - \sum_i \frac{\partial}{\partial x_i} (\hat{B}_i(x) \cdot) + Q(x) \quad (3.1)$$

in a d-dimensional smooth bounded Euclidean domain D.

 $A(\cdot) = (a_{ij})_{1 \le ij \le d}$: $R^d \to R^d \otimes R^d$ is a smooth, symmetric matrix-valued function and we assume that A is uniformly elliptic. That is, there is a constant $\lambda > 1$ such that

$$\frac{1}{\lambda} I_{d \times d} \le A(\cdot) \le \lambda I_{d \times d}. \tag{3.2}$$

Here $B = (B_1, ..., B_d)$ and $\hat{B} = (\hat{B}_1, ..., \hat{B}_d)$: $R^d \rightarrow R^d$ are Borel measurable

 R^d -valued functions, and Q is a Borel measurable function on R^d such that:

$$I_D(|B|^2 + |\hat{B}|^2 + |Q|) \in L^p(D)$$

for some p > d.

Satisfying the following mixed boundary condition

$$\frac{1}{2} \langle A\nabla u, \vec{n} \rangle - \langle \hat{B}, \vec{n} \rangle u = 0, \text{on}\partial D, \qquad (3.3)$$

the operator L determines a quadratic form :

$$\mathcal{Q}(u,v) = (-Lu,v) = \frac{1}{2} \sum_{i,j} \int_D a_{ij}(x) \frac{\partial u}{\partial x_i} \frac{\partial v}{\partial x_j} dx - \sum_i \int_D B_i(x) \frac{\partial u}{\partial x_i} v(x) dx - \sum_i \int_D \hat{B}_i(x) \frac{\partial v}{\partial x_i} u(x) dx - \int_D Q(x) u(x) v(x) dx,$$

where (\cdot, \cdot) stands for the inner product in $L^2(D)$ and \vec{n} denotes the inward normal vector to the boundary ∂D . For any $\xi \in \partial D$, $U(\xi)$ denotes a neighborhood of ξ . For every $\xi \in \partial D$ there exists a neighborhood $U(\xi)$ and a smooth function ψ such that $\partial D \cap U(\xi) = \{x : \psi(x_1, ..., x_d) = 0\}$, and $D \cap U(\xi) = \{x : \psi(x_1, ..., x_d) > 0\}$. Then \vec{n} has the following expression in this local coordinates: for $x \in \partial D \cap U(\xi)$,

$$\vec{n}(x) = \left(\frac{\partial \psi}{\partial x_1}(x), ..., \frac{\partial \psi}{\partial x_d}(x)\right) / \left(\sum_{i=1}^d \left(\frac{\partial \psi}{\partial x_i}(x)\right)^2\right)^{\frac{1}{2}}.$$

Set $\vec{\gamma}(x) = A(x)\vec{n}(x)$ and denote $\frac{\partial u}{\partial \vec{\gamma}} := \langle A\nabla u, \vec{n} \rangle$.

The domain of the quadratic form is

$$\mathcal{D}(\mathcal{E}) = W^{1,2}(D) := \{ u : u \in L^2(D), \frac{\partial u}{\partial x_i} \in L^2(D), i = 1, ..., d \}.$$

We use $\{T_t, t \ge 0\}$ to denote the semigroup generated by L, and we will prove in the following discussion that there exists a function l(t, x, y) which is the heat kernel associated with the semigroup T_t in the sense: $T_tg(x) = \int_D l(t, x, y)g(y)dy$. The purpose of this chapter is to provide both upper and lower bound estimates for the heat kernel l(t, x, y) associated with operator L equipped with the mixed boundary conditions (3.3). By a "time reverse" technique (see Section 3.2) introduced in [7], we transform the problem of estimating the heat kernels with mixed boundary condition associated with the general operator L in (3.1) to a problem of estimating the fundamental solution p(t, x, y) of the following simpler problem:

$$\begin{cases} \frac{\partial u}{\partial t} = Gu & x \in D, \\ < A \nabla u, \vec{n} >= 0 & x \in \partial D. \end{cases}$$
(3.4)

Here

$$G = \frac{1}{2}\nabla \cdot (A\nabla) + b \cdot \nabla + q$$

for an appropriate vector field b and a function q. The precise expressions of b and q will be given in Section 3.2.

For the upper bound, we use parametrix and perturbation methods. For the lower bound, we need to assume that the domain is convex. Our method is inspired by the one in [3].

This chapter is organized as follows. In Section 3.2, some preliminary results are proved. The reduction of estimating the heat kernel associated with the operator Lto the estimate of the heat kernel associated with the operator G is explained. In Section 3.3, we obtain the upper bound for the heat kernel $p_2(t, x, y)$ associated with the operator

$$G_2 = \frac{1}{2}\nabla \cdot (A\nabla) + b \cdot \nabla$$

The lower bound for $p_2(t, x, y)$ is given in Section 3.4. Finally, the two sided estimates of the heat kernel associated with the general operator L are proved in Section 3.5.

3.2 A Reduction Method

Consider the following regular Dirichlet form

$$\begin{cases} \mathcal{E}_0(u,v) = \frac{1}{2} \sum_{i,j} \int_D a_{ij}(x) \frac{\partial u}{\partial x_i} \frac{\partial v}{\partial x_j} dx, \\ D(\mathcal{E}_0) = W^{1,2}(D). \end{cases}$$

Denoting the associated reflecting diffusion process by $(\Omega, \mathcal{F}_t, X_t, \theta_t, \gamma_t, P^x)$ by the discussion in Example 1, Section 2.1, we know the following decomposition holds:

$$X_t = X_0 + \int_0^t \sigma(X_s) \, dB_s + \frac{1}{2} \int_0^t \nabla A(X_s) \, ds + \frac{1}{2} \int_0^t A\vec{n}(X_s) \, dL_s. \quad P^x - a.s. \quad (3.5)$$

Here the square integrable martingale $M_t := \int_0^t \sigma(X_s) dB_s$ has the property:

$$\langle M^{i}, M^{j} \rangle_{t} = \int_{0}^{t} a_{ij}(X_{s}) \, ds.$$
 (3.6)

The following probabilistic representation of the semigroup $\{T_t\}_{t\geq 0}$ associated with the operator L was proved in [6]

$$T_t f(x) = E^x [f(X_t) \exp(\int_0^t (A^{-1}B)^* (X_s) dM_s + (\int_0^t (A^{-1}\hat{B})^* (X_s) dM_s) \circ \gamma_t - \frac{1}{2} \int_0^t (B - \hat{B})^* A^{-1} (B - \hat{B}) (X_s) ds + \int_0^t Q(X_s) ds)],$$

where B, \hat{B} and Q are the coefficients of operator L in (3.1).

Here E^x denotes the expectation under P^x and we denote by x^* the transpose of the vector x.

The following result plays an important role in the thesis. The proof will be given after some preparations.

Proposition 3.2.1. Let $f = (f_1, ..., f_d)$ be a vector-valued function defined on the

domain D satisfying that $|f| \in L^p(D)$ for p > d. If $u \in W^{1,2}(D)$ satisfies

$$\int_D \sum_{ij} a_{ij} \frac{\partial u}{\partial x_i} \frac{\partial \psi}{\partial x_i} dx = \int_D \sum_i f_i \frac{\partial \psi}{\partial x_i} dx.$$

for any function $\psi \in W^{1,2}(D)$, then $u \in W^{1,p}(D)$.

 Set

$$C^{\infty}_{\gamma}(D) := \{ \phi \in C^{\infty}(D) | \frac{\partial \phi}{\partial \gamma} = 0 \quad on \quad \partial D \},$$

where $\gamma(x) = A(x)\vec{n}(x)$.

Let two positive numbers q and q satisfying $\frac{1}{q} + \frac{1}{p} = 1$. $W^{-1,p}(D)$ denotes the dual space of $W^{1,q}(D)$.

Remark 1. Let f be the function as in Proposition 3.2.1, then we know $div(f) \in W^{-1,p}(D)$.

In fact, for any $\phi \in W^{1,q}(D)$,

$$|\int_{D} \langle f, \nabla \phi \rangle (x) dx| \leq ||f||_{L^{p}} \cdot ||\nabla \phi||_{L^{q}(D)} \leq ||f||_{L^{p}} \cdot ||\phi||_{W^{1,q}(D)},$$

which implies that

$$_{W^{-1,p}(D)} < div(f), \phi >_{W^{1,q}(D)} = \int_{D} < f, \nabla \phi > (x) dx.$$

Moreover, we have $\|div(f)\|_{W^{-1,p}(D)} \le \|f\|_{L^p}$.

For the uniformly elliptic diffusion matrix $A = (a_{ij})$, we construct a matrix $\tilde{A} = (\tilde{a}_{ij})$ whose inverse matrix $\tilde{A}^{-1} = (\tilde{a}^{ij})$ satisfies $\sqrt{\det \tilde{A}^{-1}}\tilde{A} = A$, where $\det \tilde{A}^{-1}$ is the determinant of the matrix \tilde{A}^{-1} . Denote $a = \det \tilde{A}^{-1}$.

In fact, by calculation, we know that if we set $\tilde{A} = (det(A^{-1}))^{\frac{1}{d-2}}A$, then $det(\tilde{A}^{-1}) = (detA^{-1})^{-\frac{2}{d-2}}$. Therefore, $\sqrt{det\tilde{A}^{-1}}\tilde{A} = A$ satisfies.

We know that \hat{A} forms a symmetric strictly positive definite covariant tensor of order 2. (D, \tilde{A}) can be seen as a compact Riemannian manifold with the global Euclidean coordinate system. The volume form dV on D is $dV = \sqrt{a}dx$. As usual, the metric tensor \tilde{A} , in Euclidean coordinates $(x_1, \dots x_d)$ has the following representation

$$< \frac{\partial}{\partial x_i}, \frac{\partial}{\partial x_j} >_m = \tilde{a}^{ij},$$

where \langle , \rangle_m stands for the inner product of the tangent vectors in (D, \tilde{A}) .

The gradient vector of a function f is

$$grad_m f = \left(\left(\sum_j \tilde{a}_{1j} \frac{\partial f}{\partial x_j} \right), ..., \left(\sum_j \tilde{a}_{dj} \frac{\partial f}{\partial x_j} \right) \right).$$

Let $X = (X_1, ..., X_d)$ be a smooth vector field on (D, \tilde{A}) . The divergence of X is

$$div_m(X) = \frac{1}{\sqrt{a}} \sum_i \frac{\partial}{\partial x_i} (\sqrt{a}X_i),$$

and the Laplace operator is given by

$$\Delta_m f := div_m grad_m f = \frac{1}{\sqrt{a}} \sum_{ij} \frac{\partial}{\partial x_j} (\tilde{a}_{ij} \sqrt{a} \frac{\partial f}{\partial x_i}).$$

So far using the above discussion, we know the following relationship between the integral on the manifold (D, \tilde{A}) with respect to the measure dV and the integral on the Euclidean domain D with respect to the Lebesgue measure dx under the global Euclidean coordination system ,

$$\int_{D} \triangle_{m} f(x)g(x) \, dV(x) = \int_{D} \nabla (A\nabla f)(x)g(x) \, dx.$$

Therefore, with the above Riemannian structure, we are able to apply Theorem 2.3' [39] to obtain the following lemma.

Lemma 3.2.2. Let $g \in W^{-1,p}(D)$. Suppose that $v \in L^2(D)$ satisfies

$$\int_{D} \langle v, \nabla(A\nabla\phi) \rangle(x) \, dx = {}_{W^{-1,p}(D)} \langle g, \phi \rangle_{W^{1,p}(D)}, \tag{3.7}$$

for any $\phi \in C^{\infty}_{\gamma}(D)$. Then $v \in W^{1,p}(D)$ and moreover, there exists a constant C such that,

$$\|v\|_{W^{1,p}(D)} \le C(\|g\|_{W^{-1,p}(D)} + |\int_D v(x) \, dx|).$$

Proof of Proposition 3.2.1:

Proof. (1) Firstly we prove that, if $u \in W^{1,2}(D)$ satisfies that

$$\int_D \sum_{ij} \langle A\nabla u, \nabla \psi \rangle \, dx = \int_D \sum_i \langle f, \nabla \psi \rangle \, dx,$$

for any function $\psi \in W^{1,2}(D)$, and $\int_D u(x) dx = 0$, then $u \in W^{1,p}(D)$ and moreover, there exists a constant $C_1 > 0$ such that

$$\|\nabla u\|_{L^p} \le C_1 \|f\|_{L^p}.$$

There exists $f^k = (f_1^k, ..., f_d^k)$ such that $f_i^k \in C_0^{\infty}(D), i = 1, ..., d$ and

$$\lim_{k \to \infty} \|f^k - f\|_{L^p} = 0.$$

Note that in this case, $div(f^k) = \sum_i \frac{\partial f_i^k}{\partial x_i} \in L^p(D)$. By [39], for every k, there exists $v_k \in W^{1,2}(D)$, $\int_D v_k(x) dx = 0$ such that

$$\int_D \sum_{ij} a_{ij} \frac{\partial v_k}{\partial x_i} \frac{\partial \psi}{\partial x_i} \, dx = \int_D \sum_i f_i^k \frac{\partial \psi}{\partial x_i} \, dx,$$

for $\psi \in W^{1,2}(D)$.

By Green's identity, and the fact that $f^k = 0$ on $\partial D = 0$, we see that v_k satisfies the formula (3.7) with $g = div f^k$, i.e.

$$\int_{D} v_k(x) \nabla (A \nabla \phi)(x) dx = \int_{D} div(f_k) \phi(x) dx$$
(3.8)

for every $\phi \in C^{\infty}_{\gamma}(D)$.

Therefore, v_k satisfies the conditions in Lemma 3.2.2, so we know that $v_k \in W^{1,p}(D)$ and $\|\nabla v_k\|_{W^{1,p}(D)} \leq C \|div(f^k)\|_{W^{-1,p}(D)} \leq C \|f^k\|_{L^p}$.

Set positive number p', such that $\frac{1}{p'} + \frac{1}{p} = 1$. So p' < 2 < p implies that $L^p(D) \subset L^2(D) \subset L^{p'}(D)$.

For any $\phi \in W^{1,2}(D)$,

$$\begin{aligned} |\int_{D} \langle f^{k} - f, \nabla \phi \rangle (x) \, dx| &\leq \int_{D} |f^{k} - f| \cdot |\nabla \phi| \, dx \\ &\leq \|f^{k} - f\|_{L^{p}} \cdot \|\nabla \phi\|_{L^{p'}} \\ &\leq C_{1} \|f^{k} - f\|_{L^{p}} \cdot \|\nabla \phi\|_{L^{2}}. \end{aligned}$$

So that

$$\lim_{k \to \infty} \int_D < f^k, \nabla \phi > dx = \int_D < f, \nabla \phi > dx$$

It follows that

$$\lim_{k \to \infty} \int_D \langle A\nabla v_k, \nabla \phi \rangle dx = \lim_{k \to \infty} \int_D \langle f^k, \nabla \phi \rangle dx$$
$$= \int_D \langle f, \nabla \phi \rangle dx = \int_D \langle A\nabla u, \nabla \phi \rangle dx.$$

On the other hand, we know that $\|\nabla v_k - \nabla v_{k'}\|_{L^p} \leq C \|f^k - f^{k'}\|_{L^p}$.

By the Poincaré inequality and the fact that $\int_D v_l dx = 0$, for $l \ge 1$, it follows that

$$\|v_k - v_{k'}\|_{L^p} \le M \|\nabla v_k - \nabla v_{k'}\|_{L^p} \le CM \|f^k - f^{k'}\|_{L^p}.$$

So that $\{v_k\}_{k=1}^{\infty}$ is a Cauchy sequence in $W^{1,p}(D)$. Therefore, there exists $\bar{v} \in W^{1,p}(D)$, such that $\|v_k - \bar{v}\|_{W^{1,p}(D)} \to 0$ as $k \to \infty$.

For any $\phi \in W^{1,2}(D)$, it follows that

$$\lim_{k \to \infty} \int_D \langle A\nabla v_k, \nabla \phi \rangle \, dx = \int_D \langle A\nabla \bar{v}, \nabla \phi \rangle \, dx, \quad \forall \phi \in W^{1,2}(D),$$

which implies that

$$\int_D < A\nabla(\bar{v} - u), \nabla\phi > dx = 0.$$

Then $\bar{v} - u$ satisfies Lemma 3.2.2 with $g \equiv 0$. Hence $\bar{v} - u \in W^{1,p}(D)$ and $\|\bar{v} - u\|_{W^{1,p}(D)} = 0.$

(2) In the general case, setting $\bar{u} = u - \int_D u(x) dx$, we know that the function \bar{u} satisfies the conditions in part (1). Therefore, we find that $\bar{u} \in W^{1,p}(D)$. Then the proposition is proved.

As a conclusion of this section, a reduction method is given as follows.

From [7], it follows that there exists a function $v \in \mathcal{D}(\mathcal{E}_0)$ satisfying

$$\left(\int_{0}^{t} (A^{-1}\hat{B})^{*}(X_{s}) dM_{s}\right) \circ \gamma_{t}$$

= $-\int_{0}^{t} \nabla v(X_{s}) dM_{s} + v(X_{t}) - v(X_{0}) - \int_{0}^{t} (A^{-1}\hat{B})^{*}(X_{s}) dM_{s},$ (3.9)

and moreover,

$$\begin{cases} div(A\nabla v) = -div(\hat{B}) & \text{on } D, \\ \frac{\partial v}{\partial \gamma} = -2 < \hat{B}, \vec{n} > & \text{on } \partial D. \end{cases}$$
(3.10)

Therefore, by Proposition 3.2.1, we know that $v \in W^{1,p}(D)$. In particular, by Sobolev's embedding theorem, v is bounded and continuous.

Thus the representation of T_t becomes:

$$T_{t}f(x) = e^{-v(x)}E^{x}[f(X_{t})e^{v(X_{t})}\exp(\int_{0}^{t}(A^{-1}(B-\hat{B}-A\nabla v))^{*}dM_{s} -\frac{1}{2}\int_{0}^{t}(B-\hat{B}-A\nabla v)^{*}A^{-1}(B-\hat{B}-A\nabla v)(X_{s})ds +\int_{0}^{t}(Q+\frac{1}{2}(\nabla v)^{*}A(\nabla v)-\langle B-\hat{B},\nabla v\rangle)(X_{s})ds)]$$

$$= e^{-v(x)}S_{t}[fe^{v}](x)$$
(3.11)

Here, S_t is the semigroup generated by the following operator equipped with the

boundary condition $\langle A\nabla u, \vec{n} \rangle = 0, x \in \partial D$:

$$G = \frac{1}{2}\nabla \cdot (A\nabla) + (B - \hat{B} - (A\nabla v)) \cdot \nabla + (Q + \frac{1}{2}(\nabla v)A(\nabla v)^* - \langle B - \hat{B}, \nabla v \rangle)$$

$$= \frac{1}{2}\nabla \cdot (A\nabla) + b \cdot \nabla + q. \qquad (3.12)$$

Here we set $b = B - \hat{B} - (A\nabla v)$ and $q = Q + \frac{1}{2}(\nabla v)A(\nabla v)^* - \langle B - \hat{B}, \nabla v \rangle$.

In the following discussion, we will construct the heat kernel denoted by p(t, x, y)and associated with the semigroup S_t . For any $f \in C^{\infty}(D)$, we have

$$T_t f(x) = e^{-v(x)} \int_D e^{v(y)} p(t, x, y) f(y) dy.$$
(3.13)

Hence

$$l(t, x, y) = p(t, x, y)e^{v(y) - v(x)}.$$
(3.14)

Thus, the estimates of l(t, x, y) will follow from that of p(t, x, y). Due to (3.10), it is easy to see that the corresponding boundary condition also holds:

$$\frac{1}{2}\frac{\partial l}{\partial \gamma}(t,x,y)-l(t,x,y)<\hat{B},\vec{n}>=0,\quad on\quad \partial D.$$

The rest of the chapter will be devoted to the estimates on p(t, x, y).

3.3 Upper Bound Estimates

In this section, we consider the operator of the following form,

$$G_2 = \frac{1}{2}\nabla \cdot (A\nabla) + b \cdot \nabla.$$

Let $p_2(t, x, y)$ denote the heat kernel associated with G_2 on D equipped with the boundary condition $\langle A\nabla u, \vec{n} \rangle = 0$. We aim to establish an upper estimate for $p_2(t, x, y)$. To this end, we first consider the heat kernel $p_1(t, x, y)$ associated with G_1 :

$$G_1 = \frac{1}{2} \nabla \cdot (A \nabla).$$

3.3.1 Upper Bounds for Heat Kernels Associated with G_1

Local coordinate transformations will be used in this section to deal with the boundary of the domain. Rewrite the operator G_1 in the following form:

$$G_{1} = \frac{1}{2} \sum_{i=1}^{d} \frac{\partial}{\partial x_{i}} \left(\sum_{j=1}^{d} a_{ij} \frac{\partial}{\partial x_{j}}\right)$$
$$= \frac{1}{2} \sum_{i,j=1}^{d} a_{ij} \frac{\partial^{2}}{\partial x_{i} \partial x_{j}} + \frac{1}{2} \sum_{j=1}^{d} \left(\sum_{i=1}^{d} \frac{\partial a_{ij}}{\partial x_{i}}\right) \frac{\partial}{\partial x_{j}}.$$

Note that the expression of G_1 depends on the choice of the coordinate system $x = (x_1, ..., x_d)$. For convenience, we denote the global Euclidean coordinate mapping by $\sigma_0 : D \to R^d$ with $\sigma_0(x) = (x_1, ..., x_d)$.

If we consider another coordinate system $\sigma(x) = (\bar{x}_1, ..., \bar{x}_d) : D \to \mathbb{R}^n$ with $\bar{x}_i = \bar{x}_i(x_1, ..., x_d), i = 1, ..., d$, being smooth, then an easy calculation yields that

$$G_{1} = \frac{1}{2} \sum_{k,l=1}^{d} \left(\sum_{i,j=1}^{d} a_{ij} \frac{\partial \bar{x}_{k}}{\partial x_{i}} \frac{\partial \bar{x}_{l}}{\partial x_{j}}\right) \frac{\partial^{2}}{\partial \bar{x}_{l} \partial \bar{x}_{k}} + \frac{1}{2} \sum_{k=1}^{d} \left(\sum_{l=1}^{d} \frac{\partial \partial \bar{x}_{l}}{\partial \bar{x}_{l}} \left(\sum_{i,j=1}^{d} a_{ij} \frac{\partial \bar{x}_{k}}{\partial x_{i}} \frac{\partial \bar{x}_{l}}{\partial x_{j}}\right)\right) \frac{\partial \bar{x}_{k}}{\partial \bar{x}_{k}}$$
$$= \frac{1}{2} \sum_{k,l=1}^{d} \bar{a}_{kl} \frac{\partial^{2}}{\partial \bar{x}_{k} \partial \bar{x}_{l}} + \frac{1}{2} \sum_{k=1}^{d} \left(\sum_{l=1}^{d} \frac{\partial \bar{a}_{kl}}{\partial \bar{x}_{l}}\right) \frac{\partial \bar{x}_{k}}{\partial \bar{x}_{k}}.$$
(3.15)

here $\bar{a}_{kl} = \sum_{i,j=1}^{d} a_{ij} \frac{\partial \bar{x_k}}{\partial x_i} \frac{\partial \bar{x_l}}{\partial x_j}$.

Therefore G_1 under the new coordinate system $\sigma(x) = (\bar{x}_1, ..., \bar{x}_d)$ has the same form as under the Euclidean coordinate system σ_0 with a diffusion matrix $\bar{A} = (\bar{a}_{kl})_{1 \leq k, l \leq d}$.

Based on this observation, we see that the diffusion matrix $(a_{ij})_{1 \le i,j \le d}$ is transformed between two local coordinates in the following way. Suppose that $U_1(\xi)$ and $U_2(\xi)$ are two neighborhoods of the point $\xi \in \overline{D}$ and the mappings $\sigma_1 : U_1(\xi) \to \mathbb{R}^d$ and $\sigma_2 : U_2(\xi) \to \mathbb{R}^d$ are the coordinate systems on $U_1(\xi)$ and $U_2(\xi)$ respectively. For any $z \in U_1(\xi) \bigcap U_2(\xi) \bigcap \overline{D}$ with coordinates $\sigma_1(z) = (z_1, ..., z_d)$ and $\sigma_2(z) = (\overline{z}_1, ..., \overline{z}_d)$, we have

$$\bar{a}_{ij} = \sum_{kl} \frac{\partial \bar{z}_i}{\partial z_k} \frac{\partial \bar{z}_j}{\partial z_l} a_{kl}$$
(3.16)

here $\bar{a}_{ij}(z)$ and $a_{ij}(\bar{z})$ denote the diffusion matrix associated with the coordinate systems σ_2 and σ_1 respectively.

The Neumann boundary condition

$$\frac{\partial f}{\partial \gamma} := = 0, \quad \text{on} \quad \partial D$$

which is described precisely in Section 3.1, actually has different expressions under different local coordinate systems. Suppose that $U(\xi_0)$ is a neighborhood of the point $\xi_0 \in \partial D$ with coordinate mapping $\sigma(\xi) = (\xi_1, ..., \xi_d)$ for $\xi \in U(\xi_0) \cap D$. Recall that there exists a smooth function ϕ such that $U(\xi_0) \cap \partial D = \{\xi, \phi(\xi_1, ..., \xi_d) = 0\}$ and $U(\xi_0) \cap D = \{\xi, \phi(\xi_1, ..., \xi_d) > 0\}.$

Set

$$\gamma(\xi) := (\sum_{i} a_{1i}(\xi) \frac{\partial \phi(\xi)}{\partial \xi_i}, ..., \sum_{i} a_{di}(\xi) \frac{\partial \phi(\xi)}{\partial \xi_i}) / (\sum_{ij} a_{ij} \frac{\partial \phi(\xi)}{\partial \xi_i} \frac{\partial \phi(\xi)}{\partial \xi_j})^{\frac{1}{2}}$$

and

$$\frac{\partial f}{\partial \gamma}(\xi) := \sum_{ij} a_{ij}(\xi) \frac{\partial \phi(\xi)}{\partial \xi_i} \frac{\partial f}{\partial \xi_j} / (\sum_{ij} a_{ij} \frac{\partial \phi(\xi)}{\partial \xi_i} \frac{\partial \phi(\xi)}{\partial \xi_j})^{\frac{1}{2}}$$

where f is the smooth function on \overline{D} .

The operator G_1 satisfying the Neumann boundary condition means that for a function u in the domain of G_1 , u must satisfy $\frac{\partial u}{\partial \gamma}(\xi) = 0$ for $\xi \in \partial D \cap U(\xi_0)$.

 $A^{-1}(x) = (a^{ij}(x))_{ij}$ denotes the inverse matrix of A(x), which forms a symmetric strictly positive definite covariant tensor of order 2. A^{-1} is used to define the length of a piecewise C^1 smooth curve as follows.

Suppose a curve C is defined by $z : \theta \in [0, 1] \to z(\theta) \in \overline{D}$. Then the length of C, calculated in the coordinate system $z = (z_1, ..., z_d)$, is

$$L(C) = \int_{[0,1]} \left(\sum_{ij} a^{ij}(z(\theta)) \frac{dz_i}{d\theta} \frac{dz_j}{d\theta}\right)^{\frac{1}{2}} d\theta.$$

Remark 2. Set $a = \sqrt{det A^{-1}}$, where $det A^{-1}$ is the determinant of the matrix A^{-1} . By the transformation (3.16), it is easy to verify that the length of the curve is independent of the choice of the local coordinate system.

Define the distance between $x, y \in \overline{D}$, d(x, y), as the infimum of the length of all smooth curves contained in \overline{D} which connect x and y. And write $d(x, \partial D) = \inf_{y \in \partial D} d(x, y)$.

Lemma 3.3.1. Fix $\xi \in \overline{D}$. Let $U(\xi)$ be a convex neighborhood of ξ with coordinate system $\sigma(z) = (z_1, z_2, ..., z_d)$ for $z \in U(\xi) \cap \overline{D}$. Then there exist two constants $K_1, K_2 > 0$, such that for $z^1, z^2 \in U(\xi) \cap \overline{D}$

$$K_2\left(\sum_{i=1}^d |z_i^1 - z_i^2|^2\right)^{\frac{1}{2}} \le d(z^1, z^2) \le K_1\left(\sum_{i=1}^d |z_i^1 - z_i^2|^2\right)^{\frac{1}{2}}$$
(3.17)

Proof. (1) Let C be a C^1 piecewise smooth curve, given by :

$$C: \theta \in [0,1] \to c(\theta) \in \overline{D}$$

such that $c(0) = z^1$ and $c(1) = z^2$. Then we obtain

$$l(C) = \int_{[0,1]} (\sum_{i,j} a^{ij}(c(\theta)) \frac{dc_i(\theta)}{d\theta} \frac{dc_j(\theta)}{d\theta})^{\frac{1}{2}} d\theta \ge \frac{1}{K} \sum_i \int_0^1 |\frac{dc_i(\theta)}{d\theta}| d\theta \ge \frac{1}{K} |z_i^1 - z_i^2|,$$

where the first inequality comes from the uniformly ellipticity of the matrix $(a^{ij})_{1 \le i,j \le d}$.

This implies

$$l(C) \ge \frac{1}{K\sqrt{d}} \left(\sum_{i=1}^{d} |z_i^1 - z_i^2|^2\right)^{\frac{1}{2}}.$$

As C is arbitrary, we have $d(z^1, z^2) \ge \frac{1}{K\sqrt{d}} (\sum_{i=1}^d |z_i^1 - z_i^2|^2)^{\frac{1}{2}}$.

Here, the constant K is independent of the choice of z but dependent on the choice of the local coordinate system. Setting $K_2 = \frac{1}{K\sqrt{d}}$, we have proved the first half of (3.17).

(2) Define a curve P from z^1 to z^2 by

$$P: \theta \in [0,1] \to p(\theta) = \sigma^{-1}(z_1^1 + \theta(z_1^2 - z_1^1), ..., z_d^1 + \theta(z_d^2 - z_d^1))$$

then we get that

$$d(z^{1}, z^{2}) \leq l(P) = \int_{0}^{1} (\sum_{i,j} a^{ij}(p(\theta))(z_{i}^{2} - z_{i}^{1})(z_{j}^{2} - z_{j}^{1}))^{\frac{1}{2}} d\theta \leq K_{1} (\sum_{i=1}^{d} |z_{i}^{2} - z_{i}^{1}|^{2})^{\frac{1}{2}}$$

Remark 3. Because of this Lemma, for two neighborhoods $U_1(\xi), U_2(\xi)$ of ξ with coordinate systems $\sigma_1(z) = (z_1, \dots, z_d), \sigma_2(z) = (\bar{z}_1, \dots, \bar{z}_d)$ respectively, there are positive numbers C_1, C_2 , such that $C_2(\sum_{i=1}^d |\bar{z}_i^1 - \bar{z}_i^2|^2)^{\frac{1}{2}} \leq (\sum_{i=1}^d |z_i^1 - z_i^2|^2)^{\frac{1}{2}} \leq C_1(\sum_{i=1}^d |\bar{z}_i^1 - \bar{z}_i^2|^2)^{\frac{1}{2}}$, for $z^1, z^2 \in U_1(\xi) \cap U_2(\xi)$.

The following result is from [22].

Lemma 3.3.2. Fix any $x_0 \in \partial D$, there is a neighborhood U_0 of x_0 with coordinate system $(\bar{x}_1, \dots, \bar{x}_d)$, such that (1) $\partial D \cap U_0 = \{x : \bar{x}_d = 0, x \in \bar{D} \cap U_0\}, D \cap U_0 = \{x : \bar{x}_d > 0, x \in \bar{D} \cap U_0\}.$ (2) For $x \in U_0 \cap \partial D$, $\bar{a}_{id}(x) = \bar{a}_{di}(x) = \delta_{id}$. Here $\{\bar{a}_{ij}\}$ denotes the diffusion matrix A associated with the local coordinate (\bar{x}_i) and δ_{ij} denotes Kronecker's delta.

From now on we call this coordinate system the canonical coordinate neighborhood of $x_0 \in \partial D$.

Theorem 3.3.3. There exist a positive constant T_1 and a function $p_1(t, x, y)$ defined

on $[0, T_1] \times \overline{D} \times \overline{D}$ which solves the following equation:

$$\begin{cases} \frac{\partial p_1}{\partial t}(t, x, y) = G_1 p_1(t, x, y) & (t, x, y) \in (0, T_1] \times D \times D, \\ \frac{\partial p_1}{\partial \gamma}(t, x, y) = 0 & x \in \partial D. \end{cases}$$
(3.18)

And moreover $p_1(t, x, y)$ admits an upper bound of Gaussian type:

$$p_1(t, x, y) \le C_1 t^{-\frac{d}{2}} \exp(-C_2 t^{-1} |x - y|^2), \quad t \le T_1,$$
(3.19)

here C_1 and C_2 are positive constants.

Proof. From [34], we can choose a finite number of canonical coordinate neighborhoods U_i , $1 \leq i \leq M$, open subsets B_{ij} , $1 \leq j \leq M_i$, of U_i and non-negative functions λ_{ij} in $C^2(\bar{D})$ with supports contained in B_{ij} , satisfying the following conditions: $\{B_{ij}, 1 \leq i \leq M, 1 \leq j \leq M_i\}$ is a covering of \bar{D} ; if B_{ij} intersects $B_{i'j'}$, then $\bar{B}_{i'j'} \subset U_i$; $\sum_{ij} \lambda_{ij}(x)^2 = 1$ for $x \in \bar{D}$; $\frac{\partial \lambda_{ij}}{\partial \bar{n}}(x) = 0$ for $x \in \partial D$.

Suppose that U_i contains boundary points for $1 \le i \le i_0$, while U_i $(i_0+1 \le i \le M)$ not. Let $\sigma_i(x) = (x_{(i)}^1, \dots, x_{(i)}^d)$ be the canonical coordinate system in $U_i(1 \le i \le i_0)$. We will use the Euclidean coordinate system $\sigma_0(x) = (x_1, \dots, x_d)$ in U_i $(i_0+1 \le i \le M)$.

From the appendix in [34], we know that there is a smooth function q(t, x, y)defined on $[0, T] \times \overline{D} \times \overline{D}$, where T is an arbitrarily fixed positive number, satisfying the reflecting boundary condition $\frac{\partial q}{\partial \gamma}(t, x, y) = 0$, $x \in \partial D$ and $\lim_{t\to 0} q(t, x, y) = \delta_x(y)$. Moreover, q satisfies the following upper bound: for some constants $K_1, K_2 > 0$, $x, y \in \overline{D}$:

$$|q(t,x,y)| \le K_1 \sum_{ij} \lambda_{ij}(x) \lambda_{ij}(y) t^{-\frac{d}{2}} \exp(-K_2 t^{-1} \sum_{k=1}^d |x_{(i)}^k - y_{(i)}^k|^2), \quad x,y \in \bar{D}.$$

By the Remark 3, there is a constant C > 0 independent of x and coordinate neighborhood U_i such that $\sum_{k=1}^d |x_{(i)}^k - y_{(i)}^k|^2 \ge C \sum_{k=1}^d |x_k - y_k|^2$, where the right side denotes the corresponding Euclidean coordinate. And because $|\lambda_{ij}| \leq 1$, we get

$$|q(t,x,y)| \le M_1 t^{-\frac{d}{2}} \exp(-M_2 t^{-1} \sum_{k=1}^d |x_k - y_k|^2)$$
(3.20)

where $M_1, M_2 > 0$ are constants.

Let f(t, x, y) be a solution of the following integral equation

$$f(t,x,y) = (G_1 - \frac{\partial}{\partial t})q(t,x,y) + \int_0^t ds \int_{\bar{D}} (G_1 - \frac{\partial}{\partial t})q(t-s,x,z)f(s,z,y) dz$$

where dz denotes the Lebesgue measure on the domain D. This is an integral equation of Volterra type. We will follow the method in [22] and [21] to solve this equation by the method of iteration in the following discussion.

Define $p_1(t, x, y)$ by

$$p_1(t, x, y) = q(t, x, y) + \int_0^t ds \int_{\bar{D}} q(t - s, x, z) f(s, z, y) \, dz.$$
(3.21)

It is easy to verify that $p_1(t, x, y)$ satisfies the equation (3.18).

To obtain the upper bound of $p_1(t, x, y)$, we first establish an upper bound for f(t, x, y). For this, we write f(t, x, y) as a series.

 Set

$$e_0(t, x, y) = (G_1 - \frac{\partial}{\partial t})q(t, x, y),$$

$$e_{n+1}(t,x,y) = \int_0^t ds \int_{\bar{D}} e_0(t-s,x,z) e_n(s,z,y) \, dz.$$

Then the following equation holds if the series is convergent,

$$f(t, x, y) = \sum_{n=0}^{\infty} e_n(t, x, y)$$

In fact, by [34], there exists a constant M_3 , such that $e_0(t, x, y)$ can be chosen to satisfy

$$|e_0(t, x, y)| \le M_3 t^{-\frac{d+1}{2}} \exp(-M_2 t^{-1} \sum_{i,j=1}^d |x_i - y_i|^2).$$

Let $|x - y|^2 = \sum_{i=1}^d |x_i - y_i|^2$. Then it follows, for $t \in [0, T]$,

$$\begin{aligned} |e_{1}(t,x,y)| &\leq \int_{0}^{t} ds \int_{\bar{D}} |e_{0}(t-s,x,z)e_{0}(s,z,y)| dz \\ &\leq (M_{3})^{2} \int_{0}^{t} ds \int_{\bar{D}} (t-s)^{-\frac{d+1}{2}} \exp(-M_{2}(t-s)^{-1}|x-z|^{2})s^{-\frac{d+1}{2}} \\ &\times \exp(-M_{2}s^{-1}|z-y|^{2}) dz \\ &= (M_{3})^{2} (2\pi)^{d} \int_{0}^{t} \frac{ds}{(t-s)^{\frac{1}{2}}s^{\frac{1}{2}}} \int_{\bar{D}} \frac{\exp(-\frac{M_{2}|x-z|^{2}}{t-s})}{(2\pi(t-s))^{\frac{d}{2}}} \frac{\exp(-\frac{M_{2}|z-y|^{2}}{s})}{(2\pi(s))^{\frac{d}{2}}} dz \\ &\leq (M_{3})^{2} (2\pi)^{\frac{d}{2}} t^{-\frac{d}{2}} \exp(-\frac{M_{2}|x-y|^{2}}{t}) \sqrt{\frac{2}{t}} \int_{0}^{\frac{t}{2}} \frac{ds}{s^{\frac{1}{2}}} \\ &\leq (M_{3})^{2} (2\pi)^{\frac{d}{2}} 2\sqrt{T} t^{-\frac{d+1}{2}} \exp(-\frac{M_{2}|x-y|^{2}}{t}). \end{aligned}$$
(3.22)

Iterating this calculation, we get, for $t \in [0, T]$,

$$|e_k(t, x, y)| \le M_3^{k+1} (2\pi)^{k\frac{d}{2}} 2^k T^{\frac{k}{2}} t^{-\frac{d+1}{2}} \exp(-\frac{M_2|x-y|^2}{t}).$$

Therefore, there exist positive numbers T_1 and M_4 , such that, for any $t \in [0, T_1]$,

$$|f(t,x,y)| \le \sum_{n=0}^{\infty} |e_n(t,x,y)| \le M_4 t^{-\frac{d+1}{2}} \exp(-\frac{M_2|x-y|^2}{t}).$$
(3.23)

Combining (3.20), (3.21) and (3.23), we get:

$$\begin{aligned} |p_{1}(t,x,y)| &\leq M_{1}t^{-\frac{d}{2}}\exp(-M_{2}t^{-1}|x-y|^{2}) + \int_{0}^{t}ds\int_{\bar{D}}M_{1}(t-s)^{-\frac{d}{2}}\times\\ &\times\exp(-M_{2}(t-s)^{-1}|x-z|^{2})\times M_{4}s^{-\frac{d+1}{2}}\exp(-M_{2}s^{-1}|z-y|^{2})dz\\ &\leq M_{1}t^{-\frac{d}{2}}\exp(-M_{2}t^{-1}|x-y|^{2})\\ &+M_{1}M_{4}t^{-\frac{d}{2}}\exp(-M_{2}t^{-1}|x-y|^{2})(2\pi)^{\frac{d}{2}}\int_{0}^{t}\frac{ds}{\sqrt{s}}\\ &\leq M_{5}t^{-\frac{d}{2}}\exp(-M_{2}t^{-1}|x-y|^{2}), \end{aligned}$$
(3.24)

as $t \in [0, T_1]$. Here M_5 is a positive number, depending on M_1, M_4 and T_1 .

Lemma 3.3.4. Let $p_1(t, x, y)$ be defined as in Theorem 3.3.3. There exist two constants $\lambda_1, \lambda_2 > 0$ such that

$$|\nabla p_1| \le \lambda_1 t^{-\frac{d+1}{2}} \exp(-\lambda_2 t^{-1} |x - y|^2), t \in [0, T_1], \quad x, y \in D.$$
(3.25)

Proof. Let U_i, B_{ij} and λ_{ij} be the same as in the Theorem 3.3.3. By the proof of lemma 2.2 in [34], we have, for $x, y \in U_i$,

$$\left|\frac{\partial q}{\partial x_{(i)}^{l}}(t,x,y)\right| \le M_{6}t^{-\frac{d+1}{2}}\exp(-M_{7}t^{-1}\sum_{k=1}^{d}|y_{(i)}^{k}-x_{(i)}^{k}|^{2}).$$

Recall that coordinate system $\sigma_0(x) = (x_1, \dots, x_d)$ denotes the global Euclidean coordinate system. Assume $x \in U_i$. Because the mapping $\sigma_i \circ \sigma_0^{-1} : (x_1, \dots, x_d) \in$ $R^d \to (x_{(i)}^1, \dots, x_{(i)}^d) \in R^d$ is continuous and bounded, there exists a constant $K_3 > 0$ independent of x, k and j, such that

$$\frac{\partial q}{\partial x_{j}} = \left| \sum_{k=1}^{d} \frac{\partial q}{\partial x_{(i)}^{k}} \frac{\partial x_{(i)}^{k}}{x_{j}} \right| \\
\leq K_{3} \sum_{k} \left| \frac{\partial q}{\partial x_{(i)}^{k}} (t, x, y) \right| \\
\leq M_{8} t^{-\frac{d+1}{2}} \exp(-M_{7} t^{-1} |y - x|^{2}) \quad (3.26)$$

The last equality is due to the fact $\sum_{k=1}^{d} |y_{(i)}^k - x_{(j)}^k|^2 \leq C \sum_{k=1}^{d} |y_k - x_k|^2$. Thus there exist constants $M_9, M_{10} > 0$ such that

$$\begin{aligned} |\frac{\partial p_1}{\partial x_i}| &\leq |\frac{\partial q}{\partial x_i}| + \int_0^t \int_{\bar{D}} |\frac{\partial q(t-s,x,z)}{\partial x_i}| \cdot |f(s,z,y)| \, dz ds \\ &\leq M_8 t^{-\frac{d+1}{2}} \exp(-M_7 t^{-1} |y-x|^2) \\ &+ M_8 M_4 \int_0^t \frac{ds}{(t-s)^{\frac{1}{2}} s^{\frac{1}{2}}} \int_{\bar{D}} \frac{\exp(-\frac{M_7 |x-z|^2}{t-s})}{(t-s)^{\frac{d}{2}}} \frac{\exp(-\frac{M_2 |z-y|^2}{s})}{(s)^{\frac{d}{2}}} \, dz \\ &\leq M_9 t^{-\frac{d+1}{2}} \exp(-M_{10} t^{-1} |x-y|^2), \end{aligned}$$

for $t \in [0, T_1]$.

3.3.2 Upper Bounds for Heat Kernels Associated with G_2

Recall that $G_2 = G_1 + b \cdot \nabla$, where b was defined in (3.12). The following theorem is the main result of this section.

Theorem 3.3.5. There exist a constant $T_2 > 0$ and a continuous function $p_2(t, x, y)$, which is the heat kernel associated with the operator G_2 . $p_2(t, x, y)$ admits an upper bound of Gaussian type. That is, there exist some constants $C_1, C_2 > 0$, such that

$$|p_2(t, x, y)| \le C_1 t^{-\frac{d}{2}} \exp(-C_2 t^{-1} |x - y|^2)$$
(3.27)

for $t \in [0, T_2], x, y \in D$.

Let $B: D \to \mathbb{R}^d$ is a vector-valued measurable function. Set

$$N_h^{\alpha}(B) := \sup_{x \in D} \int_0^h \int_D |B(y)| s^{-\frac{d+1}{2}} \exp(-\alpha \frac{|x-y|^2}{s}) \, dy ds \tag{3.28}$$

Definition 3.3.6. We say that B satisfies condition K if

$$\lim_{h \to 0} N_h^{\alpha}(B) = 0, for \quad all \quad \alpha > 0.$$
(3.29)

Lemma 3.3.7. If $|B| \in L^p(D)$ (p > d), then B satisfies condition K.

Proof. For $|B| \in L^p(D)$ and p > d, we have |B| is of the Kato class K_{d+1} , i.e., $\lim_{\delta \to 0} \sup_{x \in D} \int_{\{|y-x| \le \delta\}} \frac{|B(y)|}{|y-x|^{d-1}} \, dy = 0.$ In fact, let q > 0, $\frac{1}{q} + \frac{1}{p} = 1$, then for any $\delta > 0$

$$\begin{split} & \int_{\{|y-x| \le \delta\}} \frac{|B(y)|}{|y-x|^{d-1}} \, dy \\ \le & (\int_D |B(y)|^p dy)^{\frac{1}{p}} (\int_{\{|y-x| \le \delta\}} \frac{1}{|y-x|^{(d-1)q}} \, dy)^{\frac{1}{q}} \\ \le & C \|B\|_{L^p} (\int_0^\delta \frac{1}{r^{(d-1)(q-1)}} \, dr)^{\frac{1}{q}}. \end{split}$$

Because d < p, we know that (d-1)(q-1) < 1. Then

$$\sup_{x} \int_{\{|y-x| \le \delta\}} \frac{|B(y)|}{|y-x|^{d-1}} \, dy \le \tilde{C} \|B\|_{L^{p}} \delta^{1-(d-1)(q-1)}.$$

That is, for any $\varepsilon > 0$, there is a $\delta \in (0, 1)$, such that

$$\sup_x \int_{\{|y-x| \le \delta\}} \frac{|B(y)|}{|y-x|^{d-1}} \, dy \le \varepsilon.$$

For the above δ ,

$$\leq \underbrace{\int_{D} \int_{0}^{h} |B(y)| s^{-\frac{d+1}{2}} \exp(-\alpha \frac{|x-y|^{2}}{s}) \, dy ds}_{I} \\ \leq \underbrace{\int_{\{|y-x| \le \delta\}} \cdots \, dy}_{I} + \underbrace{\int_{\{|y-x| \ge \delta\}} \cdots \, dy}_{II}.$$

First, we get

$$\begin{split} (I) &\leq \int_{\{|y-x| \leq \delta\}} |B(y)| \int_0^h s^{-\frac{d+1}{2}} \exp(-\alpha \frac{|x-y|^2}{s}) \, dy ds \\ &\leq \int_{\{|y-x| \leq \delta\}} |B(y)| \int_0^\infty l^{\frac{d+1}{2}-2} \exp(-\alpha |x-y|^2 l) \, dl dy \\ &\leq \int_{\{|y-x| \leq \delta\}} |B(y)| \frac{1}{\alpha^{\frac{d-1}{2}} |x-y|^{d-1}} \int_0^\infty l^{\frac{d-1}{2}-1} e^{-l} \, dl dy \\ &\leq C \int_{\{|y-x| \leq \delta\}} \frac{|B(y)|}{|y-x|^{d-1}} \, dy \leq C \varepsilon. \end{split}$$

By the Hölder's inequality, it follows that

$$(II) \leq \left(\int_{\{|y-x| \ge \delta\}} |B(y)|^p \left(\int_0^h \frac{\exp(-\alpha \frac{|x-y|^2}{s})}{s^{\frac{d+1}{2}}} \, ds \right) \, dy \right)^{\frac{1}{p}} \\ \left(\int_{\{|y-x| \ge \delta\}} \left(\int_0^h \frac{\exp(-\alpha \frac{|x-y|^2}{s})}{s^{\frac{d+1}{2}}} \, ds \right) \, dy \right)^{\frac{1}{q}}.$$

On the set $\{|y - x| \ge \delta\}$, we get

$$\begin{split} \int_{0}^{h} \frac{\exp(-\alpha \frac{|x-y|^{2}}{s})}{s^{\frac{d+1}{2}}} \, ds &\leq \int_{0}^{h} \frac{\exp(-\alpha \frac{\delta^{2}}{s})}{s^{\frac{d+1}{2}}} \, ds = \int_{\frac{1}{h}}^{\infty} e^{-\alpha \delta^{2}s} s^{\frac{d+1}{2}-1} s^{-1} \, ds \\ &\leq h \int_{0}^{\infty} e^{-\alpha \delta^{2}s} s^{\frac{d+1}{2}-1} \, ds = \frac{1}{\alpha^{\frac{d+1}{s}} \delta^{d+1}} \Gamma(\frac{d+1}{2}) h, \end{split}$$

which implies that

$$(II) \leq \left(\int_{\{|y-x| \ge \delta\}} |B(y)|^p \left(\frac{C_2}{\delta^{d+1}} h \right) dy \right)^{\frac{1}{p}} \left(\int_0^h \frac{1}{s^{\frac{1}{2}}} \left(\int \frac{\exp(-\alpha \frac{|x-y|^2}{s})}{s^{\frac{d}{2}}} dy \right) ds \right)^{\frac{1}{q}} \\ \leq \frac{C_3}{\delta^{\frac{d+1}{p}}} \|B\|_{L^p} h^{\frac{1}{p} + \frac{1}{2q}}.$$

Therefore, for arbitrary $\varepsilon_0 > 0$, there exists a positive number $\delta_0 > 0$ such that for any $h < \delta_0$, $N_h^{\alpha}(B) < \varepsilon$. That is $\lim_{h \to 0} N_h^{\alpha}(B) = 0$.

Lemma 3.3.8. ([41])

Suppose 0 < a < b, there exist positive constants $C_{a,b}$ and c depending only on a and b such that

$$\int_{0}^{t} \int_{D} \frac{\exp(-a(t-s)^{-1}|x-z|^{2})}{(t-s)^{\frac{d}{2}}} |B(z)| \frac{\exp(-bs^{-1}|z-y|^{2})}{s^{\frac{d+1}{2}}} dz ds$$

$$\leq C_{a,b} N_{h}^{c}(B) \frac{\exp(-at^{-1}|x-y|^{2})}{t^{\frac{d}{2}}} \tag{3.30}$$

and

$$\int_{0}^{t} \int_{D} \frac{\exp(-a(t-s)^{-1}|x-z|^{2})}{(t-s)^{\frac{d+1}{2}}} |B(z)| \frac{\exp(-bs^{-1}|z-y|^{2})}{s^{\frac{d+1}{2}}} dz ds$$

$$\leq C_{a,b} N_{h}^{c}(B) \frac{\exp(-at^{-1}|x-y|^{2})}{t^{\frac{d+1}{2}}} \tag{3.31}$$

for t < h.

Proof of Theorem 3.3.5:

By the standard parametrix method ([13]), the fundamental solution of the parabolic equation associated with G_2 has the following expression:

$$p_2(t, x, y) = p_1(t, x, y) + \int_0^t \int_D p_1(t - s, x, z) \Phi(s, z, y) \, dz ds,$$

where

$$\Phi(t,x,y) = b(x) \cdot \nabla_x p_1(t,x,y) + \int_0^t \int_D b(x) \cdot \nabla_x p_1(t-s,x,z) \Phi(s,z,y) \, dz \, ds$$

and $p_1(t, x, y)$ is as in section 3.1.

Let

$$f_1(t, x, y) = b(x) \cdot \nabla_x p_1(t, x, y),$$

$$f_{n+1}(t, x, y) = \int_0^t \int_D b(x) \cdot \nabla_x p_1(t - s, x, z) f_n(s, z, y) \, dz \, ds.$$

Then

$$\Phi(t, x, y) = \sum_{n=1}^{\infty} f_n(t, x, y).$$

By the upper estimates on p_1 , ∇p_1 and Lemma 3.3.8, we have

$$\begin{aligned} &|\int_{0}^{t} \int_{D} p_{1}(t-s,x,z) f_{1}(s,z,y) \, dz ds| \\ &= |\int_{0}^{t} \int_{D} p_{1}(t-s,x,z) b(z) \cdot \nabla_{z} p_{1}(s,z,y) \, dz ds| \\ &\leq M_{11} N_{h}^{\tilde{M}_{12}}(b) \frac{\exp(-M_{12}t^{-1}|x-y|^{2})}{t^{\frac{d}{2}}}. \end{aligned}$$
(3.32)

Let $g(t, x, y) = \int_0^t \int_D p_1(t - s, x, z) b(z) \cdot \nabla_z p_1(s, z, y) dz ds$. Therefore

$$\begin{split} &|\int_{0}^{t} \int_{D} p_{1}(t-s,x,z) f_{n+1}(s,z,y) \, dz ds| \\ &= |\int_{0}^{t} \int_{D} p_{1}(t-s,x,z) \{ \int_{0}^{s} \int_{D} b(z) \cdot \nabla_{z} p_{1}(s-l,z,\omega) f_{n}(l,\omega,y) m(d\omega) dl \} \, dz ds| \\ &= |\int_{0}^{t} \int_{D} \{ \int_{0}^{t-l} \int_{D} p_{1}(t-l-s,x,z) b(z) \cdot \nabla_{z} p_{1}(s,z,\omega) \, dz ds \} f_{n}(l,\omega,y) m(dw) \, dl | \\ &= |\int_{0}^{t} \int_{D} g(t-l,x,\omega) f_{n}(l,\omega,y) \, d\omega ds|. \end{split}$$

By the estimates (3.32) for g(t, x, y) and the Lemma 3.3.8, we get:

$$|\int_{0}^{t} \int_{D} p_{1}(t-s,x,z) f_{n+1}(s,z,y) \, dz \, ds| \\ \leq (N_{h}^{\tilde{M}_{12}}(b) M_{11})^{n} t^{-\frac{d}{2}} \exp(-M_{12}t^{-1}|x-y|^{2}).$$

Choosing h small enough, it follows that there exists $T_2 > 0$ such that

$$p_2(t, x, y) \le M_{13} t^{-\frac{d}{2}} \exp(-M_{12} t^{-1} |x - y|^2)$$
 on $[0, T_2] \times \bar{D} \times \bar{D}$.

3.4 Lower Bound Estimates

In this section, we establish the lower bound for the heat kernel of operator G_2 equipped with the Neumann boundary condition. Recall that $G_2 = \frac{1}{2} \nabla \cdot (A \nabla) + b \cdot \nabla$.

We first establish the lower bound for the heat kernel associated with the diffusion operator $G_1 = \frac{1}{2} \nabla \cdot (A \nabla)$.

3.4.1 Lower Bounds for Heat Kernels Associated with G_1

Recall that the reflecting diffusion $\{X_t, P^x\}_{t\geq 0}$ associated with the regular Dirichlet form $\mathcal{E}_0(\cdot, \cdot)$ has the generator $G_1 = \frac{1}{2}\nabla \cdot (A\nabla)$. And we have shown that the heat kernel $p_1(t, x, y)$ of the operator G_1 equipped with Neumann boundary condition, has upper bounds of Gaussian type:

$$p_1(t, x, y) \le C_1 t^{-\frac{d}{2}} \exp(-\frac{C_2 |x - y|^2}{t})$$

for $t \in [0, T_1]$ $x, y \in D$, where C_1, C_2 and T_1 are positive constants.

Proposition 3.4.1. There are positive numbers C_3 and C_4 such that, for $t \in [0, T_1]$, $x \in D$ and any $\varepsilon > 0$,

$$P^{x}[\sup_{s \le t} |X_{s} - x| \ge \varepsilon] \le C_{3} \exp\left(-\frac{C_{4}\varepsilon^{2}}{t}\right)$$
(3.33)

Proof. Note that

$$P^{x}[|X_{t} - x| \geq \varepsilon] = \int_{\{|x-y|\geq\varepsilon\}\cap D} p_{1}(t, x, y) \, dy$$

$$\leq C_{1}t^{-\frac{d}{2}} \int_{\{|y-x|\geq\varepsilon\}\cap D} e^{-\frac{C_{2}|x-y|^{2}}{t}} \, dy$$

$$\leq \widetilde{C_{1}} \exp(-\frac{C_{2}\varepsilon^{2}}{2t}).$$

Define the stopping time $\tau = \inf \{t > 0, |X_t - X_0| \ge \varepsilon \}.$

Then we have

$$P^{x}[\sup_{s \leq t} |X_{s} - x| \geq \varepsilon]$$

$$= P^{x}[\tau \leq t]$$

$$= P^{x}[|X_{t} - X_{0}| \geq \frac{\varepsilon}{2}; \tau \leq t] + P^{x}[|X_{t} - X_{0}| \leq \frac{\varepsilon}{2}; \tau \leq t]$$

$$\leq P^{x}[|X_{t} - X_{0}| \geq \frac{\varepsilon}{2}] + P^{x}[|X_{t} - X_{\tau}| \geq \frac{\varepsilon}{2}; \tau \leq t]$$

$$= P^{x}[|X_{t} - X_{0}| \geq \frac{\varepsilon}{2}] + E^{x}[E^{X_{\tau}}[|X_{t-\tau} - X_{0}| \geq \frac{\varepsilon}{2}], \tau \leq t].$$

Note that on $\{\tau \leq t\}$,

$$E^{X_{\tau}}[|X_{t-\tau} - X_0| \ge \frac{\varepsilon}{2}] \le \sup_{x \in D, s \le t} P^x[|X_s - x| \ge \frac{\varepsilon}{2}] \le \widetilde{C_1} \exp(-\frac{C_2 \varepsilon^2}{8t}),$$

and (3.33) is proved.

Set

$$L_r = \frac{1}{2} \nabla \cdot (A_r \nabla) \quad on \quad D_r := \{ rx : x \in D \}, \tag{3.34}$$

here $A_r(x) = (a_{ij}^r(x))_{1 \le i,j \le d} := (a_{ij}(x/r))_{1 \le i,j \le d}$.

Proposition 3.4.2. For any $x \in D_r$, if we define the probability measure $P_r^x = P^{x/r}$, then the process $\{rX_{t/r^2}, P_r^x\}_{t\geq 0}$ is a reflecting diffusion on D_r associated with the generator operator L_r equipped with the Neumann boundary condition.

Proof. Firstly, we prove that the heat kernel $p_r(t, x, y)$ of the process $\{rX_{t/r^2}, P_r^x\}_{t\geq 0}$ is equal to $\frac{1}{r^d}p_1(\frac{t}{r^2}, \frac{x}{r}, \frac{y}{r})$, where $p_1(t, x, y)$ is the heat kernel associated with the operator G_1 on D.

Let $Y_t = rX_{t/r^2}$. For any function $f \in C^{\infty}(D_r)$, define the semigroup associated with (Y_t, P_r^x) as $T_t^r f(x) = E_r^x [f(Y_t)]$. On the one hand, we have, for $x \in D_r$, $T_t^r f(x) = \int_{D_r} f(y) p_r(t, x, y) \, dy$.

On the other hand, we have, for $x \in D_r$,

$$\begin{split} T_t^r f(x) &= E_r^x [f(Y_t)] = E^{x/r} [f(rX_{t/r^2})] &= \int_D f(ry) p_1(\frac{t}{r^2}, \frac{x}{r}, y) \, dy \\ &= \frac{1}{r^d} \int_{D_r} f(y) p_1(\frac{t}{r^2}, \frac{x}{r}, \frac{y}{r}) \, dy \end{split}$$

This implies

$$p_r(t, x, y) = \frac{1}{r^d} p_1(\frac{t}{r^2}, \frac{x}{r}, \frac{y}{r})$$

Secondly, we prove that the generator of the semigroup T_t^r is L_r . Fix any $f \in C^{\infty}(\bar{D}_r)$, then $\tilde{f}(z) := f(rz)$ for $z \in D$ belongs to $C^{\infty}(\bar{D})$. For any $x \in D_r$,

$$\begin{split} \lim_{t \to 0} \frac{f - T_t^r f}{t}(x) &= \lim_{t \to 0} \frac{1}{t} \left(f(x) - \frac{1}{r^d} \int_{D_r} f(y) p_1(\frac{t}{r^2}, \frac{x}{r}, \frac{y}{r}) \, dy \right) \\ &= \lim_{t \to 0} \frac{1}{t} \left(f(x) - \int_D f(ry) p_1(\frac{t}{r^2}, \frac{x}{r}, y) \, dy \right) \\ &= \lim_{t \to 0} \frac{1}{t} \left(\tilde{f}(x/r) - \int_D \tilde{f}(y) p_1(\frac{t}{r^2}, \frac{x}{r}, y) \, dy \right) \\ &= \lim_{t \to 0} \frac{1}{t} \left(\tilde{f} - T_{t/r^2} \tilde{f} \right) (x/r) = \frac{1}{r^2} (L_1 \tilde{f}) (x/r) \\ &= \frac{1}{2r^2} \nabla \cdot (A \nabla \tilde{f}) (x/r) = \frac{1}{2} \nabla \cdot (A_r \nabla f) (x). \end{split}$$

The last equality follows from the fact that $\frac{\partial \tilde{f}}{\partial x_i}(x/r) = r \frac{\partial f}{\partial x_i}(x)$ and $r \frac{\partial a_{ij}}{\partial x_i}(x) = \frac{\partial a_{ij}}{\partial x_i}(x/r)$, for $x \in D_r$. Without loss of generalization, we assume that there is a smooth function ϕ such that $\partial D = \{x : \phi(x) = 0\}$. Defining $\tilde{\phi}(x) = \phi(x/r)$ for $x \in D_r$, we get $\partial D_r = \{x : \tilde{\phi}(x) = 0\}$. Therefore, for any $x \in \partial D_r$,

$$< A_{r}(x)\vec{n}(x), \nabla p_{r}(t, x, y) > = \frac{1}{|\nabla \phi(x)|} < A_{r}(x)\nabla \tilde{\phi}(x), \frac{1}{r^{1+d}}\nabla p_{1}(t/r^{2}, x/r, y/r) >$$
$$= \frac{1}{r^{2+d}|\nabla \phi(x)|} < A(x/r)\nabla \phi(x/r), \nabla p_{1}(t/r^{2}, x/r, y/r) >$$
$$= 0$$

This proves that p_r satisfies the Neumann boundary condition.

Proposition 3.4.3. There exist two constants c and c' such that

$$p_1(t, x, y) \ge \frac{c}{t^{\frac{d}{2}}}$$
 whenever $|x - y| \le c'\sqrt{t}$.

Proof. By [3] and Proposition 3.4.1, there are some constants $t_0 > 0$ and \tilde{c} depending on the Lipschitz constant γ of the boundary of domain D and the ellipticity constant λ , such that

 $p_1(t_0, x, y) \ge \tilde{c}$ whenever $|x - y| \le 1$

Set $c' = 1/\sqrt{t_0}$. Fix any t > 0 and $x, y \in D$ with $|x - y| \le c'\sqrt{t}$. Set $r = \sqrt{t_0/t}$. Since the domain D_r shares the same Lipschitz constant with D and $p_r(t, \cdot, \cdot)$ admits the same upper bound as $p_1(t, x, y)$, we know that $p_r(t_0, m, n) \ge \tilde{c}$ if $|m - n| \le 1$.

Therefore,

$$|rx - ry| \le rc'\sqrt{t} = c'\sqrt{t_0} = 1$$

implies that $p_r(t_0, rx, ry) \geq \tilde{c}$. So

$$p_r(t_0, rx, ry) = \frac{1}{r^d} p_1(t_0/r^2, x, y) = \frac{1}{r^d} p_1(t, x, y) \ge \tilde{c}.$$

Setting $c = \tilde{c} t_0^{\frac{d}{2}}$, we get $p_1(t, x, y) \ge \frac{c}{t^{\frac{d}{2}}}$ for $|x - y| \le c'\sqrt{t}.$

After the preparation of Proposition 3.4.3, the following result follows similarly as the proof of Theorem 2.7 in [14].

Theorem 3.4.4. Suppose that domain D is convex. There exist constants $C_1, C_2 > 0$ and $T_3 > 0$, such that for $x, y \in D, t \in [0, T_3]$

$$p_1(t, x, y) \ge C_1 t^{-\frac{d}{2}} \exp(-\frac{C_2 |x - y|^2}{t})$$

Proof. Fix $t > 0, x, y \in D$. If $|x - y| < c'\sqrt{t}$, by last proposition, we have $p_1(t, x, y) \ge ct^{-\frac{d}{2}} \ge ct^{-\frac{d}{2}} \exp(-\frac{|x-y|^2}{t})$. Therefore we suppose that $|x - y| \ge c'\sqrt{t}$.

Because ∂D is compact and smooth, there exist two constants $r_0 > 0$ and $\delta > 0$ only depending on D, such that $|D \cap B(x,r)| \ge \delta |B(x,r)|$ for $x \in \overline{D}$, $r \le r_0$. Here $B(z,r) = \{x \in \mathbb{R}^d : |x-z| < r\}$ and $|\cdot|$ means the volume of the set \cdot .

Since D is convex, the set $c(x, y) := \{(1 - t)x + ty, t \in [0, 1]\} \subset D$. Let k be the smallest positive integer dominating $\frac{9|x-y|^2}{c'^2t}$ and set $s_i = x + \frac{i(y-x)}{k}(1 \le i \le k-1)$, $r = \frac{c'}{3}\sqrt{\frac{t}{k}}$. So that $r = \frac{c'}{3}\sqrt{\frac{t}{k}} \ge \frac{|x-y|}{k}$ and $r < r_0$.

Denote $S_i = B(s_i, r)$, where $B(z, r) = \{x \in \mathbb{R}^d : |x - z| < r\}$. Then $\{S_i\}_{i=0,...,k}$ is a sequence of balls. For $\xi_i \in S_i$, $max\{|x - \xi_1|, |y - \xi_{k-1}|, |\xi_l - \xi_{l+1}|, 1 \le l < k-1\} < c'\sqrt{\frac{t}{k}}$. Hence, by the last proposition,

$$p_{1}(t, x, y) = \int_{D} \cdots \int_{D} p_{1}(\frac{t}{k}, x, \xi_{1}) p_{1}(\frac{t}{k}, \xi_{1}, \xi_{2}) \cdots p_{1}(\frac{t}{k}, \xi_{k-1}, y) d\xi_{1} \cdots \xi_{k-1}$$

$$\geq \int_{S_{1} \cap D} \cdots \int_{S_{k-1} \cap D} p_{1}(\frac{t}{k}, x, \xi_{1}) p_{1}(\frac{t}{k}, \xi_{1}, \xi_{2}) \cdots p_{1}(\frac{t}{k}, \xi_{k-1}, y) d\xi_{1} \cdots \xi_{k-1}$$

$$\geq (c(\frac{t}{k})^{-d/2})^{k} |\delta S_{1}|^{k-1}$$

Here, the volume of the ball $|S_1| = \omega_d r^d = \omega_d (\frac{c'}{3})^d (\frac{t}{k})^{d/2}$, and the constant ω_d only depends on d.

Therefore, we get that

$$p_1(t, x, y) \ge \left(\frac{3}{\omega_d c'}\right) t^{-d/2} k^{d/2} \delta^{k-1} \left(\frac{1}{3} \omega_d c' c\right)^k =: C_1 t^{-d/2} k^{d/2} C_2^k$$

Then, there is a constant $C_3 > 0$, such that

$$\log p_1(t, x, y) \ge \log C_1 t^{-d/2} + k \log C_2 \ge \log C_1 t^{-d/2} - C_3 k$$

Since $\frac{9|x-y|^2}{c'^2t} \le k < \frac{9|x-y|^2}{c'^2t} + 1$, we know that

$$p_1(t, x, y) \ge C_4 t^{-\frac{d}{2}} e^{\frac{-C_6 |x-y|^2}{t}}.$$
(3.35)

Here, the constants C_4, C_5, C_6 only depend on λ, T_3 and d.

3.4.2 Lower Bounds for Heat Kernels Associated with G_2

Recall that

$$G_2 = \frac{1}{2}\nabla \cdot (A\nabla) + b \cdot \nabla .$$

As in the proof of theorem 3.3.5, we set

$$f_1(t, x, y) = b(x) \cdot \nabla_x p_1(t, x, y),$$
$$f_{n+1}(t, x, y) = \int_0^t \int_D b(z) \cdot \nabla_x p_1(t - s, x, z) f_n(s, z, y) \, dz \, ds.$$

Then by the construction of function $p_2(t, x, y)$ in Theorem 3.3.5, the following inequality holds: there are constants $\lambda_1 > 0, \lambda_2 > 0$, such that

$$\begin{aligned} |p_2(t,x,y) - p_1(t,x,y)| &= |\sum_{k=1}^{\infty} \int_0^t \int_D p_1(t-s,x,z) f_{n+1}(s,z,y) \, dz \, ds| \\ &\leq \lambda_1 N_h^{\lambda_2}(b) t^{-\frac{d}{2}} \exp(-\frac{\lambda_2 |x-y|^2}{t}), \end{aligned}$$

and consequently,

$$p_2(t, x, y) \ge p_1(t, x, y) - \lambda_1 N_h^{\lambda_2}(b) t^{-\frac{d}{2}} \exp(-\frac{\lambda_2 |x - y|^2}{t})$$
(3.36)

Because $\lim_{h\to 0} N_h^{\lambda_2}(b) = 0$, we can chose h small enough so that

$$p_{2}(t, x, y) \geq p_{1}(t, x, y) - \lambda_{1} N_{h}^{\alpha}(b) t^{-\frac{d}{2}} \exp(-\frac{\lambda_{2}|x - y|^{2}}{t})$$

$$\geq C_{4} t^{-\frac{d}{2}} e^{\frac{-C_{6}|x - y|^{2}}{t}} - \lambda_{1} N_{h}^{\alpha}(b) t^{-\frac{d}{2}} \exp(-\frac{\lambda_{2}|x - y|^{2}}{t})$$

$$\geq \frac{1}{2} C_{4} t^{-\frac{d}{2}} e^{\frac{-C_{6} \vee \lambda_{2}|x - y|^{2}}{t}}, \qquad (3.37)$$

for $t \leq h$.

3.5 Two-sided Estimates for the Heat Kernel p(t, x, y)of Operator G

Recall that $G = \frac{1}{2}\nabla \cdot (A\nabla) + b \cdot \nabla + q$ with b and q defined in (3.12). In this section, we establish both the upper and lower bounds of the heat kernel p(t, x, y) associated with the operator G equipped with the Neumann boundary condition.

3.5.1 Upper Bounds for the Heat Kernel p(t, x, y)

Define the exponential martingale

$$Z_t = e^{\int_0^t A^{-1}b(X_s)dM_s - \frac{1}{2}\int_0^t \langle b, A^{-1}b \rangle \langle X_s\rangle ds}.$$

where the martingale $\{M_t, t \ge 0\}$ was defined in (3.5) and (3.6).

As $|b| \in L^p$, p > d, then by [27], we know that Z_t is an exponential martingale on the probability space $\{\Omega, \mathcal{F}_t, P^x\}_{t\geq 0}$. Define a family of measures $(\tilde{P}^x, x \in D)$ on \mathcal{F}_{∞} by

$$\frac{d\tilde{P}^x}{dP^x}|_{\mathcal{F}_t} = Z_t.$$

Then, by the Feynmann-Kac formula, if $f \in C^{\infty}(D)$, we have:

$$S_t f(x) = E^x [Z_t e^{\int_0^t q(X_s) ds} f(X_t)] = \tilde{E}^x [e^{\int_0^t q(X_s) ds} f(X_t)]$$

= $\tilde{E}^x [f(X_t) \sum_{n=0}^\infty \frac{1}{n!} (\int_0^t q(X_s) ds)^n].$

Here \tilde{E}^x denotes the expectation under \tilde{P}^x . Set

$$Q_0(t,x) = \tilde{E}^x[f(X_t)]$$
 (3.38)

$$Q_n(t,x) = \tilde{E}^x \left[\int_0^t q(X_s) Q_{n-1}(t-s,X_s) \, ds \right]$$
(3.39)

Then if the following series is convergent, we have

$$S_t f(x) = \sum_{n=0}^{\infty} Q_n(t, x)$$

Theorem 3.5.1. The heat kernel p(t, x, y) associated with the operator G equipped with Neumann boundary condition has the following upper bound of Gaussian type: for $K_1, K_2 > 0$,

$$p(t, x, y) \le K_1 t^{-\frac{d}{2}} e^{-\frac{K_2 |x-y|^2}{t}}, \quad for 0 \le t \le T_4,$$
(3.40)

where T_4 is a positive constant.

Proof. Firstly, we establish the upper bounds for $Q_n(t, x)$:

$$\begin{aligned} |Q_{0}(t,x)| &= |\int_{D} p_{2}(t,x,y)f(y)dy| \leq \int_{D} M_{13}t^{-\frac{d}{2}}e^{-\frac{M_{12}|x-y|^{2}}{t}}|f(y)|\,dy, \end{aligned} \tag{3.41} \\ |Q_{1}(t,x)| &= |\int_{0}^{t} \tilde{E}^{x}[q(X_{s})Q_{0}(t-s,X_{s})]\,ds| \\ &= |\int_{0}^{t} \int_{D} p_{2}(s,x,y)q(y)Q_{0}(t-s,y)\,dyds| \\ &\leq \int_{D} |f(z)|(\int_{0}^{t} \int_{D} M_{13}s^{-\frac{d}{2}}e^{-\frac{M_{12}|x-y|^{2}}{s}}|q(y)|M_{13}(t-s)^{-\frac{d}{2}}e^{-\frac{M_{12}|y-z|^{2}}{t-s}}\,dyds)\,dz \\ &\leq M_{13}^{2}N_{h}^{M_{12}}(|q|)\int_{D}t^{-\frac{d}{2}}e^{-\frac{M_{12}|x-z|^{2}}{t}}|f(z)|\,dz. \end{aligned}$$

The last inequality follows from the fact that q satisfies condition K. Iterating the calculation, we get

$$|Q_n(t,x)| \le M_{13}^{n+1} (N_h^{M_{12}}(|q|))^n \int_D t^{-\frac{d}{2}} e^{-\frac{M_{12}|x-z|^2}{t}} |f(z)| \, dz.$$

There exits $T_4 > 0$, such that if $h < T_4$, then $M_{13}N_h^{M_{12}}(|q|) < 1$. This implies

$$\sum_{n=0}^{\infty} Q_n(t,x) \le \frac{M_{13}}{1 - M_{13}^2 N_h^{M_{12}}(|q|)} \int_D t^{-\frac{d}{2}} e^{-\frac{M_{12}|x-z|^2}{t}} |f(z)| \, dz$$

for $t \leq h \leq T_4$.

Therefore, we conclude that there exists $K_1, K_2 > 0$ such that

$$|p(t, x, y)| \le K_1 t^{-\frac{d}{2}} e^{-\frac{K_2|x-z|^2}{t}}, \quad \text{for} 0 \le t \le T_4.$$

3.5.2 Lower Bounds for the Heat Kernel p(t, x, y)

Theorem 3.5.2. Given a positive function $f \in L^p$ for some p > d. Let $\xi(t, x, y)$ be the heat kernel of the operator $G_f := \frac{1}{2}\nabla(A\nabla) + b \cdot \nabla - f$, that is,

$$\tilde{E}^{x}[\exp\left(-\int_{0}^{t} f(X_{s})ds\right)\cdot] = \int_{D} \xi(t, x, y) \cdot dy.$$

Then there are two constants $k_1, k_2 > 0$ and T > 0, such that for $t \in [0, T]$

$$\xi(t, x, y) \ge k_1 t^{-\frac{d}{2}} e^{-\frac{k_2 |x-y|^2}{t}}.$$
(3.43)

Proof. We set $p_f(t, x, y)$ to be the heat kernel such that

$$\tilde{E}^{x}[\exp\left(\int_{0}^{t} f(X_{s})ds\right)\cdot] = \int_{D} p_{f}(t, x, y) \cdot dy.$$

Let $g \in C(\overline{D})$ be a non-negative function satisfying $||g||_{L^1(D)} = 1$. By Hölder's inequality, we get

$$\begin{aligned} |\int_{D} p_{2}(t,x,y)g(y) \, dy| &= |\tilde{E}^{x}[g(X_{t})]| &\leq \tilde{E}^{x}[e^{\int_{0}^{t} f(X_{s}) \, ds}g(X_{t})]^{\frac{1}{2}}\tilde{E}^{x}[e^{-\int_{0}^{t} f(X_{s}) \, ds}g(X_{t})]^{\frac{1}{2}} \\ &= (\int_{D} p_{f}(t,x,y)g(y) \, dy)^{\frac{1}{2}} (\int_{D} \xi(t,x,y)g(y) \, dy)^{\frac{1}{2}} \end{aligned}$$

Let $g_k, k \ge 0$ be a sequence of nonnegative continuous functions on D with $\|g_k\|_{L^1(D)} = 1$, such that g_k tends to the Dirac measure δ_{y_0} at y_0 . Then we have

$$p_2(t, x, y_0)^2 \le p_f(t, x, y_0)\xi(t, x, y_0).$$

By the lower bound of $p_2(t, x, y)$ and upper bound of $p_f(t, x, y)$ established in Section 3.4.2 and Theorem 3.5.1 respectively, we get the required lower bounds in (3.43).

Decompose function q as follows:

$$q = q^+ - q^-,$$

where $q^+ = \max\{q, 0\}$ and $q^- = -\min\{q, 0\}$.

Let f be a nonnegative continuous function on \overline{D} . Then we have

$$\int_{D} p(t, x, y) f(y) \, dy = \tilde{E}^{x} [\exp\left(\int_{0}^{t} q(X_{s}) \, ds\right) f(X_{t})] \\ = \tilde{E}^{x} [\exp\left(\int_{0}^{t} q^{+}(X_{s}) - q^{-}(X_{s}) \, ds\right) f(X_{t})] \\ \ge \tilde{E}^{x} [\exp\left(\int_{0}^{t} -q^{-}(X_{s}) \, ds\right) f(X_{t})] \\ = \int_{D} \hat{\xi}(t, x, y) f(y) \, dy$$
(3.44)

where $\hat{\xi}(t, x, y)$ is the heat kernel associated with the semigroup

$$\hat{S}_t \cdot := \tilde{E}^x [\exp\left(\int_0^t -q^-(X_s) \, ds\right) \cdot].$$

By Theorem 3.5.2, we know that there exist constants $\hat{k}_1, \hat{k}_2 > 0$ and $T_5 > 0$, such that the heat kernel $\hat{\xi}(t, x, y) \ge \hat{k}_1 t^{-\frac{d}{2}} e^{-\frac{\hat{k}_2 |x-y|^2}{t}}$ for $t \le T_5$.

Therefore, the heat kernel p(t, x, y) associated with operator G also admits the following lower bound:

$$p(t, x, y) \ge \hat{\xi}(t, x, y) \ge \hat{k}_1 t^{-\frac{d}{2}} e^{-\frac{\hat{k}_2 |x-y|^2}{t}}, \quad \text{for} 0 \le t \le T_5.$$

Chapter 4

Neumann Problems for Semilinear Elliptic PDEs

4.1 Introduction

In this chapter, our aim is to use probabilistic methods to solve the mixed boundary value problem for semilinear second order elliptic partial differential equations (PDEs in abbreviation) of the following form:

$$\begin{cases} Lu(x) = -F(x, u(x), \nabla u(x)) & \text{on } D, \\ \frac{1}{2} \frac{\partial u}{\partial \vec{\gamma}}(x) - \widehat{B} \cdot \vec{n}(x)u(x) = \Phi(x) & \text{on } \partial D, \end{cases}$$

$$(4.1)$$

on the bounded convex domain D with smooth boundary.

Both the elliptic operator $L = \frac{1}{2} \nabla \cdot (A\nabla) + B \cdot \nabla - \nabla \cdot (\hat{B} \cdot) + Q$ and the quadratic form $\mathcal{Q}(\cdot, \cdot)$ associated with L are as in Chapter 3.

The function $F(\cdot, \cdot, \cdot)$ in (4.1) is a nonlinear function defined on $R^d \times R \times R^d$. Φ is a bounded measurable function defined on the boundary ∂D . Set $\vec{\gamma} = A\vec{n}$, where \vec{n} denotes the inward normal vector field defined on the boundary ∂D .

As discussed in Chapter 3, the term $\nabla \cdot (\hat{B} \cdot)$ is tackled using the time-reversal of Girsanov transform with the symmetric diffusion $(\Omega, P_x^0, X_t^0, t \ge 0)$ associated with the operator

$$G_1 = \frac{1}{2} \nabla \cdot (A \nabla),$$

which is the symmetric part of L. The semigroup T_t associated with the operator L has the following form:

$$T_{t}f(x) = E_{x}^{0}[f(X_{t}^{0})\exp(\int_{0}^{t}(A^{-1}B)^{*}(X_{s}^{0})dM_{s}^{0} + (\int_{0}^{t}(A^{-1}\hat{B})^{*}(X_{s}^{0})dM_{s}^{0})\circ\gamma_{t}^{0} - \frac{1}{2}\int_{0}^{t}(B-\hat{B})^{*}A^{-1}(B-\hat{B})(X_{s}^{0})ds + \int_{0}^{t}Q(X_{s}^{0})ds)] = e^{-v(x)}S_{t}[fe^{v}](x),$$

$$(4.2)$$

where v is got in the Section 3.2.

Here M^0 is the martingale part of the diffusion X^0 , γ_t^0 is the reverse operator and $\{S_t\}$ is the semigroup associated with the operator $G = \frac{1}{2}\nabla \cdot (A\nabla) + b \cdot \nabla + q$.

Please note that in this chapter, for convenience, we will use $(\Omega, P_x^0, X_t^0, t \ge 0)$ to denote the process associated the operator G_1 and use $(\Omega, P_x, X(t), t > 0)$ to denote the process associated with $G_2 = \frac{1}{2}\nabla \cdot (A\nabla) + b \cdot \nabla$. This notation is different from that in Chapter 3.

Chapter 4 is organized as follows. In the next section, we obtain a pair of L^2 solutions (Y, Z) of the BSDEs with infinite horizon:

$$dY(t) = -F(X(t), Y(t), Z(t))dt + e^{\int_0^t q(X(u))dt} \Phi(X(s))dL_t + \langle Z(t), dM(t) \rangle,$$

$$\lim_{t \to \infty} e^{\int_0^t d(X(u))du} Y_t = 0 \quad in \quad L^2(\Omega),$$
(4.3)

where $\{X(t)\}_{t>0}$ is the reflecting diffusion associated with the infinitesimal generator $G_2 = \frac{1}{2}\nabla(A\nabla) + b \cdot \nabla$, M(t) is the martingale part of $\{X(t)\}_{t>0}$ and function d is associated with the semilinear function F(x, y, z) and the ellipticity constant λ . Here we call (Y, Z) the L^2 solution because we estimate the solution in $L^2(\Omega)$ in the whole Section 4.2. The crucial point in this section is that the BSDE to be considered has a term associated with the local time and needs to be solved on an infinite time interval.

In section 4.3, we solve the linear PDEs of the form:

$$\begin{cases} \frac{1}{2}\nabla(A\nabla u)(x) + b \cdot \nabla u(x) + qu(x) = F(x) & \text{on } D\\ \frac{1}{2}\frac{\partial u}{\partial \vec{\gamma}}(x) = \phi(x) & \text{on } \partial D \end{cases}.$$

$$(4.4)$$

Some useful estimates, which are used in the subsequent discussions, are also proved in this section.

In section 4.4, we obtain the solution to the semilinear PDE:

$$\begin{cases} \frac{1}{2}\nabla(A\nabla u)(x) + b \cdot \nabla u(x) + qu(x) = G(x, u(x), \nabla u(x)) & \text{on } D, \\ \frac{1}{2}\frac{\partial u}{\partial \vec{\gamma}}(x) = \phi(x) & \text{on } \partial D. \end{cases}$$

$$(4.5)$$

Using the solution $(Y_x(t), Z_x(t))$ to the BSDE (4.3), we set $F(x) := G(x, u_0(x), v_0(x))$, where $u_0(x) = E_x[Y_x(0)]$ and $v_0(x) = E_x[Z_x(0)]$, so that we can transform the semilinear equation (4.5) to the linear equation (4.4). With the solution u(x) of the linear case shown in Section 4.3, what we need to do is to prove that $u_0(x)$ coincides with u(x). Then we will show that u(x) is the solution to (4.5).

In section 4.5, we finally solve the non-linear equation:

$$\begin{cases} Lu(x) = -F(x, u(x)), & \text{on } D\\ \frac{1}{2}\frac{\partial u}{\partial \vec{\gamma}}(x) - \widehat{B} \cdot n(x)u(x) = \Phi(x) & \text{on } \partial D. \end{cases}$$

$$(4.6)$$

The relationship between the operator L and G is the crucial point to complete the proof. We apply the transformation introduced in [7] and Section 3.2 to transform the problem (4.6) to a similar problem as (4.5), then an inverse transformation will solve the final problem (4.6).

In section 4.6, we also obtain the L^1 solutions of the BSDEs (4.3). Then using the same methods introduced in Section 4.4 and 4.5, we solve the equations (4.5) and (4.6). Since all of the solutions are estimated in $L^1(\Omega)$ in this section, different conditions on the coefficients of operator L are obtained.

4.2 BSDEs with Singular Coefficients and Infinite Horizon

Consider the operator

$$G_2 = \frac{1}{2} \sum_{i,j=1}^d \frac{\partial}{\partial x_i} \left(a_{ij}(x) \frac{\partial}{\partial x_j} \right) + \sum_{i=1}^d b_i(x) \frac{\partial}{\partial x_i},$$

on the domian D equipped with the Neumann boundary condition:

$$\frac{\partial}{\partial \vec{\gamma}} := < A \vec{n}, \nabla \cdot > = 0, \quad \text{on} \quad \partial D.$$

By [26], there exists a unique reflecting diffusion process denoted by $(\Omega, \mathcal{F}_t, X_x(t), P_x, \theta_t, x \in D)$ associated with the generator G_2 . Here $\theta : \Omega \to \Omega$ is the shift operator defined as follows:

$$X_x(s)(\theta_t \cdot) = X_x(t+s), \quad s, t \ge 0.$$

Let E_x denote the expectation under the measure P_x . Set $\tilde{b} = {\tilde{b}_1, ..., \tilde{b}_d}$, where $\tilde{b}_i = \frac{1}{2} \sum_j \frac{\partial a_{ij}}{\partial x_j} + b_i$.

Then the process $X_x(t)$ has the following decomposition:

$$X_x(t) = X_x(0) + M_x(t) + \int_0^t \tilde{b}(X_x(s)) \, ds + \int_0^t A\vec{n}(X_x(s)) \, dL_s, \quad P_x - a.s.. \quad (4.7)$$

Here M_x is a \mathcal{F}_t -measurable square integrable continuous martingale additive functional:

$$M_x(t) = \int_0^t \sigma(X_s) dB_s,$$

where the matrix $\sigma(x)$ is the positive definite symmetric square root of the matrix

A(x) and $\{(B_t\}_{t\geq 0}$ is a standard Brownian motion.

 L_t is a positive increasing continuous additive functional satisfying

$$L_t = \int_0^t I_{\{X_x(s) \in \partial D\}} \, dL_s$$

We write $X_x(t)$ as X(t) for short in the following discussion.

In this section, we will study the backward stochastic differential equations with singular coefficients and infinite horizon associated with the martingale part $M_x(t)$ and the local time L_t . The unique L^2 solution of such kind of BSDE is obtained.

Let $g(\omega, t, y, z) : \Omega \times R^+ \times R \times R^d \to R$ be a progressively measurable function. Consider the following conditions:

(A.1) $(y_1 - y_2)(g(t, y_1, z) - g(t, y_2, z)) \le -a_1(t)|y_1 - y_2|^2$, a.s., (A.2) $|g(t, y, z_1) - g(t, y, z_2)| \le a_2|z_1 - z_2|$, a.s., (A.3) $|g(t, y, z)| \le |g(t, 0, 0)| + a_3(t)(1 + |y|)$, a.s..

Here $a_1(t)$ and $a_3(t)$ are two progressively measurable processes and a_2 is a constant. Set $a(t) = -a_1(t) + \delta a_2^2$, for some constant $\delta > \frac{1}{2\lambda}$, where λ is the elliptic constant which appeared in (3.2).

Lemma 4.2.1. Assume the conditions (A.1)-(A.3) hold and

$$E_x \left[\int_0^\infty e^{2\int_0^t a(u) \, du} |g(t,0,0)|^2 \, dt \right] < \infty.$$

Then there exists a unique solution $(Y_x(t), Z_x(t))$ to the following backward stochastic differential equation:

$$Y_x(t) = Y_x(T) + \int_t^T g(s, Y_x(s), Z_x(s)) \, ds - \int_t^T \langle Z_x(s), dM_x(s) \rangle, \quad \text{for} \quad 0 \le t \le T;$$

$$\lim_{t \to \infty} e^{\int_0^t a(u) \, du} Y_x(t) = 0, \quad \text{in} \quad L^2(\Omega). \tag{4.8}$$

Moreover,

$$E_x[\sup_t e^{2\int_0^t a(u)\,du} |Y_x(t)|^2] < \infty \quad and \quad E_x[\int_0^\infty e^{2\int_0^s a(u)\,du} |Z_x(s)|^2\,ds] < \infty.$$
(4.9)

Proof. Existence:

The proof of this lemma is similar to that of Theorem 3.2 in [42], but the terminal conditions here are different. By Theorem 3.1 in [42], the following BSDE has a unique solution $(Y_x^n(t), Z_x^n(t))$:

$$Y_x^n(t) = \int_t^n g(s, Y_x^n(s), Z_x^n(s)) \, ds - \int_t^n \langle Z_x^n(s), dM_x(s) \rangle, \quad \text{for} \quad 0 \le t \le n,$$
(4.10)

and moreover,

$$Y_x^n(t) = 0, \quad Z_x^n(t) = 0, \quad t > n.$$

Fix t > 0 and n > m > t. It follows that

$$\begin{split} e^{2\int_{0}^{t}a(u)\,du}|Y_{x}^{n}(t)-Y_{x}^{m}(t)|^{2} \\ &+\int_{t}^{n}e^{2\int_{0}^{s}a(u)\,du} < A(X(s))(Z_{x}^{n}(s)-Z_{x}^{m}(s)), (Z_{x}^{n}(s)-Z_{x}^{m}(s)) > ds \\ &= -2\int_{t}^{n}a(s)e^{2\int_{0}^{s}a(u)\,du}|Y_{x}^{n}(s)-Y_{x}^{m}(s)|^{2}\,ds \\ &+2\int_{t}^{n}e^{2\int_{0}^{s}a(u)\,du}(Y_{x}^{n}(s)-Y_{x}^{m}(s))(g(s,Y_{x}^{n}(s),Z_{x}^{n}(s))-g(s,Y_{x}^{m}(s),Z_{x}^{m}(s)))\,ds \\ &+2\int_{m}^{n}e^{2\int_{0}^{s}a(u)\,du}(Y_{x}^{n}(s)-Y_{x}^{m}(s))g(s,0,0)\,ds \\ &-2\int_{t}^{n}e^{2\int_{0}^{s}a(u)\,du}(Y_{x}^{n}(s)-Y_{x}^{m}(s)) < Z_{x}^{n}(s)-Z_{x}^{m}(s),dM_{x}(s) > . \end{split}$$

Choose two positive numbers δ_1 and δ_2 such that $\delta_1 > \frac{1}{2\lambda}$ and $\delta_1 + \delta_2 < \delta$.

Then by

$$2\int_{t}^{n} e^{2\int_{0}^{s} a(u) \, du} (Y_{x}^{n}(s) - Y_{x}^{m}(s)) (g(s, Y_{x}^{n}(s), Z_{x}^{n}(s)) - g(s, Y_{x}^{m}(s), Z_{x}^{m}(s))) \, ds$$

$$\leq -2\int_{t}^{n} a_{1}(s) e^{2\int_{0}^{s} a(u) \, du} |Y_{x}^{n}(s) - Y_{x}^{m}(s)|^{2} \, ds$$

$$+ 2\delta_{1}a_{2}^{2}\int_{t}^{n} e^{2\int_{0}^{s} a(u) \, du} |Y_{x}^{n}(s) - Y_{x}^{m}(s)|^{2} \, ds$$

$$+ \frac{1}{2\lambda\delta_{1}}\int_{t}^{n} e^{2\int_{0}^{s} a(u) \, du} < A(X(s))(Z_{x}^{n}(s) - Z_{x}^{m}(s)), (Z_{x}^{n}(s) - Z_{x}^{m}(s)) > ds$$

and

$$2\int_{m}^{n} e^{2\int_{0}^{s} a(u) \, du} (Y_{x}^{n}(s) - Y_{x}^{m}(s))g(s, 0, 0) \, ds$$

$$\leq 2\delta_{2}a_{2}^{2}\int_{m}^{n} e^{2\int_{0}^{s} a(u) \, du} |Y_{x}^{n}(s) - Y_{x}^{m}(s)|^{2} \, ds + \frac{1}{2\delta_{2}a_{2}^{2}}\int_{m}^{n} e^{2\int_{0}^{s} a(u) \, du} |g(s, 0, 0)|^{2} \, ds,$$

it follows that

$$\begin{split} E_x[e^{2\int_0^t a(u)\,du}|Y_x^n(t) - Y_x^m(t)|^2] &+ \frac{1}{\lambda}(1 - \frac{1}{2\lambda\delta_1})E_x[\int_t^\infty e^{2\int_0^s a(u)\,du}|Z_x^n(s) - Z_x^m(s)|^2\,ds]\\ &\leq \frac{1}{2\delta_2 a_2^2}E_x[\int_m^n e^{2\int_0^s a(u)\,du}|g(s,0,0)|^2\,ds]. \end{split}$$

This implies that

$$E_x[\int_0^\infty e^{2\int_0^s a(u)\,du} |Z_x^n(s) - Z_x^m(s)|^2\,ds] \to 0, \quad as \quad m, n \to \infty.$$

Hence there exists \tilde{Z}_x such that

$$\tilde{Z}_x = \lim_{n \to \infty} e^{\int_0^{\cdot} a(u) \, du} Z_x^n \quad \text{in} \quad L^2([0,\infty) \times \Omega).$$

At the same time, we also obtain the following estimate:

$$\sup_{t} e^{2\int_{0}^{t} a(u) \, du} |Y_{x}^{n}(t) - Y_{x}^{m}(t)|^{2} \\
\leq \frac{1}{2\delta_{2}a_{2}^{2}} \int_{m}^{n} e^{2\int_{0}^{s} a(u) \, du} |g(s, 0, 0)|^{2} \, ds \\
+ 2\sup_{t} |\int_{t}^{n} e^{2\int_{0}^{s} a(u) \, du} (Y_{x}^{n}(s) - Y_{x}^{m}(s)) < Z_{x}^{n}(s) - Z_{x}^{m}(s), \, dM_{x}(s) > |.$$

Taking expectation on both sides of the above inequality, by the Burkholder-Davis-Gundy inequality, we obtain

$$E_{x}[\sup_{t} e^{2\int_{0}^{t} a(u) \, du} |Y_{x}^{n}(t) - Y_{x}^{m}(t)|^{2}]$$

$$\leq \frac{1}{2\delta_{2}a_{2}^{2}} E_{x}[\int_{m}^{n} e^{2\int_{0}^{s} a(u) \, du} |g(s, 0, 0)|^{2} \, ds]$$

$$+ C_{1}E_{x}[\{\int_{t}^{n} e^{4\int_{0}^{s} a(u) \, du} |Y_{x}^{n}(s) - Y_{x}^{m}(s)|^{2} |Z_{x}^{n}(s) - Z_{x}^{m}(s)|^{2} \, ds\}^{\frac{1}{2}}]$$

$$\leq \frac{1}{2\delta_{2}a_{2}^{2}} E_{x}[\int_{m}^{n} e^{2\int_{0}^{s} a(u) \, du} |g(s, 0, 0)|^{2} \, ds] + \frac{1}{2} E_{x}[\sup_{t} e^{2\int_{0}^{t} a(u) \, du} |Y_{x}^{n}(t) - Y_{x}^{m}(t)|^{2}]$$

$$+ C_{2}E_{x}[\int_{0}^{\infty} e^{2\int_{0}^{s} a(u) \, du} |Z_{x}^{n}(s) - Z_{x}^{m}(s)|^{2} \, ds].$$

Thus

$$E_{x}[\sup_{t} e^{2\int_{0}^{t} a(u) \, du} |Y_{x}^{n}(t) - Y_{x}^{m}(t)|^{2}] \\ \leq \frac{1}{\delta_{2}a_{2}^{2}} E_{x}[\int_{m}^{n} e^{2\int_{0}^{s} a(u) \, du} |g(s,0,0)|^{2} \, ds] + 2C_{2}E_{x}[\int_{0}^{\infty} e^{2\int_{0}^{s} a(u) \, du} |Z_{x}^{n}(s) - Z_{x}^{m}(s)|^{2} \, ds] \\ \to 0, \quad as \quad m, n \to \infty.$$

Therefore, there exists $\{\tilde{Y}_x(t)\}$ such that

$$\lim_{n \to \infty} E_x[\sup_t |\tilde{Y}_x(t) - e^{\int_0^t a(u) \, du} Y_x^n(t)|^2] = 0.$$

For any $\varepsilon > 0$, there exist a positive number N such that for any $n \ge N$,

$$E_x[\sup_t |\tilde{Y}_x(t) - e^{\int_0^t a(u) \, du} Y_x^n(t)|^2] < \frac{\varepsilon}{2}.$$

For t > N, noticing $Y_x^N(t) = 0$, it follows that

$$E_{x}[|\tilde{Y}_{x}(t)|^{2}] \leq 2E_{x}[|\tilde{Y}_{x}(t) - e^{\int_{0}^{t}a(u)\,du}|Y_{x}^{N}(t)|^{2}] + 2E_{x}[e^{2\int_{0}^{t}a(u)\,du}|Y_{x}^{N}(t)|^{2}]$$

$$\leq 2E_{x}[\sup_{t}|\tilde{Y}_{x}(t) - e^{\int_{0}^{t}a(u)\,du}|Y_{x}^{N}(t)|^{2}] + 2E_{x}[e^{2\int_{0}^{t}a(u)\,du}|Y_{x}^{N}(t)|^{2}]$$

$$< \varepsilon.$$

Thus we have $\lim_{t\to\infty} E_x[|\tilde{Y}_x(t)|^2] = 0.$ By the chain rule, it is easy to see that, from (4.10),

$$Y_x(t) = e^{-\int_0^t a(u) \, du} \tilde{Y}_x(t) \quad and \quad Z_x(t) = e^{-\int_0^t a(u) \, du} \tilde{Z}_x(t)$$

satisfy the equation (4.8) and

$$\lim_{t \to \infty} E_x[e^{2\int_0^t a(u) \, du} |Y_x(t)|^2] = \lim_{t \to \infty} E_x[|\tilde{Y}_x(t)|^2] = 0.$$

According to the above proof, we also conclude that (4.9) holds.

Uniqueness:

Suppose that (Y_x^1, Z_x^1) and (Y_x^2, Z_x^2) are two solutions of the equation (4.8). Set $\overline{Y}_x(t) = Y_x^1(t) - Y_x^2(t)$ and $\overline{Z}_x(t) = Z_x^1(t) - Z_x^2(t)$, a.s.. Then

$$d(e^{\int_0^t a(u) \, du} \bar{Y}_x(t)) = -e^{\int_0^t a(u) \, du} (g(t, Y_x^1(t), Z_x^1(t)) - g(t, Y_x^2(t), Z_x^2(t))) dt + a(t) e^{\int_0^t a(u) \, du} \bar{Y}_x(t) dt + e^{\int_0^t a(u) \, du} < \bar{Z}_x(t), dM_x(t) > .$$
(4.11)

By Ito's formula, we get, for any t < T,

$$e^{2\int_{0}^{t}a(u)\,du}|\bar{Y}_{x}(t)|^{2} + \int_{t}^{T}e^{2\int_{0}^{t}a(u)\,du} < A(X(s))\bar{Z}_{x}(s), \bar{Z}_{x}(s) > ds$$

$$= e^{2\int_{0}^{T}a(u)\,du}|\bar{Y}_{x}(T)|^{2} + 2\int_{t}^{T}e^{2\int_{0}^{s}a(u)\,du}\bar{Y}_{x}(s)(g(s, Y_{x}^{1}(s), Z_{x}^{1}(s)) - g(s, Y_{x}^{2}(s), Z_{x}^{2}(s)))\,ds$$

$$-2\int_{t}^{T}a(s)e^{2\int_{0}^{s}a(u)\,du}|\bar{Y}_{x}(s)|^{2}\,ds$$

$$-2\int_{t}^{T}a(s)e^{2\int_{0}^{s}a(u)\,du}\bar{Y}_{x}(s) < \bar{Z}_{x}(s), dM_{x}(s) > .$$

$$(4.12)$$

By conditions (A.1) and (A.2), we have

$$2\int_{t}^{T} e^{2\int_{0}^{s} a(u) \, du} \bar{Y}_{x}(s) (g(s, Y_{x}^{1}(s), Z_{x}^{1}(s)) - g(s, Y_{x}^{2}(s), Z_{x}^{2}(s))) \, ds$$

$$= 2\int_{t}^{T} e^{2\int_{0}^{s} a(u) \, du} \bar{Y}_{x}(s) (g(s, Y_{x}^{1}(s), Z_{x}^{1}(s)) - g(s, Y_{x}^{2}(s), Z_{x}^{1}(s))) \, ds$$

$$+ 2\int_{t}^{T} e^{2\int_{0}^{s} a(u) \, du} \bar{Y}_{x}(s) (g(s, Y_{x}^{2}(s), Z_{x}^{1}(s)) - g(s, Y_{x}^{2}(s), Z_{x}^{2}(s))) \, ds$$

$$\leq -2\int_{t}^{T} a_{1}(s) e^{2\int_{0}^{s} a(u) \, du} |\bar{Y}_{x}(s)|^{2} \, ds + a_{2} \int_{t}^{T} e^{2\int_{0}^{s} a(u) \, du} \bar{Y}_{x}(s) |\bar{Z}_{x}(s)| \, ds$$

$$\leq -2\int_{t}^{T} a_{1}(s) e^{2\int_{0}^{s} a(u) \, du} |\bar{Y}_{x}(s)|^{2} \, ds + c'a_{2} \int_{t}^{T} e^{2\int_{0}^{s} a(u) \, du} |\bar{Y}_{x}(s)|^{2} \, ds$$

$$+a_{2}\frac{1}{c'\lambda} \int_{t}^{T} e^{2\int_{0}^{s} a(u) \, du} |\bar{Z}_{x}(s)|^{2} \, ds. \qquad (4.13)$$

Choosing $c' = 2\delta a_2$, we obtain

$$|e^{\int_{0}^{t}a(u)\,du}\bar{Y}_{x}(t)|^{2} + (1 - \frac{1}{2\delta\lambda})\int_{t}^{T}e^{2\int_{0}^{s}a(u)\,du} < A(X(s))\bar{Z}_{x}(s), \bar{Z}_{x}(s) > ds$$

$$\leq e^{2\int_{0}^{T}a(u)\,du}|\bar{Y}_{x}(T)|^{2} - 2\int_{t}^{T}a(s)e^{2\int_{0}^{s}a(u)\,du}\bar{Y}_{x}(s) < \bar{Z}_{x}(s), dM_{x}(s) >$$
(4.14)

Taking expectation on both sides of the above inequality, we get that, for any t < T,

$$E_x[e^{2\int_0^t a(u)\,du}|\bar{Y}_x(t)|^2] \le E_x[e^{2\int_0^T a(u)\,du}|\bar{Y}_x(T)|^2].$$

Since both Y^1 and Y^2 satisfy the terminal condition in (4.8), we obtain that

$$\lim_{T \to \infty} E_x[e^{2\int_0^T a(u) \, du} |\bar{Y}_x(T)|^2] = 0,$$

which leads to $E_x[e^{2\int_0^t a(u) du} |\bar{Y}_x(t)|^2] = 0.$

We conclude that $Y_x^1(t) = Y_x^2(t)$ and $Z_x^1(t) = Z_x^2(t)$.

We now want to apply Lemma 4.2.1 to a particular situation.

Let $F(x, y, z) : \mathbb{R}^d \times \mathbb{R} \times \mathbb{R}^d \to \mathbb{R}$ be a Borel measurable function. Consider the following conditions:

(D.1) $(y_1 - y_2)(F(x, y_1, z) - F(x, y_2, z)) \le -d_1(x)|y_1 - y_2|^2,$ (D.2) $|F(x, y, z_1) - F(x, y, z_2)| \le d_2|z_1 - z_2|,$ (D.3) $|F(x, y, z)| \le |F(x, 0, z)| + K(x)(1 + |y|).$

Set $d(x) = -d_1(x) + \delta d_2^2$ for some constant $\delta > \frac{1}{2\lambda}$. The next result follows from Lemma 4.2.1.

Lemma 4.2.2. Assume the conditions (D.1)-(D.3) hold and

$$E_x\left[\int_0^\infty e^{2\int_0^t d(X(u))du} |F(X(t),0,0)|^2 dt\right] < \infty.$$

Then there exists a unique solution $(Y_x(t), Z_x(t))$ to the following equation:

$$Y_{x}(t) = Y_{x}(T) + \int_{t}^{T} F(X(s), Y_{x}(s), Z_{x}(s)) ds - \int_{t}^{T} \langle Z_{x}(s), dM_{x}(s) \rangle, \quad for \quad 0 \le t \le T; \lim_{t \to \infty} e^{\int_{0}^{t} d(X(u)) du} Y_{x}(t) = 0, \quad in \quad L^{2}(\Omega).$$
(4.15)

Consider the following condition instead of (D.3):

 $(\mathbf{D.3})' |F(X(t), y, z)| \le K(t)$, for any $y \in R$ and $z \in R^d$.

Let Φ be a bounded measurable function defined on ∂D , and $\tilde{q} \in L^p(D)$, for p > d. The following theorem is the main result in this section. **Theorem 4.2.3.** Assume the conditions (D.1), (D.2) and (D.3)' hold and that

$$E_{x_0} \left[\int_0^\infty e^{\int_0^s \tilde{q}(X(u)) \, du} \, dL_s \right] < \infty \tag{4.16}$$

for some $x_0 \in D$ and for $x \in D$,

$$E_x\left[\int_0^\infty e^{2\int_0^t d(X(u))\,du} \left\{ e^{2\int_0^t \tilde{q}(X(u))\,du} + |K(t)|^2 \right\} dt \right] < \infty.$$
(4.17)

Then there exists a unique solution (Y_x, Z_x) to the following BSDE:

$$Y_{x}(t) = Y_{x}(T) + \int_{t}^{T} F(X(s), Y_{x}(s), Z_{x}(s)) \, ds - \int_{t}^{T} e^{\int_{0}^{s} \tilde{q}(X(u)) dt} \Phi(X(s)) \, dL_{s} - \int_{t}^{T} \langle Z_{x}(s), dM_{x}(s) \rangle, \quad for \quad 0 \le t \le T,$$
(4.18)

and

$$\lim_{t \to \infty} e^{\int_0^t d(X(u)) \, du} Y_t = 0 \quad in \quad L^2(\Omega).$$
(4.19)

Proof. Uniqueness:

Suppose that (Y_x^1, Z_x^1) and (Y_x^2, Z_x^2) are two solutions of the equation (4.18) satisfying (4.19).

Set
$$\bar{Y}_x(t) = Y_x^1(t) - Y_x^2(t)$$
 and $\bar{Z}_x(t) = Z_x^1(t) - Z_x^2(t)$. Then

$$d(e^{\int_0^t d(X(u)) \, du} \bar{Y}_x(t))$$

$$= -e^{\int_0^t d(X(u)) \, du} (F(X(t), Y_x^1(t), Z_x^1(t)) - F(X(t), Y_x^2(t), Z_x^2(t))) dt$$

$$+ d(X(t)) e^{\int_0^t d(X(u)) \, du} \bar{Y}_x(t) dt + e^{\int_0^t d(X(u)) \, du} < \bar{Z}_x(t), dM_x(t) > .$$

By Ito's formula, we get, for any t < T,

$$e^{2\int_{0}^{t} d(X(u)) du} |\bar{Y}_{x}(t)|^{2} + \int_{t}^{T} e^{2\int_{0}^{t} d(X(u)) du} < A(X(s))\bar{Z}_{x}(s), \bar{Z}_{x}(s) > ds$$

$$= e^{2\int_{0}^{T} d(X(u)) du} |\bar{Y}_{x}(T)|^{2} + 2\int_{t}^{T} e^{2\int_{0}^{s} d(X(u)) du} \bar{Y}_{x}(s) (F(X(s), Y_{x}^{1}(s), Z_{x}^{1}(s)) - F(X(s), Y_{x}^{2}(s), Z_{x}^{2}(s))) ds$$

$$-2\int_{t}^{T} d(X(s)) e^{2\int_{0}^{s} d(X(u)) du} |\bar{Y}_{x}(s)|^{2} ds$$

$$-2\int_{t}^{T} d_{1}(X(s)) e^{2\int_{0}^{s} d(X(u)) du} \bar{Y}_{x}(s) < \bar{Z}_{x}(s), dM_{x}(s) >$$

$$(4.20)$$

By (D.1) and (D.2), we have

$$2\int_{t}^{T} e^{2\int_{0}^{s} d(X(u)) du} \bar{Y}_{x}(s) (F(X(s), Y_{x}^{1}(s), Z_{x}^{1}(s)) - F(X(s), Y_{x}^{2}(s), Z_{x}^{2}(s))) ds$$

$$= 2\int_{t}^{T} e^{2\int_{0}^{s} d(X(u)) du} \bar{Y}_{x}(s) (F(X(s), Y_{x}^{1}(s), Z_{x}^{1}(s)) - F(X(s), Y_{x}^{2}(s), Z_{x}^{1}(s))) ds$$

$$+ 2\int_{t}^{T} e^{2\int_{0}^{s} d(X(u)) du} \bar{Y}_{x}(s) (F(X(s), Y_{x}^{2}(s), Z_{x}^{1}(s)) - F(X(s), Y_{x}^{2}(s), Z_{x}^{2}(s))) ds$$

$$\leq -2\int_{t}^{T} d_{1}(X(s)) e^{2\int_{0}^{s} d(X(u)) du} |\bar{Y}_{x}(s)|^{2} ds + d_{2}\int_{t}^{\infty} e^{2\int_{0}^{s} d(X(u)) du} |\bar{Y}_{x}(s)|^{2} ds$$

$$\leq -2\int_{t}^{T} d_{1}(X(s)) e^{2\int_{0}^{s} d(X(u)) du} |\bar{Y}_{x}(s)|^{2} ds + cd_{2}\int_{t}^{T} e^{2\int_{0}^{s} d(X(u)) du} |\bar{Y}_{x}(s)|^{2} ds$$

$$+ d_{2}\frac{1}{c\lambda}\int_{t}^{T} e^{2\int_{0}^{s} d(X(u)) du} |\bar{Z}_{x}(s)|^{2} ds. \qquad (4.21)$$

Choosing $c = 2\delta d_2$, from (4.21) we obtain that

$$|e^{\int_{0}^{t} d(X(u)) \, du} \bar{Y}_{x}(t)|^{2} + (1 - \frac{1}{2\delta\lambda}) \int_{t}^{T} e^{2\int_{0}^{t} d(X(u)) \, du} < A(X(s)) \bar{Z}_{x}(s), \bar{Z}_{x}(s) > ds$$

$$\leq e^{2\int_{0}^{T} d(X(u)) \, du} |\bar{Y}_{x}(T)|^{2} - 2\int_{t}^{T} d(X(s)) e^{2\int_{0}^{s} d(X(u)) \, du} \bar{Y}_{x}(s) < \bar{Z}_{x}(s), dM_{x}(s) > .$$

$$(4.22)$$

Taking expectation on both sides of the above inequality and letting T tend to ∞ , we get that

$$E_x[e^{2\int_0^t d(X(u))\,du}|\bar{Y}_x(t)|^2] = 0$$

We conclude that $Y_x^1(t) = Y_x^2(t)$ and hence from (4.22), $Z_x^1(t) = Z_x^2(t)$.

Existence:

First of all, the assumption (4.16) and the following Lemma 4.4.3 imply that

$$\sup_{x} E_{x}\left[\int_{0}^{\infty} e^{\int_{0}^{s} \tilde{q}(X(u)) \, du} \, dL_{s}\right] < \infty.$$

1°: There exists $(p_x(t), q_x(t))$ such that

$$dp_x(t) = e^{\int_0^t \tilde{q}(X(u)) \, du} \Phi(X(t)) dL_t + \langle q_x(t), dM_x(t) \rangle, \qquad (4.23)$$

and $e^{\int_0^t d(X(u)) \, du} p_x(t) \to 0$ as $t \to \infty$, in $L^2(\Omega)$.

In fact, let

$$p_{x}(t) := -E_{x}\left[\int_{t}^{\infty} e^{\int_{0}^{s} \tilde{q}(X(u)) \, du} \Phi(X(s)) \, dL_{s} | \mathcal{F}_{t}\right]$$

$$= \int_{0}^{t} e^{\int_{0}^{s} \tilde{q}(X(u)) \, du} \Phi(X(s)) L_{s} - E_{x}\left[\int_{0}^{\infty} e^{\int_{0}^{s} \tilde{q}(X(u)) \, du} \Phi(X(s)) \, dL_{s} | \mathcal{F}_{t}\right].$$
(4.24)

By the martingale representation theorem in [42], there exists a process $q_x(t)$, such that

$$-E_{x}\left[\int_{0}^{\infty} e^{\int_{0}^{s} \tilde{q}(X(u)) \, du} \Phi(X(s)) \, dL_{s} | \mathcal{F}_{t} \right] = - E_{x}\left[\int_{0}^{\infty} e^{\int_{0}^{s} \tilde{q}(X(u)) \, du} \Phi(X(s)) \, dL_{s} \right] + \int_{0}^{t} \langle q_{x}(s), dM_{x}(s) \rangle .$$

$$(4.25)$$

Then (p_x, q_x) satisfies the equation (4.23).

Moreover,

$$p_{x}(t) = -E_{x}\left[\int_{t}^{\infty} e^{\int_{0}^{s} \tilde{q}(X(u)) \, du} \Phi(X(s)) \, dL_{s} | \mathcal{F}_{t}\right] \\ = -e^{\int_{0}^{t} \tilde{q}(X(u)) \, du} E_{x}\left[\int_{t}^{\infty} e^{\int_{t}^{s+t} \tilde{q}(X(u)) \, du} \Phi(X(s)) \, dL_{s} | \mathcal{F}_{t}\right] \\ = -e^{\int_{0}^{t} \tilde{q}(X(u)) \, du} E_{x}\left[\int_{0}^{\infty} e^{\int_{0}^{s+t} \tilde{q}(X(u)) \, du} \Phi(X(s+t)) \, dL_{s+t} | \mathcal{F}_{t}\right] \\ = -e^{\int_{0}^{t} \tilde{q}(X(u)) \, du} E_{x}\left[\int_{0}^{\infty} e^{\int_{0}^{s} \tilde{q}(X(u+t)) \, du} \Phi(X(s+t)) \, dL_{s+t} | \mathcal{F}_{t}\right] \\ = -e^{\int_{0}^{t} \tilde{q}(X(u)) \, du} E_{X(t)}\left[\int_{0}^{\infty} e^{\int_{0}^{t} \tilde{q}(X(u)) \, du} \Phi(X(l)) \, dL_{l}\right].$$
(4.26)

The last equality follows from the fact that $L_{t+s} = L_t + L_s \circ \theta_t$. Therefore,

$$\sup_{x} |p_{x}(t)| \le e^{\int_{0}^{t} \tilde{q}(X(u)) \, du} \sup_{x \in D} |\Phi(x)| \cdot \sup_{x \in \bar{D}} E_{x} [\int_{0}^{\infty} e^{\int_{0}^{t} \tilde{q}(X(u)) \, du} dL_{t}].$$

Set $M = \sup_{x \in D} |\Phi(x)| \cdot \sup_{x \in \overline{D}} E_x [\int_0^\infty e^{\int_0^t \tilde{q}(X(u)) \, du} dL_t].$ By the following Lemma 4.4.3 and condition (4.16), it follows that $M < \infty$. In view of (4.17), we have

$$\lim_{t \to \infty} e^{\int_0^t (d+\tilde{q})(X(u)) \, du} = 0$$

in $L^2(\Omega)$. Hence,

$$e^{\int_0^t d(X(u)) \, du} p_x(t) \le M e^{\int_0^t (d+\tilde{q})(X(u)) \, du} \to 0 \quad as \quad t \to \infty, \quad in \quad L^2(\Omega).$$
(4.27)

 2° : Set $g(t, y, z) = F(X(t), p_x(t) + y, q_x(t) + z)$. Then

$$(y_1 - y_2)(g(t, y_1, z) - g(t, y_2, z))$$

$$= (y_1 - y_2)(F(X(t), p_x(t) + y_1, q_x(t) + z) - F(X(t), p_x(t) + y_2, q_x(t) + z))$$

$$\leq -d_1(X(t))|y_1 - y_2|^2, \qquad (4.28)$$

and

$$|g(t, y, z_1) - g(t, y, z_2)|$$

$$= |F(X(t), p_x(t) + y, q_x(t) + z_1) - F(X(t), p_x(t) + y, q_x(t) + z_1)|$$

$$\leq d_2|z_1 - z_2|.$$

Moreover,

$$E_{x}\left[\int_{0}^{\infty} e^{2\int_{0}^{t} d(X(u)) \, du} |g(X(t), 0, 0)|^{2} \, dt\right]$$

$$\leq E_{x}\left[\int_{0}^{\infty} e^{2\int_{0}^{t} d(X(u)) \, du} |F(X(t), p_{x}(t), q_{x}(t))|^{2} \, dt\right]$$

$$\leq E_{x}\left[\int_{0}^{\infty} e^{2\int_{0}^{t} d(X(u)) \, du} |K(t)|^{2} \, dt\right]$$

$$< \infty.$$
(4.29)

Therefore, g satisfies all the conditions of the Lemma 4.2.1. So there exists a pair of processes (k_x, l_x) such that

$$dk_x(t) = -g(t, k_x(t), l_x(t)) dt + \langle l_x(t), dM_x(t) \rangle,$$

and

$$e^{\int_0^t d(X(u)) \, du} k_x(t) \to 0,$$

as $t \to \infty$.

Putting $Y_x(t) = p_x(t) + k_x(t)$ and $Z_x(t) = q_x(t) + l_x(t)$, we find that $(Y_x(t), Z_x(t))$ satisfies the following equation

$$dY_x(t) = e^{\int_0^t \tilde{q}(X(u)) \, du} \phi(X(t)) \, dL_t - F(t, Y_x(t), Z_x(t)) \, dt + \langle Z_x(t), dM_x \rangle$$

and

$$\lim_{t \to \infty} e^{\int_0^t d(X(u)) \, du} Y_t = 0.$$

Corollary 4.2.4. Assume all of the assumptions in Theorem 4.2.3 are satisfied. If

$$\sup_{x} E_{x} \left[\int_{0}^{\infty} e^{2\int_{0}^{s} d(X(u)) \, du} K^{2}(s) \, ds < \infty, \right]$$
(4.30)

then it holds that

$$\sup_{x\in D}|Y_x(0)|<\infty.$$

Proof. As shown in the proof of theorem, $Y_x(t)$ has the decomposition: $Y_x(t) = p_x(t) + k_x(t)$.

Firstly, set t = 0 in (4.26), it follows that

$$|p_{x}(0)| \leq E_{X(t)}[|\int_{0}^{\infty} e^{\int_{0}^{l} \tilde{q}(X(u)) \, du} \Phi(X(l)) dL_{l}|] \\ \leq \sup_{x} ||\Phi||_{\infty} \sup_{x} E_{x}[\int_{0}^{\infty} e^{\int_{0}^{l} \tilde{q}(X(u)) \, du} dL_{l}] \\ < \infty.$$
(4.31)

Secondly, by Ito's formula, we obtain

$$de^{2\int_{0}^{t} d(X(u)) \, du} |k_{x}(t)|^{2} = -2e^{2\int_{0}^{t} d(X(u)) \, du} k_{x}(t)g(t, k_{x}(t), l_{x}(t)) \, dt$$

+2e^{2\int_{0}^{t} d(X(u)) \, du} k_{x}(t)d(X(t)) \, dt
+2e^{2\int_{0}^{t} d(X(u)) \, du} k_{x}(t) < l_{x}(t), \, dM_{x}(t) >
+e^{2\int_{0}^{t} d(X(u)) \, du} < A(X(t))l_{x}(t), l_{x}(t) > dt.

By further calculation and choosing two positive numbers δ_1 and δ_2 such that $\delta_1 > \frac{1}{2\lambda}$ and $\delta_1 + \delta_2 < \delta$, we obtain that, for any t < T,

$$E_{x}[e^{2\int_{0}^{t}d(X(u))\,du}|k_{x}(t)|^{2}] + \frac{1}{\lambda}(1 - \frac{1}{2\lambda\delta_{1}})E_{x}[\int_{t}^{T}e^{2\int_{0}^{s}d(X(u))\,du}|l_{x}(s)|^{2}\,ds]$$

$$\leq E_{x}[e^{2\int_{0}^{T}d(X(u))\,du}|k_{x}(T)|^{2}] + \frac{1}{2\delta_{2}d_{2}^{2}}E_{x}[\int_{t}^{T}e^{2\int_{0}^{s}d(X(u))\,du}|g(s,0,0)|^{2}\,ds].$$

Setting t = 0, we get

$$|k_x(0)|^2 = E_x[|k_x(0)|^2] \leq E_x[e^{2\int_0^T d(X(u)) \, du} |k_x(T)|^2] + \frac{1}{2\delta_2 d_2^2} E_x[\int_0^T e^{2\int_0^s d(X(u)) \, du} |g(s, 0, 0)|^2 \, ds].$$

Because T > t is chosen arbitrary, it follows that

$$|k_x(0)|^2 \le \frac{1}{2\delta_2 d_2^2} E_x \left[\int_0^\infty e^{2\int_0^s d(X(u)) \, du} |g(s,0,0)|^2 \, ds \right].$$

Therefore, by condition (4.30), it follows that

$$\sup_{x} |k_{x}(0)| \leq \left(\frac{1}{2\delta_{2}d_{2}^{2}} \sup_{x} E_{x}\left[\int_{0}^{\infty} e^{2\int_{0}^{s} d(X(u)) \, du} |g(s,0,0)|^{2} \, ds\right]\right)^{\frac{1}{2}} < \infty.$$

So that $\sup_{x} |Y_x(0)| \le \sup_{x} |p_x(0)| + \sup_{x} |k_x(0)| < \infty.$

4.3 Linear PDEs

Given an operator

$$G = \frac{1}{2}\nabla \cdot (A\nabla) + b \cdot \nabla + q,$$

satisfying the Neumann boundary condition and with the domain $\mathcal{D}(G) \subset W^{1,2}(D)$ densely, where D is a bounded smooth domain in \mathbb{R}^d .

Here $b = (b_1, ..., b_d)$ is a R^d -valued Borel measurable function, and q is a Borel measurable function on R^d such that:

$$I_D(|b|^2 + |q|) \in L^p(D), \quad p > d.$$

In this section, we solve the following linear boundary value problem:

$$\begin{cases} \frac{1}{2}\nabla \cdot (A\nabla u)(x) + b \cdot \nabla u(x) + q(x)u(x) = F(x) & \text{on } D, \\ \frac{1}{2}\frac{\partial u}{\partial \vec{\gamma}}(x) = \phi & \text{on } \partial D, \end{cases}$$
(4.32)

where F and ϕ are bounded measurable functions on D and ∂D respectively.

By Green's identity, it is known that operator G defined on a bounded domain D with Neumann boundary condition $\frac{\partial}{\partial \bar{\gamma}}(x) = 0$ is associated with a quadratic form ([15], [30]):

$$\begin{aligned} \mathcal{E}(f,g): &= -\int_D Gf(x)g(x)dx \\ &= \frac{1}{2}\int_D \langle A\nabla f, \nabla g \rangle dx - \int_D b \cdot \nabla f(x)g(x)dx - \int_D q(x)f(x)g(x)dx. \end{aligned}$$

Definition 4.3.1. A bounded continuous function u defined on D is a weak solution of the problem (4.32) if $u \in W^{1,2}(D)$, and for any $g \in C^{\infty}(\overline{D})$,

$$\mathcal{E}(u,g) = -\int_{\partial D} \phi(x)g(x)\lambda(dx) - \int_{D} F(x)g(x)dx,$$

where λ denotes the d-1 dimensional Lebesgue measure on ∂D .

Consider the operator

$$G_1 = \frac{1}{2} \nabla \cdot (A \nabla u) \tag{4.33}$$

on domain D with boundary condition $\frac{\partial u}{\partial \vec{\gamma}} = 0$ on ∂D .

 G_1 is associated with a reflecting diffusion process (X^0, P_x^0) . By [26] and Example 1, X^0 has the following decomposition:

$$dX_t^0 = \sigma(X_t^0) \, dW_t + \frac{1}{2} \nabla A(X_t^0) \, dt + \vec{\gamma}(X_t^0) \, dL_t^0,$$

$$L_t^0 = \int_0^t I_{\{X_s^0 \in \partial D\}} \, dL_s^0,$$
 (4.34)

where the matrix $\sigma(x)$ is the positive definite symmetric square root of the matrix A(x) and $\{W_t\}_{t\geq 0}$ is a d-dimensional standard Brownian motion.

It is well known that operator G_1 is associated with the regular Dirichlet form

([15]):

$$\mathcal{E}^{0}(u,v) = \frac{1}{2} \int_{D} a_{ij} \frac{\partial u}{\partial x_{i}} \frac{\partial v}{\partial x_{j}} dx$$

and the domain of \mathcal{E}^0 is $W^{1,2}(D)$.

The following lemma can be proved similarly as the Corollary 3.8 in [21] using the heat kernel estimates on $p_1(t, x, y)$ associated with G_1 in Chapter 3.

Lemma 4.3.2. There exists a constant K > 0, such that

$$\sup_{x\in\bar{D}} E_x^0[L_t^0] \le K\sqrt{t} \quad and \quad \inf_{x\in\bar{D}} E_x^0[L_t^0] > 0$$

Moreover, for all positive integer n, we have $\sup_{x\in \overline{D}} E_x^0[(L_t^0)^n] \leq K_n t^{\frac{n}{2}}$, for some constant $K_n > 0$.

Set
$$M_t^0 = \int_0^t \sigma(X_s^0) \, dW_s$$
 and

$$Z_t = e^{\int_0^t \langle A^{-1}b(X_s^0), dM_s^0 \rangle - \frac{1}{2} \int_0^t b^* A^{-1}b(X_s^0) \, ds + \int_0^t q(X_s^0) \, ds},$$
(4.35)

where b^* is the transpose of the row vector b.

The proof of the following two Lemmas are inspired by that of the Lemma 2.1 and Theorem 2.2 in [21].

Lemma 4.3.3. For t > 0, there are two strictly positive functions $M_1(t)$ and $M_2(t)$ such that, for any $x \in \overline{D}$, $M_1(t) \leq E_x^0 [\int_0^t Z_s dL_s^0] \leq M_2(t)$. Furthermore, $M_2(t) \to 0$ as $t \to 0$.

Proof. 1°: Put

$$\tilde{M}(t) = e^{\int_0^t \langle A^{-1}b(X_s^0), dM_s^0 \rangle - \frac{1}{2} \int_0^t b^* A^{-1}b(X_s^0) \, ds}, \qquad (4.36)$$
$$e_q(t) = e^{\int_0^t q(X_s^0) \, ds}.$$

Then by martingale inequality and the fact that L^0 is increasing and continuous,

we have

$$\begin{split} \sup_{x \in \overline{D}} E_x^0 [\int_0^t Z_s \, dL_s^0] &= \sup_{x \in \overline{D}} E_x^0 [\int_0^t \tilde{M}(s) e_q(s) \, dL_s^0] \\ &\leq \sup_{x \in \overline{D}} E_x^0 [\max_{0 \le s \le t} |\tilde{M}(s)|^2]^{\frac{1}{2}} \cdot \sup_{x \in \overline{D}} E_x^0 [e_{2|q|}(t) (L_t^0)^2]^{\frac{1}{2}} \\ &\leq C \sup_{x \in \overline{D}} E_x^0 [|\tilde{M}_t|^2]^{\frac{1}{2}} \cdot \sup_{x \in \overline{D}} E_x^0 [e_{4|q|}(t)]^{\frac{1}{4}} \cdot \sup_{x \in \overline{D}} E_x^0 [(L_t^0)^4]^{\frac{1}{4}} \\ &\quad \underbrace{\sum_{x \in \overline{D}} (I) \quad \underbrace{\sum_{x \in \overline{D} (I) \quad \underbrace{\sum_{x \in \overline{D}} (I) \quad \underbrace{\sum_{x \in \overline{D} (I)$$

By the fact $|b|^2$, $|q| \in L^p$ and Theorem 2.1 in [27], (I) and (II) are bounded if t belongs to a bounded interval.

Because of $E_x^0[(L_t^0)^n] \leq K_n t^{\frac{n}{2}}$, we see that $M_2(t) := K(I)(II)\sqrt{t}$ is the required upper bound.

 2° : Since

$$E_x^0[L_t^0]^2 \le E_x^0[\int_0^t \tilde{M}^{-1}(s)e_{-q}(s)\,dL_s^0] \cdot E_x^0[\int_0^t \tilde{M}(s)e_q(s)\,dL_s^0],\tag{4.37}$$

we obtain

$$E_x^0[\int_0^t \tilde{M}(s)e_q(s) \, dL_s^0] \ge \frac{E_x^0[L_t^0]^2}{E_x^0[\int_0^t \tilde{M}^{-1}(s)e_{-q}(s) \, dL_s^0]}.$$
(4.38)

Here

$$\tilde{M}^{-1}(t) = e^{-\int_0^t \langle A^{-1}b(X_s^0), dM_s^0 \rangle + \frac{1}{2}\int_0^t b^* A^{-1}b(X_s^0) \, ds}$$

$$= e^{-\int_0^t \langle A^{-1}b(X_s), dM_s^0 \rangle - \frac{1}{2}\int_0^t b^* A^{-1}b(X_s^0) \, ds} \cdot e^{\int_0^t b^* A^{-1}b(X_s^0) \, ds}$$

$$:= N(t) \cdot e^{\int_0^t b^* A^{-1}b(X_s^0) \, ds}$$
(4.39)

By the proof of the first part, replacing \tilde{M}_t , q by N_t and $b^*A^{-1}b - q$ respectively, it can be seen that there exists K(t) > 0 such that $\sup_{x \in \overline{D}} E^0_x [\int_0^t \tilde{M}^{-1}(s) e_{-q}(s) dL^0_s] \le K(t)$.

As $\inf_{x\in\overline{D}} E_x^0[L_t^0] > 0$, we complete the proof of the lemma by setting $M_1(t) = \frac{\inf_{x\in\overline{D}} E_x^0[L_t^0]^2}{K(t)}$.

Set
$$L(x) := E_x^0 [\int_0^\infty Z_s \, dL_s^0].$$

Lemma 4.3.4. If there is a point $x_0 \in \overline{D}$, such that $L(x_0) < \infty$, then there are two positive constants K and β such that $\sup_{x\in\overline{D}} E_x^0[Z_t] \leq Ke^{-\beta t}$.

Proof. By Girsanov Theorem and Feynmann-Kac formula, $G = \frac{1}{2}\nabla \cdot (A\nabla) + b \cdot \nabla + q$ is associated with the semigroup $\{S_t\}_{t>0}$, where $S_t f(x) = E_x^0[Z_t f(X_t^0)]$ for $f \in L^2(D)$.

By the upper and lower bound estimates of the heat kernel p(t, x, y) associated with S_t in [37] and Chapter 3, without losing generality, we assume that

$$m_1^{-1}e^{-m_3|x-y|^2} \le p(1,x,y) \le m_1 e^{-m_2|x-y|^2}.$$

So that we have

$$p(1, x, y) \le m_1 e^{-m_2|x-y|^2} \le m_1,$$

and

$$p(1, x, y) \ge m_1^{-1} e^{-m_3|x-y|^2} \ge m_1^{-1} e^{-m_3(\sup_{x,y\in D} |x-y|^2)}.$$

The above two estimates imply that, for any positive function $f \in L^2(D)$,

$$c^{-1} \int_{D} f(x) dx \le E_x^0 [Z_1 f(X_1^0)] = \int_{D} p(1, x, y) f(y) dy \le c \int_{D} f(x) dx, \qquad (4.40)$$

where c is a positive constant. Since

$$L(x) = \sum_{n=0}^{\infty} E_x^0 [Z_n E_{X_n^0}^0 [\int_0^1 Z_s L^0(ds)]] \ge M_1(1) \sum_{n=0}^{\infty} E_x^0 [Z_n]$$

and $L(x_0) < \infty$, there is a positive integer number N such that

$$\frac{1}{2c^2} \ge E_{x_0}^0[Z_N] = E_{x_0}^0[Z_1 E_{X_1}^0[Z_{N-1}]] \ge c^{-1} \int_D E_x^0[Z_{N-1}]m(dx).$$

This implies

$$\int_{D} E_x^0 [Z_{N-1}] m(dx) \le \frac{1}{2c}.$$

Thus

$$\sup_{x\in\overline{D}} E_x^0[Z_N] = \sup_{x\in\overline{D}} E_x^0[Z_1 E_{X_1}^0[Z_{N-1}]] \le c \int_D E_x^0[Z_{N-1}]m(dx) \le \frac{1}{2}.$$
 (4.41)

For any t > 0, there exists a positive number n such that $\frac{t}{N} \in [n-1, n)$. Then by (4.41), it follows that

$$E_x^0[Z_t] \le \frac{1}{2^{n-1}} E_x^0[Z_{t-N(n-1)}] \le \left(\sup_{x \in D, 0 \le t \le N} E_x^0[Z_t]\right) \frac{1}{2^{n-1}} \\ \le 2 \sup_{x \in D, 0 \le t \le N} E_x^0[Z_t] e^{-\frac{\ln 2}{N}t}.$$

Theorem 4.3.5. If there exists $x_0 \in \overline{D}$ such that $L(x_0) < \infty$, then there exists a unique bounded continuous weak solution of the problem (4.32).

Proof. Existence :

Due to Theorem 3.2 in [8], there exists a unique, bounded, continuous weak solution u_2 of the following problem:

$$\begin{cases} Gu_2(x) = 0, & \text{on } D, \\ \frac{1}{2} \frac{\partial u_2}{\partial \vec{\gamma}}(x) = \phi, & \text{on } \partial D. \end{cases}$$

$$(4.42)$$

Thus by the linearity of the problem (4.32), we only need to show that the following problem has a bounded continuous weak solution:

$$\begin{cases} Gu_1(x) = F(x), & \text{on } D, \\ \frac{\partial u_1}{\partial \vec{\gamma}}(x) = 0, & \text{on } \partial D. \end{cases}$$

$$(4.43)$$

The semigroup associated with operator G is $\{S_t, t > 0\}$. By Lemma 4.3.4, we

have

$$\sup_{x \in D} |S_t F(x)| = \sup_{x \in D} |E_x^0[Z_t F(X_t^0)]| \le K e^{-\beta t} ||F||_{\infty}.$$

Then

$$u_1(x) := \int_0^\infty S_t F(x) dt$$

is well defined and has the following bound:

$$\sup_{x \in D} |u_1(x)| \le \frac{K}{\beta} ||F||_{\infty}.$$

Moreover, the function $u_1(x)$ is also continuous on D. In fact, fixing any $x \in D$ and $\epsilon > 0$, we can firstly choose a constant $t_0 > 0$, such that $\sup_{z \in D} |\int_0^{t_0} S_s F(z) ds| < \frac{\epsilon}{3}$. And because $S_{t_0}u_1(x)$ is continuous, there exists a constant $\delta > 0$, such that for any y with $|y - x| < \delta$, $|S_{t_0}u_1(x) - S_{t_0}u_1(y)| \le \frac{\epsilon}{3}$.

We find that

$$S_{t}u_{1}(x) = E_{x}^{0}[Z_{t}u_{1}(X_{t}^{0})] = E_{x}^{0}[Z_{t}\int_{0}^{\infty} E_{X_{t}^{0}}[Z_{s}F(X_{s}^{0})] ds]$$

$$= \int_{0}^{\infty} E_{x}^{0}[Z_{t+s}F(X_{t+s}^{0})] ds$$

$$= \int_{t}^{\infty} S_{s}F(x) ds$$

$$= u_{1}(x) - \int_{0}^{t} S_{s}F(x) ds. \qquad (4.44)$$

For any y satisfying $|y - x| < \delta$, it follows that

$$|u_1(x) - u_1(y)| \le |S_{t_0}u_1(x) - S_{t_0}u_1(y)| + |\int_0^{t_0} S_s F(x) \, ds| + |\int_0^{t_0} S_s F(y) \, ds| \le \epsilon.$$

This implies that the function u_1 is continuous on domain D.

Denote the resolvent associated with operator G by $\{G_{\beta}, \beta > 0\}$. Note that

$$G_{\beta}u_{1}(x) = \int_{0}^{\infty} e^{-\beta t} S_{t}u_{1}(x)dt$$

$$= \int_{0}^{\infty} e^{-\beta t}u_{1}(x)dt - \int_{0}^{\infty} e^{-\beta t} \int_{0}^{t} S_{s}F(x) dsdt$$

$$= \frac{1}{\beta}u_{1}(x) - \int_{0}^{\infty} \int_{0}^{t} e^{-\beta t} S_{s}F(x) dsdt$$

$$= \frac{1}{\beta}u_{1}(x) - \int_{0}^{\infty} S_{s}F(x)(\int_{s}^{\infty} e^{-\beta t} dt) ds$$

$$= \frac{1}{\beta}u_{1}(x) - \frac{1}{\beta}G_{\beta}F(x). \qquad (4.45)$$

We have

$$\beta(u_1(x) - \beta G_\beta u_1(x)) = \beta G_\beta F(x)$$

Therefore,

$$\lim_{\beta \to \infty} \int_D \beta(u_1(x) - \beta G_\beta u_1(x)) u_1(x) dx = \lim_{\beta \to \infty} \int_D \beta G_\beta F(x) u_1(x) dx$$
$$= \int_D F(x) u_1(x) dx < \infty.$$

This implies that $u_1 \in D(\mathcal{E})$ (see [30]) and u_1 is a weak solution of equation (4.43). By the linearity, $u = u_1 + u_2$ is a bounded continuous weak solution of equation (4.32). Uniqueness :

Let v_1 and v_2 be two bounded continuous weak solutions of the equation (4.32). Then $v_1 - v_2$ is the solution of equation (4.42) with $\phi = 0$. Then by the uniqueness of solutions to the equation (4.42) proved in [8], we know that $v_1 = v_2$.

4.4 Semilinear PDEs

In this section, we solve the following semilinear boundary value problem:

$$\begin{cases} \frac{1}{2}\nabla \cdot (A\nabla u)(x) + b \cdot \nabla u(x) + q(x)u(x) = -H(x, u(x), \nabla u(x)), & \text{on } D, \\ \frac{1}{2}\frac{\partial u}{\partial \vec{\gamma}}(x) = \phi(x) & \text{on } \partial D. \end{cases}$$
(4.46)

Recall $\mathcal{E}(\cdot, \cdot)$ is the quadratic form associated with the operator $G = \frac{1}{2} \nabla \cdot (A \nabla) + b \cdot \nabla + q$. Then

$$\mathcal{E}(u,v) = \frac{1}{2} \int_D \langle A\nabla u, \nabla v \rangle dx - \int_D \langle b, \nabla u \rangle v dx - \int_D quv dx.$$

Definition 4.4.1. A bounded continuous function u(x) defined on D is called a weak solution of the equation (4.46) if $u \in W^{1,2}(D)$, and for any $g \in C^{\infty}(\overline{D})$,

$$\mathcal{E}(u,g) = \int_{\partial D} \phi(x)g(x)\lambda(dx) + \int_{D} H(x,u(x),\nabla u(x))g(x)dx.$$

Recall that L_t is the boundary local time of X(t) defined in (4.7) and L_t^0 is the boundary local time of X_t^0 in (4.34).

As a consequence of the Girsanov theorem, we have the following lemma.

Lemma 4.4.2. Assume the function f satisfies $E_x[\int_0^T e^{\int_0^t f(X(u)) du} dL_t] < \infty$. Then it holds that

$$E_x[\int_0^T e^{\int_0^t f(X(u)) \, du} dL_t] = E_x^0[\int_0^T \tilde{M}_t e^{\int_0^t f(X_u^0) \, du} dL_t^0],$$

where \tilde{M}_t was defined in (4.36).

Proof. Because the boundary ∂D is smooth, it is not difficult to see that there exists a function $g \in C^2(\overline{D})$ with $\frac{\partial g}{\partial \overline{\gamma}}(x) = 1$ if $x \in \partial D$. By Ito's formula, we get

$$g(X_T^0) = g(X_0^0) + \int_0^T \langle \nabla g(X_s^0), dM_s^0 \rangle + \int_0^T (G_1g)(X_s^0) \, ds + L_T^0.$$

And by the chain rule, we obtain

$$e^{\int_0^T f(X_u^0) \, du} g(X_T^0) = g(X_0^0) + \int_0^T e^{\int_0^s f(X_u^0) \, du} < \nabla g(X_s^0), dM_s^0 > \\ + \int_0^T e^{\int_0^s f(X_u^0) \, du} (fg + G_1g)(X_s^0) \, ds + \int_0^T e^{\int_0^s f(X_u^0) \, du} \, dL_s^0.$$

Finally, it follows that

$$\begin{split} \tilde{M}_{T}e^{\int_{0}^{T}f(X_{u}^{0})\,du}g(X_{T}^{0}) \\ &= g(X_{0}^{0}) + \int_{0}^{T}\tilde{M}_{s}e^{\int_{0}^{s}f(X_{u}^{0})\,du} < \nabla g(X_{s}^{0}), dM_{s}^{0} > \\ &+ \int_{0}^{T}\tilde{M}_{s}e^{\int_{0}^{s}f(X_{u}^{0})\,du}(fg + G_{1}g)(X_{s}^{0})\,ds + \int_{0}^{T}\tilde{M}_{s}e^{\int_{0}^{s}f(X_{u}^{0})\,du}\,dL_{s}^{0} \\ &+ \int_{0}^{T}e^{\int_{0}^{s}f(X_{u}^{0})\,du}g(X_{s}^{0})d\tilde{M}_{s} + \int_{0}^{T}\tilde{M}_{s}e^{\int_{0}^{s}f(X_{u}^{0})\,du} < b, \nabla g > (X_{s}^{0})\,ds. \end{split}$$

Taking expectations on both sides of the above equation, we obtain

$$E_x^0[\tilde{M}_T e^{\int_0^T f(X_u^0) \, du} g(X_T^0)] = g(x) + E_x^0[\int_0^T \tilde{M}_s e^{\int_0^s f(X_u^0) \, du} (fg + G_2 g)(X_s^0) \, ds] + E_x^0[\int_0^T \tilde{M}_s e^{\int_0^s f(X_u^0) \, du} \, dL_s^0].$$
(4.47)

On the other hand, by the Girsanov theorem and Ito's formula, we obtain

$$E_{x}^{0}[\tilde{M}_{T}e^{\int_{0}^{T}f(X_{u}^{0})\,du}g(X_{T}^{0})]$$

$$= E_{x}[e^{\int_{0}^{T}f(X(u))\,du}g(X(T))]$$

$$= g(x) + E_{x}[\int_{0}^{T}e^{\int_{0}^{s}f(X(u))\,du}(fg + G_{2}g)(X(s))\,ds]$$

$$+ E_{x}[\int_{0}^{T}e^{\int_{0}^{s}f(X(u))\,du}\,dL_{s}]$$
(4.48)

Comparing the formulas (4.47) and (4.48), we get the final result in this lemma.

Lemma 4.4.3. Suppose that the function $\tilde{q} \in L^p(D)$ and p > d. If there exists some point $x_0 \in D$, such that

$$E_{x_0}[\int_0^\infty e^{\int_0^t \tilde{q}(X(u)) \, du} dL_t] < \infty, \tag{4.49}$$

it holds that

$$\sup_{x\in D} E_x [\int_0^\infty e^{\int_0^t \tilde{q}(X(u))\,du} dL_t] < \infty.$$

Proof. Due to the Theorem 3.2 in [8] and the condition (4.49), there is a continuous bounded function $f \in W^{1,2}(D)$, such that

$$\frac{1}{2}\nabla\cdot(A\nabla f) + b\cdot\nabla f + \tilde{q}f = 0$$

on domain D and $\frac{1}{2} \frac{\partial f}{\partial \vec{\gamma}} = 1$ on ∂D .

Moreover f has the following expression:

$$f(x) = E_x^0 \left[\int_0^\infty e^{\int_0^t \langle A^{-1}b(X^0(u)), dM_u^0 \rangle - \frac{1}{2} \int_0^t b^* A^{-1}b(X_s^0) \, ds + \int_0^t \tilde{q}(X_u^0) \, du} dL_t^0 \right].$$

By the Lemma 4.4.2, it follows that

$$f(x) = E_x\left[\int_0^\infty e^{\int_0^t \tilde{q}(X(u)) \, du} dL_t\right]$$

Then the lemma is proved, for f is bounded on the domain \overline{D} .

Let $H: \mathbb{R}^d \times \mathbb{R} \times \mathbb{R}^d \to \mathbb{R}$ be a bounded Borel measurable function. Introduce the following conditions:

(H.1) $(y_1 - y_2)(H(x, y_1, z) - H(x, y_2, z)) \leq -h_1(x)|y_1 - y_2|^2,$ (H.2) $|H(x, y, z_1) - H(x, y, z_2)| \leq h_2|z_1 - z_2|.$ Set $h(t) = -h_1(X(t)) + \delta h_2^2 + q(X(t))$ and $\tilde{h}(t) = -h_1(X(t)) + \delta h_2^2$ for some constant $\delta > \frac{1}{2\lambda}.$

Theorem 4.4.4. Suppose that the conditions (H.1) and (H.2) are satisfied. Assume

$$E_{x_1}\left[\int_0^\infty e^{2\int_0^t (q(X(u)) + \tilde{h}(u)) \, du} dt\right] < \infty, \quad for \quad some \quad x_1 \in D, \tag{4.50}$$

and there exists some point $x_0 \in D$, such that

$$E_{x_0}[\int_0^\infty e^{\int_0^t q(X(u)) \, du} dL_t] < \infty.$$
(4.51)

Then the semilinear Neumann boundary value problem (4.46) has a unique continuous weak solution.

Proof. Set

$$\tilde{H}(t, x, y, z) := e^{\int_0^t q(X(u))dt} H(x, e^{-\int_0^t q(X(u))dt} y, e^{-\int_0^t q(X(u))dt} z).$$

Then

$$(y_1 - y_2)(\tilde{H}(t, X(t), y_1, z) - \tilde{H}(t, X(t), y_2, z)) \le -h_1(X(t))|y_1 - y_2|^2$$
(4.52)

and

$$|\tilde{H}(t, X(t), y, z_1) - \tilde{H}(t, X(t), y, z_2)| \le h_2 |z_1 - z_2|.$$
(4.53)

Using the fact that H is bounded, we can show that

$$\tilde{H}(t, X(t), y, z) \le e^{\int_0^t q(X(u))dt} \|H\|_{\infty}.$$

By Theorem 4.2.3, there exists a unique process (\hat{Y}_x, \hat{Z}_x) satisfying

$$\begin{split} d\hat{Y}_x(t) &= -\tilde{H}(t, X(t), \hat{Y}_x(t), \hat{Z}_x(t))dt + e^{\int_0^t q(X(u))\,du}\phi(X(t))dL(t) + <\hat{Z}_x(t), dM_x(t) > \\ e^{\int_0^t \tilde{h}(u)\,du}\hat{Y}_x(t) \to 0 \quad as \quad t \to \infty. \end{split}$$

Furthermore, Corollary 4.2.4 implies that $\sup_{x} \hat{Y}_{x}(0) < \infty$.

From Ito's formula, it follows that

$$de^{-\int_0^t q(X(u)) \, du} \hat{Y}_x(t)$$

= $-q(X(t))e^{-\int_0^t q(X(u)) \, du} \hat{Y}_x(t) dt - e^{-\int_0^t q(X(u)) \, du} \tilde{H}(t, X(t), \hat{Y}_x(t), \hat{Z}_x(t)) dt$
 $+\phi(X(t)) dL_t + \langle e^{-\int_0^t q(X(u)) \, du} \hat{Z}_x(t), dM_x(t) \rangle.$

Setting $Y_x(t) := e^{-\int_0^t q(X(u)) \, du} \hat{Y}_x(t)$ and $Z_x(t) := e^{-\int_0^t q(X(u)) \, du} \hat{Z}_x(t)$, we obtain

$$dY_x(t) = -(q(X(t))Y_x(t) + H(X(t), Y_x(t), Z_x(t)))dt + \phi(X(t))dL_t + \langle Z_x(t), dM_x(t) \rangle$$

Moreover,

$$e^{\int_0^t h(u) \, du} Y_x(t) = e^{\int_0^t h(u) \, du} e^{-\int_0^t q(X(u)) \, du} \hat{Y}_x(t) = e^{\int_0^t \tilde{h}(X(u)) \, du} \hat{Y}_x(t) \to 0 \quad as \quad t \to \infty.$$

So by Ito's formula, we have that, for any t < T,

$$e^{\int_{0}^{t}h(u)\,du}Y_{x}(t)$$

$$= e^{\int_{0}^{T}h(u)\,du}Y_{x}(T) + \int_{t}^{T}e^{\int_{0}^{s}h(u)\,du}\left(H(X(s),Y_{x}(s),Z_{x}(t)) + q(X(s))Y_{x}(s)\right)\,ds$$

$$-\int_{t}^{T}e^{\int_{0}^{s}h(u)\,du}\phi(X(s))\,dL_{s} - \int_{t}^{T}h(s)e^{\int_{0}^{s}h(u)\,du}Y_{x}(s)\,ds$$

$$-\int_{t}^{T}e^{\int_{0}^{s}h(u)\,du} < Z_{x}(t), dM_{x}(t) > .$$
(4.54)

Put $u_0(x) = Y_x(0)$ and $v_0(x) = Z_x(0)$.

Since $Y_x(0) = \hat{Y}_x(0)$, we know that u_0 is a bounded function on domain D. By the Markov property of X and the uniqueness of (Y_x, Z_x) , it is easy to see that

$$Y_x(t) = u_0(X(t)), \quad Z_x(t) = v_0(X(t)).$$

So that $\sup_{x \in D, t > 0} |Y_x(t)| \le ||u_0||_{\infty} < \infty.$

Now consider the following problem:

$$\begin{cases} L_2 u(x) = -H(x, u_0(x), v_0(x)), & \text{on } D\\ \frac{1}{2} \frac{\partial u}{\partial \vec{\gamma}}(x) = \phi(x) & \text{on } \partial D. \end{cases}$$

$$(4.55)$$

By Theorem 4.3.5, equation (4.55) has a unique continuous weak solution u. Next we will show that $u = u_0$.

Since u belongs to the domain of the Dirichlet form associated with the process X(t), it follows from the Fukushima's decomposition:

$$du(X(t)) = -[H(X(t), u_0(X(t)), v_0(X(t))) + q(X(t))u(X(t))]dt + \phi(X(t))dL(t) + < \nabla u(X(t)), dM_x(t) > = -[H(X(t), Y_x(t), Z_x(t)) + q(X(t))u(X(t))] + \phi(X(t))dL(t) + < \nabla u(X(t)), dM_x(t) > .$$

From the condition (4.50) and the boundedness of u(x), it follows that

$$\lim_{t \to \infty} E_x[e^{2\int_0^t h(u) \, du} u^2(X(t))] \le \|u\|_\infty^2 \lim_{t \to \infty} E_x[e^{2\int_0^t (\tilde{h}+q)(u) \, du}] = 0.$$

By Ito's formula, it follows that, for any t < T,

$$e^{\int_{0}^{t}h(u)\,du}u(X(t))$$

$$= e^{\int_{0}^{T}h(u)\,du}u(X(T)) + \int_{t}^{T}e^{\int_{0}^{s}h(u)\,du}[H(X(s),Y_{x}(s),Z_{x}(s)) + q(X(s))u(X(s))]\,ds$$

$$-\int_{t}^{T}e^{\int_{0}^{s}h(u)\,du}\phi(X(s))dL(s) - \int_{t}^{T}h(s)e^{\int_{0}^{s}h(u)\,du}u(X(s))\,ds$$

$$-\int_{t}^{T}e^{\int_{0}^{s}h(u)\,du} < \nabla u(X(t)), dM_{x}(t) > .$$
(4.56)

 Set

$$v_x(t) = u(X(t)) - Y_x(t)$$
 and $R_x(t) = \nabla u(X(t)) - Z_x(t)$.

Comparing the equations (4.54) and (4.56), we obtain the following equation: for any t < T,

$$e^{\int_{0}^{t} h(u) \, du} v(X(t))$$

$$= e^{\int_{0}^{T} h(u) \, du} v(X(T)) + \int_{t}^{\infty} (q(X(u)) - h(u)) e^{\int_{0}^{s} h(u) \, du} v(X(s)) \, ds$$

$$- \int_{t}^{\infty} e^{\int_{0}^{s} h(u) \, du} < R_{x}(t), \, dM_{x}(t) >$$

$$= e^{\int_{0}^{T} h(u) \, du} v(X(T)) - \int_{t}^{T} \tilde{h}(s) e^{\int_{0}^{s} h(u) \, du} v(X(s)) \, ds$$

$$- \int_{t}^{T} e^{\int_{0}^{s} h(u) \, du} < R_{x}(t), \, dM_{x}(t) > .$$
(4.57)

Set $g(t) = e^{\int_0^t h(u) \, du} v(t)$. Taking conditional expectations on both sides of (4.57), we find that

$$g(t) = E_x[g(T) - \int_t^T \tilde{h}(s)g(s) \, ds |\mathcal{F}_t]$$

= $E_x[g(T)(1 - \int_t^T \tilde{h}(s) \, ds) + \int_t^T \int_s^T \tilde{h}(s)\tilde{h}(s_1)g(s_1) \, ds_1 ds |\mathcal{F}_t]$
= $E_x[g(T)(1 - \int_t^T \tilde{h}(s) \, ds + \frac{1}{2}(\int_t^T \tilde{h}(s) \, ds)^2)$
+ $(-1)^3 \int_t^T \int_s^T \int_{s_1}^T \tilde{h}(s)\tilde{h}(s_1)\tilde{h}(s_2)g(s_2) \, ds_2 ds_1 ds |\mathcal{F}_t].$

Keeping iterating in the way above for n times, we obtain

$$g(t) = E_x[g(T)(\sum_{k=0}^n \frac{(-\int_t^T \tilde{h}(s) \, ds)^n}{n!}) + (-1)^{n+1} \int_t^T \int_s^T \int_{s_1}^T \dots \int_{s_{n-1}}^T \tilde{h}(s)\tilde{h}(s_1)\dots\tilde{h}(s_n)g(s_n) \, ds_n\dots ds_1 ds |\mathcal{F}_t].$$

It follows that

$$|g(t)| \le E_x[|g(T)|e^{-\int_t^T \tilde{h}(s)\,ds}|\mathcal{F}_t].$$

Then by $g(t) = e^{\int_0^t h(u) \, du} v(t)$, we obtain,

$$|v(t)| \le E_x[|v(T)|e^{\int_t^T (h(s) - \tilde{h}(s)) \, ds} |\mathcal{F}_t] \le (||u_0||_\infty + ||u||_\infty) E_x[e^{\int_t^T q(X(s)) \, ds} |\mathcal{F}_t].$$
(4.58)

Hence, it follows that

$$0 \le E_x[e^{\int_0^t q(X(s))\,ds}|v(t)|] \le (||u_0||_\infty + ||u||_\infty)E_x[e^{\int_0^T q(X(s))\,ds}],$$

for any T > t.

Since the condition (4.51) and Lemma 4.3.4 imply

$$\lim_{t \to \infty} E_x[e^{\int_0^t q(X(s)) \, ds}] = 0,$$

we know that $E_x[e^{\int_0^t q(X(s)) ds} |v(t)|] = 0$. This implies that $v(t) = 0, P_x - a.s.$.

Therefore, for any t > 0, we have $u(X(t)) = Y_x(t)$. In particular, $u(x) = E_x[u(X(0))] = E_x[Y_x(0)] = u_0(x)$. This shows that u(x) is a weak solution of the equation (4.46).

If \tilde{u} is another solution of the problem (4.46). Then the processes $\tilde{Y}_x(t) := \tilde{u}(X(t))$ and $\tilde{Z}_x(t) := \nabla \tilde{u}(X(t))$ satisfy the following equation

$$d\tilde{Y}_x(t) = -H(X(t), \tilde{Y}_x(t), \tilde{Z}_x(t))dt - \phi(X(t))dL_t + \langle \tilde{Z}_x(t), dM_x(t) \rangle .$$
(4.59)

Set
$$\bar{Y}_x(t) = e^{\int_0^t q(X(u)) \, du} \tilde{Y}_x(t)$$
 and $\bar{Z}_x(t) = e^{\int_0^t q(X(u)) \, du} \tilde{Z}_x(t)$.

By the chain rule, it follows that

$$d\bar{Y}_x(t) = -\tilde{H}(X(t), \bar{Y}_x(t), \bar{Z}_x(t))dt + e^{\int_0^t q(X(u)) \, du} \phi(X(t))dL(t) + \langle \bar{Z}_x(t), dM_x(t) \rangle .$$

Moreover, because \tilde{u} is bounded,

$$\lim_{t \to \infty} e^{\int_0^t \tilde{h}(u) \, du} \bar{Y}_x(t) = \lim_{t \to \infty} e^{\int_0^t h(u) \, du} \tilde{u}(X(t)) = 0$$

is also satisfied. Therefore, from the uniqueness of the solution of the BSDE in Theorem 4.2.3, we have

$$\tilde{Y}_x(t) = Y_x(t)$$
 and $\tilde{Z}_x(t) = Z_x(t)$.

In particular,

$$\tilde{u}(x) = E_x[\tilde{Y}_x(t)] = E_x[Y_x(t)] = u(x).$$

4.5 Semilinear Elliptic PDEs with Singular Coefficients

Recall the operator

$$L = \frac{1}{2}\nabla \cdot (A\nabla) + B \cdot \nabla - div(\hat{B} \cdot) + Q$$

on the domain D equipped with the mixed boundary condition on ∂D :

$$\frac{1}{2}\frac{\partial u}{\partial \vec{\gamma}}(x) - \langle \hat{B}, n \rangle u(x) = 0.$$

 $\mathcal{Q}(u, v)$ is the quadratic form associated with L, with the domain $\mathcal{D}(\mathcal{Q}) = W^{1,2}(D)$. $\{T_t, t \ge 0\}$ denotes the semigroup generated by L.

In this section, our main aim is to solve the following equation:

$$\begin{cases} Lf(x) = -F(x, f(x)), & \text{on } D, \\ \frac{1}{2}\frac{\partial f}{\partial \vec{\gamma}}(x) - \langle \hat{B}, n \rangle (x)f(x) = \Phi(x), & \text{on } \partial D. \end{cases}$$

$$(4.60)$$

Definition 4.5.1. A bounded continuous function u defined on D is called a weak

solution of the equation (4.60) if $u \in W^{1,2}$, and for any $g \in C^{\infty}(\overline{D})$,

$$\mathcal{Q}(u,g) = \int_{\partial D} \Phi(x)g(x)\lambda(dx) + \int_{D} F(x,u(x))g(x)dx.$$

Here the function $F: \mathbb{R}^d \times \mathbb{R} \to \mathbb{R}$ is a bounded measurable function and satisfies the following condition:

(E.1)
$$(y_1 - y_2)(F(x, y_1) - F(x, y_2)) \le -r_1(x)|y_1 - y_2|^2$$
.
Set

$$\hat{Z}_{t} = \exp(\int_{0}^{t} (A^{-1}B)^{*}(X_{s}^{0})dM_{s}^{0} + (\int_{0}^{t} (A^{-1}\hat{B})^{*}(X_{s}^{0})dM_{s}^{0}) \circ \gamma_{t}^{0} - \frac{1}{2}\int_{0}^{t} (B-\hat{B})^{*}A^{-1}(B-\hat{B})(X_{s}^{0})\,ds + \int_{0}^{t} Q(X_{s}^{0})\,ds).$$
(4.61)

By the reduction method introduced in Section 3.2, there exists a bounded, continuous functions $v \in W^{1,p}(D)$ satisfying

$$T_t f(x) = e^{-v(x)} S_t [f e^v](x).$$

Here, S_t is the semigroup generated by the operator: $G = \frac{1}{2}\nabla \cdot (A\nabla) + b \cdot \nabla + q$ equipped with the boundary condition $\langle A\nabla u, n \rangle = 0$, where $b := B - \hat{B} - (A\nabla v)$ and $q := Q + \frac{1}{2}(\nabla v)^* A(\nabla v) - \langle B - \hat{B}, \nabla v \rangle$.

In this section, we will stick to this particular choice of b and q. Recall that

$$\tilde{M}(t) = e^{\int_0^t \langle A^{-1}b(X_s^0), dM_s^0 \rangle - \frac{1}{2}\int_0^t b^* A^{-1}b(X_s^0) \, ds}$$

and set $Z_t = \tilde{M}(t)e^{\int_0^t q(X_s^0) \, ds}$.

Then from (3.9), it follows that

$$\hat{Z}(t) = Z_t e^{v(X_t^0) - v(X_0^0)}.$$

The process (X, P_x) associated with operator G_2 has the following decomposition:

$$X(t) = x + \int_0^t \sigma(X(s)) dB_s + \int_0^t (\frac{1}{2}\nabla A + b)(X(s)) ds + \int_0^t \vec{\gamma}(X(s)) dL_s, \quad P_x - a.s.$$

where $\{B_t\}_{t\geq 0}$ is a d-dimensional Brownian motion and L_t is the local time satisfying $L_t = \int_0^t I_{\partial D}(X(s)) dL_s$. It is known from [29] that the processes (X, P_x) and (X^0, P_x^0) are related as follows,

$$dP_x|_{\mathcal{F}_t} = \tilde{M}_t dP_x^0|_{\mathcal{F}_t}.$$

Since we can not set up conditions on the functions q and b, which are actually the "intermediates", the following lemma gives us an important condition to prove the existence of the solution to equation (4.60).

Lemma 4.5.2. Assume that there exists $x_0 \in D$, such that

$$E_{x_0}^0 \left[\int_0^\infty |\hat{Z}_t|^2 e^{\int_0^t (2Q - 4r_1)(X_u^0) \, du} dL_t^0 \right] < \infty.$$
(4.62)

Then there exists a positive number $\varepsilon > 0$, such that if $\|\hat{B}\|_{L^p} \leq \varepsilon$, the following inequality holds:

$$\sup_{x \in D} E_x \left[\int_0^\infty e^{2\int_0^t (-r_1 + q)(X(u)) \, du} dt \right] < \infty.$$
(4.63)

Proof. First note that,

$$\begin{split} E_x[e^{2\int_0^t (-r_1+q)(X(u))\,du}] &= E_x^0[\tilde{M}(t)e^{2\int_0^t (-r_1+q)(X_u^0)\,du}] \\ &= E_x^0[Z(t)e^{\int_0^t (-2r_1+q)(X(u))\,du}] \\ &\leq C_1E_x^0[\hat{Z}(t)e^{-2\int_0^t (r_1(X(u))\,du}e^{\int_0^t (Q+\frac{1}{2})(X_u^0)\,du}] \\ &\leq C_1E_x^0[\hat{Z}^2(t)e^{2\int_0^t (Q-2r_1)(X_u^0)\,du}]^{\frac{1}{2}} \cdot E_x^0[e^{\int_0^t (X_u^0)\,du}]^{\frac{1}{2}} \end{split}$$

By Lemma 4.3.4 and condition (4.62), there exist two constants $c_2, \beta > 0$ such

that

$$\sup_{x \in D} E_x[\hat{Z}^2(t)e^{2\int_0^t (Q-2r_1)(X_u^0)\,du}] < c_2 e^{-\beta t}.$$

Moreover, for p > d, by the Theorem 2.1 in [27], there exist two positive constants c_3 and c_4 such that

$$E_x^0[e^{\int_0^t (X_u^0) \, du}] \le c_3 e^{c_4 t},$$

where $c_4 = c \| < A \nabla v - 2(B - \hat{B}), \nabla v > \|_{L^{p/2}}.$

Since $\|\nabla v\|_{L^p} \leq C \|\hat{B}\|_{L^p(D)}$ (see [37]), there exists $\varepsilon > 0$, such that $\|\hat{B}\|_{L^p(D)} \leq \varepsilon$ implies $c_4 < \beta$. Thus (4.63) holds.

Theorem 4.5.3. Assume (4.62). Then for some point $x_0 \in D$

$$E_{x_0}^0 \left[\int_0^\infty \hat{Z}_s \, dL_s^0 \right] < \infty \tag{4.64}$$

and moreover, $\|\hat{B}\|_{L^p} \leq \varepsilon$, where ε is as in Lemma 4.5.2. Then the problem (4.60) has a unique, bounded, continuous weak solution u(x).

Proof. Existence:

Set $\tilde{F}(x,y) = e^{v(x)}F(x,e^{-v(x)}y)$ and $\phi(x) = e^{v(x)}\Phi(x)$.

From the boundedness of v , \tilde{F} is also bounded and satisfies

$$(y_1 - y_2)(\tilde{F}(x, y_1) - \tilde{F}(x, y_2)) \le -r_1(x)|y_1 - y_2|^2.$$

Moreover, there is a constant c > 0, such that

$$\infty > E_{x_0}^0 [\int_0^\infty \hat{Z}_s \, dL_s^0] = E_{x_0}^0 [\int_0^\infty Z_s e^{v(X_s^0) - v(X_0^0)} \, dL_s^0]$$

$$\ge c E_{x_0}^0 [\int_0^\infty Z_s \, dL_s^0]$$

$$= c E_{x_0}^0 [\int_0^\infty \tilde{M}_s e^{\int_0^s q(X_u^0) \, du} \, dL_s^0]$$

$$= c E_{x_0} [\int_0^\infty e^{\int_0^s q(X_u) \, du} \, dL_s].$$
(4.65)

The last equality above is due to the Lemma 4.4.2.

Furthermore, by Lemma 4.4.3, it follows that

$$\sup_{x} E_x \left[\int_0^\infty e^{\int_0^t q(X(u)) \, du} dL_t \right] < \infty.$$
(4.66)

By Lemma 4.5.2, the following condition is satisfied :

$$E_x[\int_0^\infty e^{2\int_0^t (q-r_1)(X(u))\,du}dt] < \infty,\tag{4.67}$$

So \tilde{F} satisfies all of the conditions in Theorem 4.4.4 replacing G by \tilde{F} . Thus the following problem

$$\begin{cases} Gu(x) = -\tilde{F}(x, u(x)), & \text{on } D\\ \frac{1}{2} \frac{\partial u}{\partial \vec{\gamma}}(x) = \phi & \text{on } \partial D \end{cases}$$
(4.68)

has a unique bounded continuous weak solution u.

Set $f(x) = e^{-v(x)}u(x)$. Then we claim the function f(x) is the weak solution of the equation (4.60).

Because function v is continuous and bounded, f(x) is also continuous. From the fact that function u is the weak solution of the problem (4.68), we obtain, for any function $\psi \in C^{\infty}(D)$,

$$\mathcal{E}(u, e^{-v}\psi) = \frac{1}{2} \int_{D} \left(\langle A\nabla u, \nabla(e^{-v}\psi) \rangle - \langle b, \nabla u \rangle e^{-v}\psi - e^{-v}qu\psi \right) dx$$

$$= \int_{\partial D} e^{-v}\phi\psi \,d\lambda + \int_{D} \tilde{F}(x, u(x))e^{-v}\psi \,dx.$$
(4.69)

As in the proof of Theorem 5.1 in [42], we can show that the left side of the equation (4.69) is equal to

$$\mathcal{Q}(f,\psi) = \frac{1}{2} \int_{D} \left[\langle A\nabla f, \nabla \psi \rangle - \langle B, \nabla u \rangle \psi - \langle \hat{B}, \nabla \psi \rangle f - Qf\psi \right] dx.$$

At the same time, by the definition of the function ϕ and \tilde{F} , the right side of the equation (4.69) is equal to

$$\int_{\partial D} \Phi \psi \, d\lambda + \int_D F(x, f(x)) \psi \, dx.$$

Thus it follows that, for any $\psi \in C^{\infty}(D)$,

$$\mathcal{Q}(f,\psi) = \int_{\partial D} \Phi \psi d\lambda + \int_D F(x,f(x))\psi \, dx,$$

which proves that function f is a weak solution of the problem (4.60).

Uniqueness:

If \bar{f} is another solution of the problem (4.60), then $\bar{u} := e^v f$ can be shown to be the solution of the equation (4.68). Then by the uniqueness of the problem (4.68) proved in the Theorem 4.4.4, we find $\bar{u} = u$. Therefore, $f = \bar{f}$.

4.6 L^1 Solutions to the BSDEs and Semilinear PDEs

In this section, we consider the L^1 solutions of the BSDEs and use this result to solve the nonlinear elliptic partial differential equation with the mixed boundary condition.

Let $f : \Omega \times R^+ \times R \to R$ be progressively measurable. Consider the following conditions:

(I.1) $(y - y')(f(t, y) - f(t, y')) \le d(t)|y - y'|^2 - a.s.$, where d(t) is a progressively measurable process;

- $(\mathbf{I.2})E[\int_0^\infty e^{\int_0^s d(u)\,du} |f(s,0)|\,ds] < \infty;$
- (I.3) $P_x a.s.$, for any $t > 0, y \to f(t, y)$ is continuous;

(I.4)
$$\forall r > 0$$
, $T > 0$, $\psi_r(t) := \sup_{|y| \le r} |f(t,y) - f(t,0)| \in L^1([0,T] \times \Omega, dt \times P_x)$
The following lamma is deduced from Corollary 2.2 in [4]

The following lemma is deduced from Corollary 2.3 in [4].

Lemma 4.6.1. Suppose a pair of progressively measurable processes (Y, Z) with values in $R \times R^d$ such that $t \to Z_t$ belongs to $L^2([0,T])$ and $t \to f(t,Y_t)$ belongs to $L^1([0,T])$, $P_x - a.s.$. If

$$Y_t = \xi + \int_t^T f(r, Y_r) \, dr - \int_t^T \langle Z_r, dM_r \rangle, \qquad (4.70)$$

then the following inequality holds, for $0 \le t < u \le T$,

$$|Y_t| \le |Y_u| + \int_t^u \hat{Y}_s f(s, Y_s) \, ds - \int_t^u \hat{Y}_s < Z_r, \, dM_r > .$$

where $\hat{y} = \frac{y}{|y|} I_{\{y \neq 0\}}$.

The following lemma can be proved by modifying the proof of Proposition 6.4 in [4].

Lemma 4.6.2. Assume the conditions (I.1)-(I.4) with $d(t) \equiv 0$. Then there exists a unique solution (Y, Z) of the BSDE

$$Y_t = \int_t^T f(r, Y_r) \, dr - \int_t^T \langle Z_r, dM_r \rangle, \quad for \quad 0 \le t \le T.$$
(4.71)

Moreover, for each $\beta \in (0,1)$, $E[\sup_{t \le T} |Y_t|^{\beta}] + E[(\int_0^T |Z_r|^2 dr)^{\frac{\beta}{2}}] < \infty$.

Suppose $\beta \in (0, 1)$.

 \mathcal{S}^{β} denotes the set of real-valued, adapted and continuous processes $\{Y_t\}_{t\geq 0}$ such that

$$||Y||^{\beta} := E[\sup_{t>0} |Y_t|^{\beta}] < \infty.$$

It is known that $\|\cdot\|^{\beta}$ induces a complete metric on the space of real-valued continuous processes ([4]).

 M^{β} denotes the set of R^{d} -valued predictable processes $\{Z_{t}\}$ such that

$$||Z||_{M^{\beta}} := E[(\int_0^\infty |Z_t|^2 dt)^{\frac{\beta}{2}}] < \infty.$$

 M^{β} is also a complete metric space with the distance deduced by $\|\cdot\|_{M^{\beta}}$.

Lemma 4.6.3. Assume that conditions (I.1)-(I.4) with $d(t) \equiv 0$, then there exists a unique solution (Y, Z) of the BSDE

$$Y_{t} = Y_{T} + \int_{t}^{T} f(r, Y_{r}) dr - \int_{t}^{T} \langle Z_{r}, dM_{r} \rangle, \quad for \quad 0 \le t \le T;$$
$$\lim_{t \to \infty} Y_{t} = 0, \quad P - a.s.. \tag{4.72}$$

Proof. Existence:

By Lemma 4.6.2, there exists (Y^n, Z^n) such that, for $0 \le t \le n$,

$$Y_t^n = \int_t^n f(r, Y_r^n) \, dr - \int_t^n \langle Z_r^n, dM_r \rangle,$$

and $Y_t^n = Z_t^n = 0$, for $t \ge n$.

Fix t > 0 and t < n < n + i, then

$$\begin{aligned} Y_t^{n+i} - Y_t^n &= \int_t^{n+i} (f(r, Y_r^{n+i}) - f(r, Y_r^n)) \, dr - \int_t^{n+i} < (Z_r^{n+i} - Z_r^n), \, dM_r > \\ &+ \int_n^{n+i} f(r, 0) \, dr. \end{aligned}$$

Set

$$F^{n}(r, y) = f(r, y + Y_{r}^{n}) - f(r, Y_{r}^{n}) + f(r, 0)I_{\{r>n\}},$$
$$y_{t}^{n} = Y_{t}^{n+i} - Y_{t}^{n} \quad \text{and} \quad z_{t}^{n} = Z_{t}^{n+i} - Z_{t}^{n}.$$

Then (y_t^n, z_t^n) is the solution of the following BSDE:

$$y_t^n = \int_t^{n+i} F^n(r, y_r^n) \, dr - \int_t^{n+i} \langle z_r^n, dM_r \rangle \,. \tag{4.73}$$

So that by the condition (I.1) with $d(t) \equiv 0$ and Lemma 4.6.1, it follows that

$$\begin{aligned} |y_t^n| &\leq \int_t^{n+i} <\hat{y}_r^n, F^n(r, y_r^n) > dr - \int_t^{n+i} \hat{y}_r^n < z_r^n, dM_r > \\ &\leq \int_t^{n+i} \frac{I_{\{y_r^n \neq 0\}}}{|y_r^n|} < y_r^n, f(r, y_r^n + Y_r^n) - f(r, Y_r^n) > dr + \int_n^{n+i} |f(s, 0)| \, ds \\ &- \int_t^{n+i} \hat{y}_r^n < z_r^n, dM_r > \\ &\leq \int_n^{n+i} |f(s, 0)| \, ds - \int_t^{n+i} \hat{y}_r^n < z_r^n, dM_r > . \end{aligned}$$

$$(4.74)$$

Taking conditional expectations on both sides of the inequality, we get

$$|y_t^n| \le E[\int_n^{n+i} |f(s,0)| \, ds \quad | \mathcal{F}_t] := M_t^n,$$

where M_t^n is a martingale. Then by the Doob's inequality and condition (I.2), it follows that, for $\beta \in (0, 1)$,

$$E[\sup_{t} |y_{t}^{n}|^{\beta}] \leq E[\sup_{t} (M_{t}^{n})^{\beta}] \leq \frac{1}{1-\beta} (E[\int_{n}^{n+i} |f(s,0)| \, ds])^{\beta}$$

$$\rightarrow 0, \quad as \quad n \to \infty.$$
(4.75)

Therefore we know that $\{Y^n\}$ is a Cauchy sequence under the norm $\|\cdot\|_{\infty}^{\beta}$. Hence there is a process Y such that $E[\sup_t |Y_t - Y_t^n|^{\beta}] \to 0$.

This also implies that $Y_t \to 0$, as $t \to \infty$, $P_x - a.s.$.

Moreover, by the equation (4.73), Ito's formula and the condition (I.1), it follows that

$$\begin{split} |y_t^n|^2 + \int_t^{n+i} &< A(X(r))z_r^n, z_r^n > dr \\ &= 2\int_t^{n+i} y_r^n F^n(r, y_r^n) \, dr - 2\int_t^{n+i} y_r^n < z_r^n, dM_r > \\ &\leq 2\int_n^{n+i} y_r^n f(r, 0) \, dr + 2|\int_t^{n+i} y_r^n < z_r^n, dM_r > | \\ &\leq \sup_r |y_r^n|^2 + (\int_n^{n+i} |f(r, 0)| \, dr)^2 + 2|\int_t^{\infty} y_r^n < z_r^n, dM_r > |, \end{split}$$

and thus that

$$\left(\int_{t}^{n+i} |z_{r}^{n}|^{2} dr\right)^{\frac{\beta}{2}} \leq c_{1}\left[\sup_{r} |y_{r}^{n}|^{\beta} + \left(\int_{n}^{n+i} |f(r,0)| dr\right)^{\beta} + \left|\int_{t}^{n+i} y_{r}^{n} < z_{r}^{n}, dM_{r} > \right|^{\frac{\beta}{2}}\right].$$

Taking expectations on both sides of the inequality and applying the Burkholder-Davis-Gundy inequality, we obtain

$$E[\left(\int_{t}^{n+i}|z_{r}^{n}|^{2} dr\right)^{\frac{\beta}{2}}]$$

$$\leq c_{1}E[\sup_{r}|y_{r}^{n}|^{\beta}] + c_{1}(E[\int_{n}^{n+i}|f(r,0)| dr])^{\beta} + c_{2}E[\left(\int_{t}^{n+i}|y_{r}^{n}|^{2}|z_{r}^{n}|^{2} dr\right)^{\frac{\beta}{4}}]$$

$$\leq c_{1}E[\sup_{r}|y_{r}^{n}|^{\beta}] + c_{1}(E[\int_{n}^{n+i}|f(r,0)| dr])^{\beta} + c_{2}E[(\sup_{r}|y_{r}^{n}|^{\frac{\beta}{2}} \int_{n}^{n+i}|z_{r}^{n}|^{2} dr)^{\frac{\beta}{4}}]$$

$$\leq (c_{1} + \frac{c_{2}^{2}}{2})E[\sup_{r}|y_{r}^{n}|^{\beta}] + c_{1}(E[\int_{n}^{n+i}|f(r,0)| dr])^{\beta} + \frac{1}{2}E[\left(\int_{t}^{n+i}|z_{r}^{n}|^{2} dr\right)^{\frac{\beta}{2}}].$$

Therefore, taking t = 0, we know that there is a constant C > 0, such that

$$E\left[\left(\int_{0}^{\infty}|z_{s}^{n}|^{2} ds\right)^{\frac{\beta}{2}}\right]$$

$$\leq CE\left[\sup_{t}|y_{t}^{n}|^{\beta}\right] + C\left(E\left[\int_{n}^{n+i}|f(r,0)| dr\right]\right)^{\beta}$$

$$\rightarrow 0 \quad as \quad n \rightarrow \infty.$$

Hence $\{Z_t^n\}$ is a Cauchy sequence in M^{β} . Let Z denote the limit of $\{Z^n\}$. Finally, by the condition **(I.3)**, we find that

$$\int_0^T f(t, Y_t^n) dt \to \int_0^T f(t, Y_t) dt, \quad P_x - a.s..$$
(4.76)

Therefore, (Y,Z) is a solution of the BSDE (4.72).

Uniqueness:

Suppose (Y, Z) and (Y', Z') are two solutions to (4.72). Then by the estimate in

Lemma 4.6.1 and the fact $d(t) \equiv 0$, it follows that, for t < T,

$$\begin{aligned} |Y_t - Y'_t| &\leq |Y_T - Y'_T| + \int_t^T \frac{I_{Y_r \neq Y'_r}}{|Y_r - Y'_r|} (Y_r - Y'_r) (f(r, Y_r) - f(r, Y'_r)) \, dr \\ &- \int_t^T \langle Z_r - Z'_r, dM_r \rangle, \end{aligned}$$

which implies that

$$E|Y_t - Y'_t| \le E|Y_T - Y'_T| \to 0, \quad as \quad T \to \infty.$$

Therefore, it holds that

$$\forall t > 0, \quad |Y_t - Y'_t| = 0, \quad P - a.s..$$

(I.5) The process d(t) is a progressively measurable process satisfying the following condition:

$$d(\cdot) \in L^1[[0,T] \times \Omega, dt \otimes P], \quad for \quad any \quad T > 0.$$

Theorem 4.6.4. Assume the conditions (I.1)-(I.5) hold. Then there exists a unique process (Y, Z) such that,

$$Y_{t} = Y_{T} + \int_{t}^{T} f(r, Y_{r}) dr - \int_{t}^{T} \langle Z_{r}, dM_{r} \rangle, \quad for \quad 0 \le t \le T;$$
$$\lim_{t \to \infty} e^{\int_{0}^{t} d(u) du} Y_{t} = 0, \quad P - a.s.$$
(4.77)

Proof. Existence:

Set
$$\hat{f}(t,y) = e^{\int_0^t d(u) \, du} f(t, e^{-\int_0^t d(u) \, du} y) - d(t)y$$
. Then
(1) $(y - y')(\hat{f}(t,y) - \hat{f}(t,y')) \le 0$;
(2) $\hat{f}(t,0) = e^{\int_0^t d(u) \, du} f(t,0)$. So

$$E[\int_0^\infty |\hat{f}(s,0)| \, ds] = E[\int_0^\infty e^{\int_0^s d(u) \, du} |f(s,0)| \, ds] < \infty.$$

(3) $\sup_{|y| \le r} |\hat{f}(t,y) - \hat{f}(t,0)| \le \psi_r(t) + |d(t)|r$, where the process satisfies

$$\psi_r(t) + |d(t)| r \in L^1([0,T] \times \Omega, dt \otimes P),$$

for T > 0.

Therefore, \hat{f} satisfies all of the conditions of the Lemma 4.6.3. So there exists a pair of process (\hat{Y}, \hat{Z}) satisfying the equation:

$$\hat{Y}_t = \hat{Y}_T + \int_t^T \hat{f}(r, \hat{Y}_r) \, dr - \int_t^T \langle \hat{Z}_r, dM_r \rangle,$$

and obviously $\lim_{t\to\infty} \hat{Y}_t = 0.$

By the chain rule and the definition of the function \hat{f} , it follows that

$$de^{-\int_0^t d(u) \, du} \hat{Y}_t = -f(t, e^{-\int_0^t d(u) \, du} \hat{Y}_t) dt + \langle e^{-\int_0^t d(u) \, du} \hat{Z}_t, dM_t \rangle$$

Set $Y_t = e^{-\int_0^t d(u) \, du} \hat{Y}_t$ and $Z_t = e^{-\int_0^t d(u) \, du} \hat{Z}_t$. Then the process (Y, Z) is the solution to the equation (4.77).

Uniqueness:

The uniqueness of the solution to (4.77) follows from the uniqueness of the solution to equation (4.72).

Let $H : \mathbb{R}^d \times \mathbb{R} \to \mathbb{R}$ be a bounded Borel measurable function. Consider the following conditions:

 $(\mathbf{H.1})' (y_1 - y_2)(H(x, y_1) - H(x, y_2)) \le -h_1(x)|y_1 - y_2|^2, \text{ where } h_1 \in L^p(D) \text{ for } p > \frac{d}{2}.$ (**H.2**)' $y \to H(x, y)$ is continuous.

Theorem 4.6.5. Assume the Conditions (H.1)' and (H.2)' hold and that there is some point $x_0 \in D$, such that

$$E_{x_0}\left[\int_0^\infty e^{\int_0^s (-h_1+q)(X(u))\,du}\,dL_s\right] < \infty,\tag{4.78}$$

and for some point $x_1 \in D$,

$$E_{x_1}[\int_0^\infty e^{\int_0^s q(X(u))\,du}\,dL_s] < \infty.$$
(4.79)

Then the semilinear Neumann boundary value problem

$$\begin{cases} Gu(x) = -H(x, u(x)), & on D\\ \frac{\partial u}{\partial \vec{\gamma}}(x) = \phi(x) & on \partial D \end{cases}$$
(4.80)

has a unique continuous weak solution.

Proof. Step 1

Set $\tilde{H}(t, x, y) = e^{\int_0^t q(X(u))dt} H(x, e^{-\int_0^t q(X(u))dt}y)$, then there exists a unique solution (\hat{Y}_x, \hat{Z}_x) to the following BSDE: for any T > 0 and 0 < t < T,

$$\hat{Y}_{x}(t) = \hat{Y}_{x}(T) + \int_{t}^{T} \tilde{H}(s, X(s), \hat{Y}_{x}(s)) \, ds - \int_{t}^{T} e^{\int_{0}^{s} q(X(u)) dt} \phi(X(s)) \, dL_{s} \\
- \int_{t}^{T} \langle \hat{Z}_{x}(s), dM_{x}(s) \rangle$$

and

$$\lim_{t \to \infty} e^{-\int_0^t h_1(X(u)) \, du} \hat{Y}_t = 0 \quad P_x - a.s.$$

The uniqueness follows from the uniqueness proved in the last Theorem. Only the existence of solution (\hat{Y}_x, \hat{Z}_x) needs to be proved:

(a) Similarly as the proof of Theorem 4.2.3, we can show that there exists $(p_x(t), q_x(t))$ such that

$$dp_x(t) = e^{\int_0^t q(X(u)) \, du} \phi(X(t)) dL_t + \langle q_x(t), dM_x(t) \rangle,$$

$$e^{-\int_0^t h_1(X(u)) \, du} p_x(t) \to 0, \quad as \quad t \to \infty, \quad P_x - a.s..$$
(4.81)

(b) Set $g(t, x, y) = \tilde{H}(t, x, y + p_x(t))$. Then it follows that

$$(y - y')(g(t, X(t), y) - g(t, X(t), y')) \le -h_1(X(t))|y - y'|^2.$$

The condition (4.78) and Lemma 4.3.4 imply, for $x \in D$,

$$E_x\left[\int_0^\infty e^{\int_0^s (-h_1+q)(X(u))\,du}\,ds\right] < \infty.$$

Furthermore, as the function H is bounded, we know that condition (I.2) is satisfied:

$$E_{x}\left[\int_{0}^{\infty} e^{-\int_{0}^{s} h_{1}(X(u)) \, du} |g(s, X(s), 0)| \, ds\right]$$

$$= E_{x}\left[\int_{0}^{\infty} e^{-\int_{0}^{s} h_{1}(X(u)) \, du} |\tilde{H}(s, X(s), p_{x}(s))| \, ds\right]$$

$$= E_{x}\left[\int_{0}^{\infty} e^{\int_{0}^{s} (-h_{1}+q)(X(u)) \, du} |H(X(s), e^{-\int_{0}^{s} q(X(u)) \, du} p_{x}(s))| \, ds\right]$$

$$\leq ||H||_{\infty} E_{x}\left[\int_{0}^{\infty} e^{\int_{0}^{t} (-h_{1}+q)(X(u)) \, du} dt\right]$$

$$< \infty.$$
(4.82)

Since $y \to g(x, y)$ is continuous, the condition (I.3) is satisfied. Moreover, the condition (I.4) is also satisfied. In fact, for any r > 0,

$$\psi_r(t) = \sup_r |\tilde{H}(t, X(t), y) - \tilde{H}(t, X(t), 0)| \le 2 ||H||_{\infty} e^{\int_0^t q(X_t) dt},$$

and for any T > 0, by the fact that $q \in L^p(D)$ with p > d and Theorem 2.1 in [27], $E_x[\int_0^T e^{\int_0^t q(X_u) du} dt] < \infty.$

Therefore, the function g satisfies all of the conditions of Theorem 4.6.4. There exists a pair of process $(y_x(t), z_x(t))$ such that for any T > 0 and 0 < t < T,

$$y_x(t) = y_x(T) + \int_t^T g(X(s), y_x(s)) \, ds - \int_t^T \langle z_x(s), dM_x(s) \rangle$$
(4.83)

and

$$\lim_{t \to \infty} e^{-\int_0^t h_1(X(u)) \, du} y_x(t) = 0 \quad P_x - a.s..$$
(4.84)

Put $\hat{Y}_x(t) = p_x(t) + y_x(t)$ and $\hat{Z}_x(t) = q_x(t) + z_x(t)$. It follows that $(\hat{Y}_x(t), \hat{Z}_x(t))$ satisfies the following equation

$$d\hat{Y}_{x}(t) = e^{\int_{0}^{t} q(X(u)) \, du} \phi(X(t)) dL_{t} - \tilde{H}(t, X(t), \hat{Y}_{x}(t)) dt + \langle \hat{Z}_{x}(t), dM_{x} \rangle,$$
$$\lim_{t \to \infty} e^{-\int_{0}^{t} h_{1}(X(u)) \, du} \hat{Y}_{t} = 0 \quad P_{x} - a.s..$$

Step 2.

Put
$$Y_x(t) := e^{-\int_0^t q(X(u)) \, du} \hat{Y}_x(t)$$
 and $Z_x(t) := e^{-\int_0^t q(X(u)) \, du} \hat{Z}_x(t)$, we have

$$dY_x(t) = -F(X(t), Y_x(t)) + \phi(X(t))dL_t + \langle Z_x(t), dM_x(t) \rangle,$$

where F(x, y) = q(x)y + H(x, y).

Moreover,

$$e^{\int_0^t (-h_1+q)(X(u)) \, du} Y_x(t) = e^{\int_0^t (-h_1+q)(X(u)) \, du} e^{-\int_0^t q(X(u)) \, dt} \hat{Y}_x(t)$$
$$= e^{-\int_0^t h_1(X(u)) \, du} \hat{Y}_x(t) \to 0 \quad as \quad t \to \infty$$

Put $u_0(x) = Y_x(0)$ and $v_0(x) = Z_x(0)$.

Now as in Theorem 4.4.4, we can solve the following equation

$$\begin{cases} Gu(x) = -H(x, u_0(x)), & \text{on } D, \\ \frac{1}{2} \frac{\partial u}{\partial \vec{\gamma}}(x) = \phi(x) & \text{on } \partial D, \end{cases}$$

$$(4.85)$$

and prove that the solution u coincides with $u_0(x)$. This completes the proof of the whole theorem.

Recall the operators L and G introduced in Section 4.5. Suppose that F : $R^d \times$

 $R \to R$ is a bounded measurable function and $r_1 \in L^p(D)$. Consider the following conditions :

(E.1) $(y_1 - y_2)(F(x, y_1) - F(x, y_2)) \le -r_1(x)|y_1 - y_2|^2;$

(E.2) $y \to F(x, y)$ is continuous.

Now following the same proof as that of Theorem 4.5.3, we obtain

Theorem 4.6.6. Suppose that the function F satisfies the condition (E.1) and (E.2), and there exist $x_0, x_1 \in D$ such that

$$E_{x_0}^0 \left[\int_0^\infty e^{-\int_0^t h_1(X_u^0) \, du} \hat{Z}_t dL_t^0 \right] < \infty, \tag{4.86}$$

and

$$E_{x_1}^0 [\int_0^\infty \hat{Z}_t dL_t^0] < \infty.$$
(4.87)

Then the following problem

$$\begin{cases} Lu(x) = -F(x, u(x)), & \text{on } D\\ \frac{1}{2} \frac{\partial u}{\partial \vec{\gamma}}(x) - \langle \hat{B}, n \rangle (x) u(x) = \Phi(x), & \text{on } \partial D \end{cases}$$

$$(4.88)$$

has a unique, bounded, continuous weak solution.

Chapter 5

Future Studies

5.1 An Inspiring Example

In this chapter, we consider the operator

$$L = \frac{1}{2}\nabla(A\nabla) + \langle b, \nabla \rangle,$$

which is associated with the process $\{\Omega, \mathcal{F}_t, X_t, P_x\}$, semigroup $\{P_t\}$ and the transition density function p(t, x, y).

It is known that the adjoint operator L^* of L, satisfying, for $u \in D(L)$, $v \in D(L^*)$:

$$\int L(u)vdm = \int u(L^*v)dm,$$

and has the following expression:

$$L^* = \frac{1}{2}\nabla(A\nabla\cdot) - div(b\cdot).$$
(5.1)

In this sense, the term in the form $div(b \cdot)$ leads us to study the adjoint operator and adjoint process. In the following discussion, we fix time t > 0 and then define the backward filtration $\{\overleftarrow{\mathcal{F}}_s^t, s < t\}$ as $\overleftarrow{\mathcal{F}}_s^t = \overline{\sigma(X_r, r \in [s, t])}$. In [9] and [23], the reverse process $\{Y_s = X_s \circ r_t : s \in [0, t]\}$ is a Markov process with respect to the filtration $\{\overleftarrow{\mathcal{F}}_s^t : s \in [0,t]\}$. L^* is associated with the semigroup P^* and the transition density function $p^*(t,x,y) = p(t,y,x)$.

Notice that the adjoint process Y is not homogenous under the measure P_x . Based on this notice, we apply the "time-reversal" method introduced before to deal with the divergence term. But we also find that, if we keep calculating the semigroup of the adjoint process Y under the probability P_x , something interesting will happen.

We try to study the simplest Dirichlet problem in this section:

$$\begin{cases} L^* u = \frac{1}{2} \nabla (A \nabla u)(x) - div(b(x)u(x)) = -F(x, u(x), \nabla u(x)), & \text{on } D \\ u = \Phi, & \text{on } \partial D. \end{cases}$$
(5.2)

Calculating rigorously, if $u \in \mathcal{D}(L) \cap \mathcal{D}(L^*)$, we get, for s < t

$$u(X_{t}) - u(X_{s}) = \int_{s}^{t} \nabla u(X_{r}) dM_{r} + \int_{s}^{t} Lu(X_{r}) dr$$

$$= \int_{s}^{t} \nabla u(X_{r}) dM_{r} + \int_{s}^{t} L^{*}u(X_{r}) dr + \int_{s}^{t} div(bu)(X_{r}) dr + \int_{s}^{t} < b, \nabla u > (X_{r}) dr.$$

(5.3)

But the term $\int_s^t div(bu)(X_r) dr$ does not have a real meaning, because b is just measurable. Inspired by [28] and [35], we will try to transform $\int_s^t div(bu)(X_r) dr$ in the following discussion.

Fix x_0 and set $p_t(x) := p(t, x_0, x)$, for every $u \in W^{1,2}$, we define

$$\alpha^{u}(s,t) := \int_{s}^{t} \langle A\nabla \ln p_{t}, \nabla u \rangle \langle X_{r} \rangle dr,$$

$$\beta^{u}(s,t) := \int_{s}^{t} \langle b, \nabla u \rangle \langle X_{r} \rangle dr.$$

Denote the semigroup associate with the adjoint process under P_{x_0} by

$$S_{s,t}f(x) := E_{x_0}[f(X_s)|X_t = x] = \int \frac{p(s, x_0, y)p(t - s, y, x)}{p(t, x_0, x)} f(y)m(dy)$$
$$= \frac{1}{p_t(x)} P_{t-s}^*(p_s f)(x), \quad s < t.$$

By the following fact

$$L^* P_u^*(p_{t-u}f) = P_u^* L^*(p_{t-u}f)$$

= $P_u^*[(L^* p_{t-u})f + \langle A\nabla p_{t-u}, \nabla f \rangle + p_{t-u}(\frac{1}{2}\nabla(A\nabla f) - \langle b, \nabla f \rangle)],$

we get

$$\begin{split} S_{s,t}f(x) &- f(x) \\ = \ \frac{1}{p_t(x)} [P_{t-s}^*(p_s f)(x) - p_t f(x)] \\ = \ \frac{1}{p_t(x)} \int_0^{t-s} \partial_u [P_u^*(p_{t-u} f)(x)] du \\ = \ \frac{1}{p_t(x)} \int_0^{t-s} (\partial_u P_u^*)(p_{t-u} f)(x) - P_u^*(\partial_u p_{t-u} f)(x)) du \\ = \ \frac{1}{p_t(x)} \int_0^{t-s} (L^* P_u^*)(p_{t-u} f)(x) - P_u^*[(L^* p_{t-u}) f](x)) du \\ = \ \int_0^{t-s} \frac{1}{p_t(x)} P_u^*[< A \nabla p_{t-u}, \nabla f > + p_{t-u}(\frac{1}{2} \nabla (A \nabla f) - < b, \nabla f >)] du \\ = \ \int_s^t \frac{1}{p_t(x)} P_{t-u}^*(p_u < A \nabla \ln p_u, \nabla f >) du \\ + \int_s^t \frac{1}{p_t(x)} P_{t-u}^*(p_u < \frac{1}{2} \nabla (A \nabla f) - < b, \nabla f >) du \\ = \ \int_s^t S_{r,t}(< A \nabla \ln p_u, \nabla f > + \frac{1}{2} \nabla (A \nabla f) - < b, \nabla f >)(x) du. \end{split}$$

By the Markovian property of the adjoint process, it follows that

$$E_{x_0}[f(X_s) - f(X_t) - \alpha^f(s, t) - \frac{1}{2} \int_s^t \nabla (A \nabla f)(X_r) \, dr + \beta^f(s, t) |\overleftarrow{\mathcal{F}}_s^t]$$

= $E_{x_0}[f(X_s) - f(X_t) - \alpha^f(s, t) - \frac{1}{2} \int_s^t \nabla (A \nabla f)(X_r) \, dr + \beta^f(s, t) |X_t]$
= 0. (5.4)

So far we know, when the time t > 0 is fixed, the process

$$\overleftarrow{M}^f(s,t) := f(X_s) - f(X_t) - \frac{1}{2} \int_s^t \nabla(A\nabla f)(X_r) \, dr - \alpha^f(s,t) + \beta^f(s,t), \quad s < t(5.5)$$

is a martingale with respect to the filtration $\{\overleftarrow{\mathcal{F}}_s^t, s < t\}$. And it is not difficult to see the sharp process of the backward martingale $\overleftarrow{M}^f(s, t)$ is

$$\langle \overleftarrow{M}^{f}(\cdot,t), \overleftarrow{M}^{f}(\cdot,t) \rangle_{s}^{t} = \int_{s}^{t} \langle A\nabla f, \nabla f \rangle \langle X_{r} \rangle dr.$$

Since the coordinate of point x is $x = (x_1, ..., x_d)$, we set $f(x) = x_i \in W^{1,2}$. Then it follows

$$\overleftarrow{M}^i(s,t) = X_s^i - X_t^i - \frac{1}{2} \sum_j \int_s^t \partial_j(a_{ij})(X_r) \, dr - \alpha^i(s,t) + \beta^i(s,u).$$

We write the backward martingale as following for short:

$$\overleftarrow{M}(s,t) = X_s - X_t - \frac{1}{2} \int_s^t \nabla A(X_r) \, dr - \int_s^t A \nabla (\ln p_r) (X_r) \, dr + \int_s^t b(X_r) \, dr.$$

Define the stochastic integral with respect to the backward martingale $\overleftarrow{M}^{i}(s,t)$ as follows, for a function g,

$$\int_{s}^{t} g(X_{r}) d\overleftarrow{M}^{i}(r,t) = \lim_{\|\Delta\| \to 0} \sum_{j=0}^{k} g(X_{t_{j+1}}) (\overleftarrow{M}^{i}(t_{j},t) - \overleftarrow{M}^{i}(t_{j+1},t)),$$

where \triangle : $s = t_0 < t_1 < ... < t_k = t$ is the partition of the interval [s,t] and

 $\|\triangle\| = \max_j (t_{j+1} - t_j).$

Similarly as the "forward" process, we conclude that

$$\int_{s}^{t} \nabla u(X_{r}) d\overline{M}(r,t) = \overline{M}^{u}(s,t).$$
(5.6)

In fact,

$$\int_{s}^{t} \nabla u(X_{r}) d\overline{M}(r,t)$$

$$= \lim_{\|\Delta\|\to 0} \sum_{j=0}^{k} \nabla u(X_{t_{j+1}}) (X_{t_{j}} - X_{t_{j+1}}) - \frac{1}{2} \int_{s}^{t} < \nabla A, \nabla u > (X_{r}) dr$$

$$- \int_{s}^{t} < A \nabla (\ln p_{r}), \nabla u > (X_{r}) dr + \int_{s}^{t} < b, \nabla u > (X_{r}) dr.$$
(5.7)

Set the first term in the right side of the equation (5.7) as (I), then it follows

$$(I) = \lim_{\|\Delta\|\to 0} \sum_{j=0}^{k} \nabla u(X_{t_{j}})(X_{t_{j}} - X_{t_{j+1}}) + \lim_{\|\Delta\|\to 0} \sum_{j=0}^{k} (\nabla u(X_{t_{j+1}} - \nabla u(X_{t_{j}}))(X_{t_{j}} - X_{t_{j+1}}))$$

$$= -\int_{s}^{t} \nabla u(X_{r})dM_{r} - \int_{s}^{t} < \frac{1}{2}\nabla A + b, \nabla u > (X_{r})dr - \int_{s}^{t} \sum_{ij} a_{ij}\frac{\partial^{2}u}{\partial x_{i}\partial x_{j}}(X_{r})dr$$

$$= u(X_{s}) - u(X_{t}) + \int_{s}^{t} Lu(X_{r})dr - \int_{s}^{t} < \frac{1}{2}\nabla A + b, \nabla u > (X_{r})dr$$

$$- \int_{s}^{t} \sum_{ij} a_{ij}\frac{\partial^{2}u}{\partial x_{i}\partial x_{j}}(X_{r})dr.$$

By the above expression of (I), we get the conclusion (5.6).

Therefore, by (5.3) and (5.5), we conclude the following two results.

(1) For any $f \in W^{1,2}$, it follows

$$f(X_t) - f(X_s) = \frac{1}{2} \int_s^t \nabla f(X_r) dM_r - \frac{1}{2} \int_s^t \nabla f(X_r) d\overleftarrow{M}(r, t) - \frac{1}{2} \alpha^f(s, t) + \beta^f(s, t).$$
(5.8)

(2)For any function $f \in \mathcal{D}(L)$, it follows

$$-\int_{s}^{t} \nabla (A\nabla f)(X_{r}) dr$$

=
$$\int_{s}^{t} \nabla f(X_{r}) dM_{r} + \int_{s}^{t} \nabla f(X_{r}) d\overleftarrow{M}(r,t) + \int_{s}^{t} \langle A\nabla \ln p_{r}, \nabla f \rangle \langle X_{r} \rangle dr.$$

We use the notation in [35]. For function $g = (g_1, ..., g_d)$, define

$$\int_{s}^{t} g(X_{r}) * dX_{r}$$

$$= \sum_{i} \left(\int_{s}^{t} g_{i}(X_{r}) dM_{r}^{i} + \int_{s}^{t} g_{i}(X_{r}) d\overleftarrow{M}^{i}(r,t) \right) + \int_{s}^{t} \langle A\nabla \ln p_{r}, g \rangle (X_{r}) dr$$

We can generalize the second conclusion (2) as follows, which is inspired by [35].

Lemma 5.1.1. Assume $g \in L^2(\mathbb{R}^d, \mathbb{R}^d)$, $f \in L^2(\mathbb{R}^d)$ and div(Ag) = f in the weak sense. Then it holds that

$$-\int_{s}^{t} f(X_r) dr = \int_{s}^{t} g(X_r) * dX_r.$$

The function $u \in W^{1,2}$ is the weak solution of the equation (5.2), which can be rewritten as $div(A(\frac{1}{2}\nabla u - A^{-1}b)) = -F(x, u(x))$ in the weak sense. By Lemma 5.1.1, we have

$$\frac{1}{2}\int_{s}^{t} \nabla u(X_{r}) * dX_{r} - \int_{s}^{t} A^{-1}b(X_{r}) * dX_{r} = \int_{s}^{t} F(X_{r}, u(X_{r}) dr) dr$$

By (5.8), we know that

$$\frac{1}{2} \int_{s}^{t} \nabla u(X_{r}) * dX_{r}$$

$$= \frac{1}{2} \int_{s}^{t} \nabla u(X_{r}) dM_{r} + \frac{1}{2} \int_{s}^{t} \nabla u(X_{r}) d\overline{M}(r,t) + \int_{s}^{t} \langle A\nabla \ln p_{r}, \nabla u \rangle (X_{r}) dr$$

$$= -u(X_{t}) + u(X_{s}) + \int_{s}^{t} \nabla u(X_{r}) dM_{r} + \int_{s}^{t} \langle b, \nabla u \rangle (X_{r}) dr.$$

Then

$$\int_{s}^{t} F(X_{r}, u(X_{r})) dr = -u(X_{t}) + u(X_{s}) + \int_{s}^{t} \nabla u(X_{r}) dM_{r} + \int_{s}^{t} \langle b, \nabla u \rangle (X_{r}) dr - \int_{s}^{t} A^{-1}b(X_{r}) * dX_{r}.$$

Therefore, we have the following expression of u(X),

$$u(X_t) - u(X_s) = \int_s^t \nabla u(X_r) dM_r - \int_s^t F(X_r, u(X_r)) dr + \int_s^t \langle b, \nabla u \rangle (X_r) dr - \int_s^t A^{-1} \hat{b}(X_r) * dX_r.$$
(5.9)

As a conclusion of this example, we say that the equation (5.9) supplies a candidate of the solution to the BSDE. Suppose u is the solution of the equation (5.2) and substitute the fixed time t for the first hitting time τ of the boundary ∂D . Then it follows

$$u(X_s) = \Phi(X_{\tau}) + \int_s^{\tau} F(X_r, u(X_r)) \, dr - \int_s^{\tau} \langle b, \nabla u \rangle (X_r) \, dr \\ + \int_s^t A^{-1} \hat{b}(X_r) * dX_r - \int_s^{\tau} \nabla u(X_r) \, dM_r.$$

5.2 Future Studies

Set

$$H_T := \{ u \in L^2([0,T] \times \mathbb{R}^d) : t \mapsto u(t,\cdot) \text{ is continuous in } L^2(\mathbb{R}^d) \text{ on } [0,T], \\ u(t,\cdot) \in H^1(\mathbb{R}^d) \text{ and } \int_0^T (A\nabla u(t,\cdot), \nabla u(t,\cdot)) dt < \infty \}.$$

The following theorem is the main result in [35], which is the essential inspiration of our future study.

Theorem 5.2.1. If $\Phi \in L^2(\mathbb{R}^d)$ and $f : [0,T] \times \mathbb{R}^d \times \mathbb{R} \times \mathbb{R}^d \to \mathbb{R}$, $g : [0,T] \times \mathbb{R}^d \times \mathbb{R} \times \mathbb{R}^d \to \mathbb{R}$ satisfy the conditions,

$$(1) \ f(\cdot, \cdot, 0, 0) \in L^{2}([0, T] \times R^{d}) \ and \ g(\cdot, \cdot, 0, 0) \in L^{2}([0, T] \times R^{d}; R^{d}),$$

$$(2) \ |f(t, x, y, z) - f(t, x, y', z')| \leq C(|y - y'| + |z - z'|),$$

$$(3) \ |g(t, x, y, z) - g(t, x, y', z')| \leq C(|y - y'| + \alpha|z - z'|),$$

with some constant C > 0 and $\alpha \in (0, 1)$. Then there exists a unique determined solution $u \in H_T$ of the following equation:

$$(\partial_t + \nabla(A\nabla))u(t, x) + f(t, x, u(t, x), \frac{1}{\sqrt{2}}\nabla u(t, x)\sigma(x))$$
$$-div(Ag)(t, x, u(t, x), \frac{1}{\sqrt{2}}\nabla u(t, x)\sigma(x)) = 0.$$
$$u(T, x) = \Phi(x)$$
(5.10)

Moreover, set $Y_t = u(t, X_t)$ and $Z_t = (\nabla u)(t, X_t)$, then (Y, Z) is a solution of the following BSDE,

$$Y_{t} = \Phi(X_{T}) - \int_{t}^{T} \langle Z_{r}, dM_{r} \rangle + \int_{t}^{T} f(r, X_{r}, Y_{r}, \frac{1}{\sqrt{2}} Z_{r} \sigma(X_{r})) dr + \frac{1}{2} \int_{t}^{T} g(r, X_{r}, Y_{r}, \frac{1}{\sqrt{2}} Z_{r} \sigma(X_{r})) * dX, \qquad (5.11)$$

for any $0 \le t \le T$, $P^m - a.s.$.

We notice that, the processes Y and Z as a pair of solutions to the BSDE (5.11) must be two functions u and ∇u composed of X_t respectively, i.e. Y = u(X) and $Z = \nabla u(X)$. But as a general solution to the BSDE, this is not easy to be satisfied.

Last but not least, we list three problems for our future studies as the conclusion of the whole thesis.

(1) Study the BSDE (5.11), not only in the PDE (5.10) point of view, but also its own properties as a backward differential equation.

(2) Consider the following Dirichlet boundary problem:

$$\begin{cases} \frac{1}{2}\nabla(A\nabla u)(x) + b \cdot \nabla u(x) - div\hat{b}(x, u(x)) + q(x)u(x) = -F(x, u(x), \nabla u(x)), & \text{on } D\\ u = \Phi, & \text{on } \partial D \end{cases}$$

(3) Consider the following Neumann boundary problem:

$$\begin{cases} \frac{1}{2}\nabla(A\nabla u)(x) + b \cdot \nabla u(x) - div\hat{b}(x, u(x)) + q(x)u(x) = -F(x, u(x), \nabla u(x)), & \text{on } D\\ \frac{1}{2} < A\nabla u, \vec{n} > = <\hat{b}(x, u(x)), \vec{n} >, & \text{on } \partial D \end{cases}$$

Bibliography

- D.G.Aronson, Bounds for the Fundamental Solution of a Parabolic Equation, Bulletin of the American Mathematical Society 73 (1967), 890-896.
- [2] D.G.Aronson and J.Serrin, Local Behavior of Solutions of Quasilinear Parabolic Equations, Arch. Rational Mech. Anal. 25 (1967), 81-122.
- [3] R.F.Bass and P.Hsu, Some Potential Theory for Reflecting Brownian Motion in Hölder and Lipschitz Domains, Ann. Probab. 19 (1991), 486-508.
- [4] Ph. Brianda, B. Delyona, Y. Hu, E. Pardoux and L. Stoica, L_p Solutions of Backward Stochastic Differential Equations, Stochastic Process. Appl. 108 (2003), 109-129.
- [5] Z.Q.Chen, On Reflecting Diffusion Processes and Skorokhod Decompositions, Probab. Theory Related Fields, 94 (1993), 281-315,
- [6] Z.Q. Chen, P.J.Fitzsimmons, K.Kuwae and T.S.Zhang, Perturbation of Symmetric Markov Processes, Proba. Theory Related Fields 140 (2008), 239-275.
- Z.Q.Chen and T.S.Zhang, *Time-reversal and Elliptic Boundary Value Problems*, Ann. Probab. **37** (2009), 1008-1043.
- [8] Z. Q. Chen and T. S. Zhang, A Probabilistic Approach to Mixed Boundary Value Problems for Elliptic Operators with Singular Coefficients, Preprint.
- [9] K. L. Chung and J. B. Walsh, *To Reverse a Markov Process*, Acta Math. **123** (1970), 225-251.
- [10] E. Coddington, and N. Levinson, *Theory of Ordinary Differential Equations*, McGraw-Hill, New York, (1955).

- [11] R. W. R. Darling and E. Pardoux, Backwards SDE with Random Terminal Time and Applications to Semilinear Elliptic PDE, Ann. Probab. 25 (1997), 1135-1159.
- [12] S. Z. Fang, S. G., Peng, L. M. Wu and J.A. Yan, elect topics in stochastic analysis (in Chinese), Science Press, Beijing, (1997).
- [13] A. Friedman, Partial differential equations of parabolic type, Prentice-Hall (1964).
- [14] E. B. Fabes and D. W. Stroock, A New Proof of Moser's Parabolic Harnack Inequality Using the Old Ideas of Nash, Arch. Rational Mech. Anal. 96 (1986), 327-338.
- [15] M. Fukushima, Y. Oshima and M.Takeda, Dirichlet Forms and Symmetric Markov Processes, De Gruyter, New York, (1994).
- [16] W. D. Gerhard, The Probabilistic Solution of the Dirichlet Problem for $\frac{1}{2}\nabla + \langle a, \nabla \rangle + b$ with Singular Coefficients, J. Theor. Probab. 5 (1992), 503-520.
- [17] D. Gilbarg and N. S. Trudinger, *Elliptic Partial Differential Equations of Second Order*,
 2nd ed. Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences] 224. Springer, Berlin, (1983).
- [18] S.W.He, J.G.Wang and J.A.Yan, Semimartingale Theory and Stochastic Calculus, S. Kexue Chubanshe (Science Press), Beijing, (1992).
- [19] Y. Hu, Probabilistic Interpretation of a System of Quasilinear Elliptic Partial Differential Equations under Neumann Boundary Conditions, Stochastic Process. Appl. 48 (1993),108-121.
- [20] L. Hörmander, The Analysis of Linear Partial Differential Operators I: Distribution Theory and Fourier Analysis, Berlin, Springer (2003).
- [21] P. Hsu, Reflecting Brownian Motion, Boundary Local Time and the Neumann Problem, thesis, Stanford University, (1984).
- [22] S. Ito, A Boundary Value Problem of Partial Differential Equations of Parabolic Type, Duke Math. J. 24 (1957), 299-312.

- [23] Y. Le Jan, Dual Markovian Semigroups and Processes, Notes Math. 923(1981), Springer-Verlag.
- [24] S. Kakutani, Two-dimensional Brownian Motion and Harmonic Functions, Proc. Imp.Acad. 20 (1944), 706-714.
- [25] A. Lejay, BSDE Driven by Dirichlet Process and Semi-linear Parabolic PDE. Application to Homogenization, Stochastic Process. Appl, 97 (2002), 1-39.
- [26] P. L. Lions and A. S. Sznitman, Stochastic Differential Equations with Reflecting Boundary Conditions, Comm. Pure Appl. Math. 37 (1984), 511-537.
- [27] J. Lunt, T. J. Lyons and T. S. Zhang, Integrability of Functionals of Dirichlet Processes, Probabilistic Representations of Semigroups, and Estimates of Heat Kernels, J. Funct. Anal. 153 (1998), 320-342.
- [28] T.J.Lyons and L.Stoica, The Limits of Stochastic Integrals of Differential Forms, Ann.
 Probab. 27 (1999), 1-49.
- [29] T. J. Lyons and T. S. Zhang, Convergence of Non-symmetric Dirichlet Processes, Stochastics Stochastics Rep. 57 (1996), 159-167.
- [30] Z. M. Ma and M. Röckner, Introduction to the Theory of Nonsymmetric Dirichlet Forms, Springer, Berlin, (1992).
- [31] É. Pardoux and S. G. Peng, Backward Stochastic Differential Equations and Quasilinear Parabolic Partial Differential Equations, In Stochastic Partial Differential Equations and Their Applications (Charlotte, NC, 1991) (B. L. Rozovskii and R. B. Sowers, eds.). Lecture Notes in Control and Inform. Sci. **176**, 200-217. Springer, Berlin
- [32] S. G. Peng, Probabilistic Interpretation for Systems of Quasilinear Parabolic Partial Differential Equations, Stochastics Stochastics Rep. 37 (1991), 61-74.
- [33] R. Song, Sharp Bounds on Density, Green Function and Jumping Function of Subordinate Killed BM. Probability Theory and Related Fields. 128 (2004), 606-628.

- [34] K.Sato and T.Ueno, Multi-dimensional Diffusion and the Markov Process on the Boundary, J. Math. Kyoto Univ. 4 (1964), 529-605.
- [35] I. L. Stoica, A Probabilistic Interpretation of the Divergence and BSDE's, Stochastic Processes and their Applications 103 (2003), 31–55.
- [36] N. S. Trudinger, Linear Elliptic Operators with Measurable Coefficients, Ann. Scuola Norm. Sup. Pisa 27(1973), 265-308.
- [37] X. Yang and T. S. Zhang, The Estimates of Heat Kernels with Neumann Boundary Conditions, (Accepted for publication by Potential Analysis).
- [38] K.Yosida, Functional Analysis, Springer-Verlag, 6th edition, (1980).
- [39] K.Wehrheim, Uhlenbeck Compactness, Series of Lectures in Mathematics, European Mathematical Society, (2003).
- [40] Q.Zhang, A Harnack Inequality for the Equation $\nabla(a\nabla) + b\nabla u = 0$, when $|b| \in K_{n+1}$, Manuscripta math. 89 (1995), 61-77.
- [41] Q.Zhang, Gaussian Bounds for the Fundamental Solutions of ∇(A∇u)+B∇u-ut = 0, Manuscripta math. 93 (1997), 381-390.
- [42] T. S. Zhang, A Probabilistic Approach to Dirichlet Problems of Semilinear Elliptic PDEs with Singular Coefficients, Ann.Probab. 39 (2011),1502-1527.